UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): January 19, 2017

COLONY NORTHSTAR, INC.

(Exact name of registrant as specified in its charter)

Maryland
(State or other jurisdiction of incorporation)

001-37980 (Commission File Number) 46-4591526 (IRS Employer Identification No.)

515 S. Flower Street, 44th Floor Los Angeles, CA (Address of principal executive offices)

90071 (Zip Code)

Registrant's telephone number, including area code: (310) 282-8820

Not Applicable

(Former name or former address, if changed since last report.)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following	ing
provisions (see General Instruction A.2. below):	

Ш	Written Communications pursuant to Rule 423 under the Securities Act (17 GFR 230.423)
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Item 1.01. Entry into a Material Definitive Agreement.

The information set forth under Item 2.01 below related to the Purchase Agreement (as defined below) is incorporated by reference into this Item 1.01.

Item 2.01. Completion of Acquisition or Disposition of Assets.

On January 19, 2017, Colony NorthStar, Inc. (the "Company") completed the previously announced sale of an 18.7% interest in the Company's healthcare real estate portfolio (the "Healthcare Portfolio") to Derwood Limited ("Derwood"), in exchange for \$350,000,000 (including \$20,000,000 of pre-funded capital items) (the "Transaction"), pursuant to that certain purchase and sale agreement, dated as of November 4, 2016, by and among NorthStar Realty Finance Limited Partnership, NorthStar Healthcare JV Holdings, LLC, NorthStar Healthcare REIT, LLC, NorthStar TK Healthcare Operating Company, LLC, NorthStar Healthcare JV, LLC, NRFC Healthcare Holding Company, LLC and Derwood (the "Purchase Agreement").

The Healthcare Portfolio is currently comprised of the Company's ownership interest, excluding existing minority interest holders, in 191 senior housing properties, 113 medical office properties, 14 hospitals and 107 skilled nursing facilities (such properties collectively referred to as the "Healthcare Properties"). The Transaction represents an implied valuation for the Healthcare Properties of approximately \$5.4 billion, which excludes the portfolio of medical office buildings sold prior to the Transaction for an aggregate value of approximately \$0.8 billion. The Healthcare Properties are generally operated under net leases or through management agreements with independent third-party operators.

NorthStar Realty Finance Limited Partnership (and/or certain affiliated transaction parties) made certain customary representations, warranties and covenants concerning such parties and the Healthcare Portfolio. NorthStar Realty Finance Limited Partnership also agreed to indemnify (subject to caps and limitations) the buyer indemnified parties for certain losses arising under the terms of the Purchase Agreement.

The Company, through an indirect subsidiary of its operating partnership (the "CLNS Partner"), and Derwood will hold their interests in the Healthcare Portfolio through a newly formed joint venture, which will be indirectly owned approximately 18.7% by Derwood and approximately 81.3% by the CLNS Partner (the "Joint Venture"). The Joint Venture in turn will own approximately 87.7% of the Healthcare Properties on an aggregate basis, resulting in Derwood, the CLNS Partner and existing minority interest holders owning approximately 16.4%, 71.3% and 12.3%, respectively, of the Healthcare Properties on an aggregate basis. The CLNS Partner will act as the manager of the Joint Venture and is generally responsible for the day-to-day affairs of the Healthcare Portfolio.

The foregoing description of the Purchase Agreement and the transactions contemplated thereby does not purport to be complete and is qualified in its entirety by reference to the Purchase Agreement, which is filed as Exhibit 10.1 to this Current Report on Form 8-K and is incorporated herein by reference.

Item 9.01. Financial Statements and Exhibits.

(b) Pro forma financial information

The pro forma financial information of the Company required by this Item 9.01 of Current Report on Form 8-K is presented in Note 5.E Merger Adjustments, footnote (6), in the unaudited pro forma condensed consolidated statements of operations of the Company for the nine months ended September 30, 2016 and for the year ended December 31, 2015, filed as Exhibit 99.1 to this Form 8-K and incorporated herein by reference.

(d) Exhibits

Exhibit No.	Description
10.1	Purchase and Sale Agreement, dated as of November 4, 2016, by and among NorthStar Realty Finance Limited Partnership, NorthStar Healthcare JV Holdings, LLC, NorthStar Healthcare REIT, LLC, NorthStar TK Healthcare Operating Company, LLC, NorthStar Healthcare JV, LLC and NRFC Healthcare Holding Company, LLC and Derwood Limited
99.1	Unaudited Pro Forma Condensed Consolidated Financial Statements of Colony NorthStar, Inc.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, as amended, the Registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Date: January 25, 2017 COLONY NORTHSTAR, INC.

By: /s/ Darren J. Tangen

Darren J. Tangen

Chief Financial Officer and Treasurer

EXHIBIT INDEX

No.	Description
10.1	Purchase and Sale Agreement, dated as of November 4, 2016, by and among NorthStar Realty Finance Limited Partnership, NorthStar Healthcare JV Holdings, LLC, NorthStar Healthcare REIT, LLC, NorthStar TK Healthcare Operating Company, LLC, NorthStar Healthcare JV, LLC and NRFC Healthcare Holding Company, LLC and Derwood Limited
99.1	Unaudited Pro Forma Condensed Consolidated Financial Statements of Colony NorthStar, Inc.

PURCHASE AND SALE AGREEMENT

by and among

Derwood Limited

as Buyer,

NorthStar Realty Finance Limited Partnership,

as Seller

NorthStar Healthcare JV Holdings, LLC,

NorthStar TK Healthcare REIT, LLC,

NorthStar TK Healthcare Operating Company, LLC,

NorthStar Healthcare JV, LLC,

and

NRFC Healthcare Holding Company, LLC

Dated as of November 4, 2016

TABLE OF CONTENTS

		<u>Page</u>
Article I DE	FINITIONS	1
1.1	Defined Terms	1
1.2	Terms Defined Elsewhere	13
1.3	Interpretation	15
Article II PU	RCHASE AND SALE OF INTERESTS; PURCHASE PRICE	16
2.1	Sale and Purchase of Purchased Interests	16
2.2	Purchase Price	16
2.3	Escrow Arrangements	17
2.4	Adjustments to Equity Value	17
2.5	Governing Documents	21
2.6	Tax Treatment	21
2.7	Allocation of Purchase Price	22
2.8	Withholding	22
2.9	Capital Advance	22
Article III R	EPRESENTATIONS AND WARRANTIES OF THE SELLER	23
3.1	Title to the Purchased Interests	23
3.2	Organization	23
3.3	Authorization of Agreement	23
3.4	Conflicts; Consents of Third Parties	24
3.5	Legal Proceedings	24
3.6	Tax Status	25
3.7	Limitations of Representations and Warranties	25
Article IV R	EPRESENTATIONS AND WARRANTIES OF THE COMPANY	25
4.1	Organization	25
4.2	Authorization of Agreement	26
4.3	Conflicts; Consents of Third Parties	27
4.4	Capitalization; Subsidiaries	27
4.5	Financial Statements	29
4.6	Undisclosed Liabilities	30
4.7	Absence of Certain Changes	30
4.8	Compliance with Laws	30
4.9	Legal Proceedings	31
4.10	Properties	32
4.11	Tax Matters	34
4.12	No Employees; No Benefit Plans	36
4.13	No Condemnation	36
4.14	Intellectual Property	36
4.15	Environmental Matters	37

4.16	Material Contracts	37
4.17	Insurance	40
4.18	Affiliated Transactions	40
4.19	Financial Advisor	41
4.20	Disclaimer of any Representations Regarding Estimates and Projections	41
4.21	Limitations of Representations and Warranties	41
Article V RE	PRESENTATIONS AND WARRANTIES OF BUYER	42
5.1	Organization	42
5.2	Authorization of Agreement	42
5.3	Conflicts; Consents of Third Parties	42
5.4	Legal Proceeding	43
5.5	Financial Advisor	43
5.6	Purchase for Investment	43
5.7	Financial Ability	43
5.8	Sophisticated Party	43
5.9	Compliance with Laws	43
5.10	Investigation; Limitation on Representations and Warranties; Disclaimer of Other Representations and Warranties	43
5.11	No Other Representations or Warranties	44
Article VI C	OVENANTS AND AGREEMENTS	44
6.1	Conduct of the Healthcare Portfolio Pending the Closing	44
6.2	Efforts; Consents and Approvals	45
6.3	Pre-Closing Restructuring	47
6.4	Financing	47
6.5	Public Announcements	48
6.6	Conveyance Taxes	48
6.7	Expenses	48
6.8	Further Assurances	49
6.9	Managers	49
6.10	Casualty and Condemnation	49
6.11	Advice of Changes	49
ARTICLE V	II CLOSING	50
7.1	Closing	50
7.2	Conditions to Each Party's Obligations to Effect the Closing	50
7.3	Conditions to Obligations of the Seller Group	50
7.4	Conditions to Obligations of Buyer	51
7.5	Closing Deliveries	52
Article VIII l	INDEMNIFICATION	53
8.1	Survival	53
8.2	Obligation of Seller to Indemnify	53
8.3	Obligation of Buyer to Indemnify	54

54

8.4

Certain Limitations

	8.5	Indemnification Procedure	55
	8.6	Measure of Indemnification	56
	8.7	Exclusivity of Indemnity	57
	8.8	Tax Treatment of Indemnity Payments	58
Art	icle IX TERM	INATION OF AGREEMENT	58
	9.1	Termination	58
	9.2	Effect of Termination	59
Art	icle X MISCE	LLANEOUS	63
	10.1	Dispute Resolution; Venue	63
	10.2	Notices	64
	10.3	Entire Agreement	65
	10.4	Waivers and Amendments	66
	10.5	Governing Law	66
	10.6	Binding Effect; Assignment	66
	10.7	Severability of Provisions	66
	10.8	Disclosure Schedule	67
	10.9	Counterparts	67
	10.10	No Personal Liability	67
	10.11	No Third Party Beneficiaries	67

Exhibits

Exhibit A	Pre-Closing Restructuring
Exhibit B	Sample Adjustment Calculation and Adjustment Principles
Exhibit C	Form of Newco LLC Agreement
Exhibit D	Form of JV LLC Agreement
Exhibit E-1	Form of Buyer Officer's Certificate
Exhibit E-2	Form of Seller Officer's Certificate
Exhibit E-3	Form of Company Officer's Certificate
Exhibit F	Asset Management Services
Exhibit G	Form of Capital Contribution Advance Guarantee
Exhibit H	Allocated Purchase Price (JV Interest)
Exhibit I	Form of JV Letter Agreement

PURCHASE AND SALE AGREEMENT

This **PURCHASE AND SALE AGREEMENT**, dated as of November 4, 2016 (this "<u>Agreement</u>"), by and among Derwood Limited, a Hong Kong limited company ("<u>Buyer</u>"), NorthStar Realty Finance Limited Partnership, a Delaware limited partnership ("<u>Seller</u>" or "<u>NRF OP</u>"), NorthStar Healthcare JV Holdings, LLC, a Delaware limited liability company ("<u>NRF OP SPV</u>"), NorthStar TK Healthcare REIT, LLC, a Delaware limited liability company ("<u>JV OpCo</u>"), NorthStar Healthcare JV, LLC, a Delaware limited liability Company ("<u>Newco</u>"), and NRFC Healthcare Holding Company, LLC, a Delaware limited liability company (the "<u>Company</u>" and, together with Seller, NRF OP SPV, JV, JV OpCo and Newco, the "<u>Seller Group</u>"). Buyer and the members of the Seller Group are each sometimes referred to herein as a "<u>Party</u>" and collectively as the "<u>Parties</u>".

WHEREAS, as of the date hereof, the Company, directly or indirectly through its ownership interests in its Subsidiaries (each a "<u>Company Subsidiary</u>" and collectively, the "<u>Company Subsidiaries</u>"), owns interests in 457 real property healthcare assets (each, together with any Eligible Property acquired after the date hereof, a "<u>Property</u>" and collectively, the "<u>Healthcare Portfolio</u>");

WHEREAS, upon completion of the Pre-Closing Restructuring, Seller (or its Affiliates), directly or indirectly through JV, will own one hundred percent (100%) of the ownership interests in Newco, which in turn will own one hundred percent (100%) of the Company; and

WHEREAS, Seller has agreed to sell (or cause to be sold) to Buyer, and Buyer has agreed to purchase from Seller, no more than 47% of the common limited liability company interests of the JV issued and outstanding (the "<u>JV HoldCo Interests</u>") representing indirect ownership of the Buyer Ownership Percentage of the Company Interests (as defined below) (the "<u>Purchased Interests</u>"), upon the terms and subject to the conditions of this Agreement.

NOW, THEREFORE, in consideration of the mutual covenants, representations, warranties and agreements entered into herein, and for other good and valuable consideration, the receipt and adequacy of which is hereby acknowledged, the parties hereto, intending to be legally bound hereby, do agree as follows:

ARTICLE I

DEFINITIONS

1.1 <u>Defined Terms</u>. For all purposes of this Agreement, the following terms shall have the respective meanings set forth in this Section 1.1 (such definitions to be equally applicable to both the singular and plural forms of the terms herein defined):

"<u>Adjustment Amount</u>" means, with respect to any Property, (a) in the case of a sale consummated prior to Closing, excluding Condemnations and any Option Sale, but including the MOB Sale, an amount equal to (i) the Allocated Purchase Price (JV Interest) for such Property minus (ii) the amount of any Company Indebtedness related to such Property that

is repaid, (b) in the case of a purchase consummated prior to the Closing, an amount equal to the sum of (i) the product of (A) the purchase price paid by the Company or a Company Subsidiary for such Property prior to Closing (including any deposit) <u>multiplied</u> by (B) the Company's direct or indirect percentage ownership interest in such Property, <u>plus</u> (ii) the out-of-pocket fees and expenses incurred or payable by the Seller Group or its Affiliates in connection with such purchase <u>minus</u> (iii) the amount of any Company Indebtedness related to such Property that is incurred, or (c) in the case of a Condemnation, the net proceeds directly or indirectly received by the Company for such Property prior to Closing, after the repayment of any Company Indebtedness related to such Property and deducting out-of-pocket fees and expenses incurred or payable, directly or indirectly, by the Company in connection with such Condemnation.

"Affiliate" means, with respect to any Person, any other Person controlling, controlled by or under common control with such Person. The term "control" (including, with correlative meaning, the terms "controlled by" and "under common control with"), as applied to any Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting or other securities, by contract or otherwise. For the avoidance of doubt, NRF and NSAM are Affiliates of Seller.

"Allocated Purchase Price (JV Interest)" means, with respect to a Property, the amount set forth opposite such Property in the column titled "Allocated Purchase Price (JV Interest)" on Exhibit H, provided, that, in the case of Properties located in the United Kingdom, the amount for each such Property shall be re-valued to be the product of (a) the amount in United Kingdom Pound Sterling utilized to determine the U.S. Dollar amount for such Property as set forth on Exhibit H on the date hereof multiplied by (b) the Closing Date Exchange Rate.

"Asset Management Services" means the asset management services described on Exhibit F.

"Balance Sheet Date" means March 31, 2016.

"<u>Business Day</u>" means a day other than Saturday, Sunday or any day on which banks located in New York, New York, Beijing, PRC or Hong Kong are authorized or obligated by Law to close.

"Buyer Ownership Percentage" means a percentage determined by dividing (x) the Purchase Price by (y) the Total Equity Value as determined at Closing pursuant to Section 2.4, as such percentage may be adjusted pursuant to the applicable Side Letter.

"<u>Casualty</u>" means damage or destruction by acts of terrorism, sabotage, war (whether or not declared), rioting, civil disturbance or armed hostility, fire, earthquakes, tornados, hurricanes, windstorms or other weather conditions or natural calamities or other force majeure events or casualty.

"CIRC" means the Insurance Regulatory Commission of the PRC or its competent local counterparts.

"Claims" means suits, actions, proceedings, investigations, demands, claims, liabilities, fines, penalties, liens, judgments, losses, injuries, damages, settlement expenses or costs of whatever kind or nature, whether direct or indirect, known or unknown, contingent or otherwise (including any action or proceeding brought or threatened or ordered by any Governmental Entity), including attorneys' and experts' fees and expenses, and investigation and remediation costs.

"Closing Account" means the account or accounts designated in writing by Seller at least two (2) Business Days prior to the Closing Date.

"Closing Date" means the date (such date, the "Initial Closing Date") that is the later of (a) January 16, 2017 and (b) thirty (30) days following the satisfaction or waiver of the conditions to the Closing set forth in Sections 7.2, 7.3 and 7.4 (other than those conditions which, by their terms, are to be satisfied or waived at the Closing, but subject to the satisfaction or waiver of such conditions), provided, that, (i) the Initial Closing Date may be extended by Seller to the date (such date, the "Extended Closing Date") that is fifteen (15) days following the Initial Closing Date upon written notice to Buyer if such notice is delivered no later than three (3) Business Days following the satisfaction of the conditions to the Closing, but subject to the satisfaction or waiver of such conditions, and (ii) the Extended Closing Date may be extended by Seller to the date that is fifteen (15) days following the Extended Closing Date upon written notice to Buyer if such notice is delivered no later than three (3) Business Days following the Initial Closing Date; provided, further, that in the event the "Closing Date" as determined in accordance with this clause (b) were to fall on January 23, 2017 through February 3, 2017, the "Closing Date" shall be such date as mutually agreed in good faith amongst the Parties but, in the event the Parties are unable to agree to a Closing Date during such period, the "Closing Date" shall be February 6, 2017; or such earlier or later date as may be mutually agreed upon in writing by Seller and Buyer.

"Closing Date Exchange Rate" means the rate of exchange of British pounds into U.S. dollars reported in The Wall Street Journal on the Business Day immediately prior to the date on which Seller delivers the Estimated Adjustment Statement to Buyer.

"Closing Date Indebtedness" means the amount of Company Indebtedness as of the Closing Date as finally determined in accordance with Section 2.4.

"Closing Date Lender Reserves" means the amount of Lender Reserves as of the Closing Date as finally determined in accordance with Section 2.4.

"Closing Date Working Capital" means the amount of Working Capital as of the Closing Date as finally determined in accordance with Section 2.4.

"Code" means the U.S. Internal Revenue Code of 1986.

"Colony Merger Agreement" means the Agreement and Plans of Merger, dated June 2, 2016, by and among Seller, NSAM, Colony Capital, Inc., New Polaris Inc., New Sirius Inc., NorthStar Realty Finance Limited Partnership, Sirius Merger Sub-T, LLC and New Sirius Merger Sub, LLC, as may be amended, supplemented or modified from time to time.

"Company Indebtedness" means the Indebtedness of the Company and the Company Subsidiaries, as determined on a consolidated basis and in accordance with the Sample Adjustment Calculation and Adjustment Principles; *provided*, that, Company Indebtedness shall not include any Indebtedness taken into account in the determination of clause (b) of the Adjustment Amount.

"Company Indebtedness Estimate Amount" means Three Billion, Five Hundred Thirty-Six Million, One Hundred Thirty-One Thousand, Three Hundred Eight Dollars (\$3,536,131,308), provided, that, in the case of Company Indebtedness in United Kingdom Pound Sterling and related to Properties located in the United Kingdom, the amount of such Company Indebtedness shall be re-valued, and the Company Indebtedness Estimate Amount shall be increased or decreased, as and if applicable, by an amount equal to the product of (a) the amount in United Kingdom Pound Sterling utilized in determining the U.S. Dollar amount for the Company Indebtedness Estimate Amount on the date hereof multiplied by (b) the Closing Date Exchange Rate.

"Company Material Adverse Effect" means any change, event, condition or effect (each, an "Effect") that, individually or in the aggregate, has had or would reasonably be expected to have a material adverse effect on the assets, properties, liabilities, financial condition, business or results of operations of the Company and Company Subsidiaries, taken as a whole; provided, however, that no Effects resulting or arising from the following shall be deemed to constitute a Company Material Adverse Effect or shall be taken into account when determining whether a Company Material Adverse Effect has occurred or would reasonably be expected to occur: (i) any changes in general United States or global economic conditions, (ii) conditions (or changes therein) in any industry in which the Company or its Subsidiaries operate, (iii) general economic, political and/or regulatory conditions (or changes therein), including any changes effecting financial, credit or capital market conditions, including changes in interest or exchange rates, (iv) any change in GAAP or interpretation thereof, (v) any adoption, implementation, promulgation, repeal, modification, amendment, reinterpretation, or other change in any applicable Law of or by any Governmental Entity, (vi) any actions taken, or the failure to take any action, as required by the terms of this Agreement or at the request or with the consent of Buyer and any Effect directly attributable to the announcement of this Agreement and the Transactions, (vii) any failure by the Company to meet any internal or published projections, estimates or expectations of the Company's revenue, earnings or other financial performance or results of operations for any period (it being understood that the facts or occurrences giving rise or contributing to such failure that are not otherwise excluded from the definition of a "Company Material Adverse Effect" may be taken into account), (viii) Effects arising out of changes in geopolitical conditions, acts of terrorism or sabotage, war (whether or not declared), the commencement, continuation or escalation of a war or acts of armed hostility, earthquakes, tornados, hurricanes or other weather conditions or natural calamities or other force majeure events, including any material worsening of such conditions threatened or existing as of the date of this Agreement, (ix) any breach, violation or non-performance by Buyer or any of its Affiliates of any of their obligations under this Agreement, (x) any bankruptcy, insolvency or reorganization of any tenant under any Company Lease or the commencement of any bankruptcy, insolvency or reorganization proceeding with respect to any tenant under any Company Lease, (xi) any Effect relating to Seller or its Affiliates other than the Company and the Company Subsidiaries, including any change in the market price or trading volume of any

securities or Indebtedness of Seller or any of its Affiliates (it being understood that the facts or occurrences giving rise or contributing to such change or any consequence resulting from such change, to the extent attributable to the Company and the Company Subsidiaries and not otherwise excluded from the definition of "Company Material Adverse Effect", may be taken into account) and (xii) changes in reimbursement rates, policies and procedures of Third-Party Payors or Government Entities, *provided*, that, if any Effect described in clauses (i), (ii), (iii), (iv), (v), (viii) and (xii) has had a materially disproportionate adverse impact on the Company relative to other companies of comparable size to the Company operating in the industry or industries in which the Company operates, then the incremental impact of such event shall be taken into account for the purpose of determining whether a Company Material Adverse Effect has occurred.

"<u>Data Tape</u>" means the Microsoft Excel file labeled "NRF HC Datatape by Asset Type_03.31.16" made available by Seller to Buyer prior to the date hereof.

"<u>Debt Financing</u>" means an amount of debt financing in Dollars sufficient for Buyer to pay the Purchase Price and all other amounts to be paid by Buyer at Closing in connection with the consummation of the Transactions.

"<u>Debt Financing Sources</u>" means the lender parties to the Loan Agreements each of whom shall be an internationally recognized banking institution reasonably acceptable to Seller.

"Deposit Amount" means Ten Million Dollars (\$10,000,000).

"<u>Deposit Release Instruction</u>" means, subject to the provisos in <u>Section 9.2(b)</u> and <u>Section 9.2(c)</u>, the joint written instructions of Buyer and Seller instructing the Escrow Agent to distribute the Deposit Amount or a portion thereof to Seller or Buyer, as applicable.

"Determination Time" means 11:59 p.m. Eastern Time on the day immediately preceding the Closing Date; provided, that, if as a result of a breach by the Seller of any of its covenants contained in Section 6.1(d) of this Agreement, Indebtedness, Working Capital or Lender Reserves shall have changed between the Determination Time and the time immediately preceding the effective time of the consummation of the Transactions, then any such changes shall be included in the calculation of Closing Date Indebtedness, Closing Date Working Capital and/or Closing Date Lender Reserves (as the case may be).

"<u>Eligible Property</u>" means any healthcare real estate assets acquired by the Company or a Company Subsidiary during the period between the date hereof and the Closing with the consent of Buyer as required pursuant to <u>Section 6.1(b)</u>.

"Environmental Laws" means any and all applicable Laws existing on the date hereof which (i) regulate or relate to the protection or clean-up of the environment; the use, treatment, storage, transportation, handling, disposal or release of, or exposure to, Hazardous Substances; the preservation or protection of waterways, groundwater, drinking water, air, wildlife, plants or other natural resources; or the protection of the health and safety of employees; or (ii) impose liability or responsibility with respect to any of the foregoing,

including the Comprehensive Environmental Response, Compensation and Liability Act (42 U.S.C. § 9601 et seq.), or any other Law of similar effect.

"<u>Equitable Principles</u>" means (a) bankruptcy, insolvency, reorganization, moratorium and similar Laws affecting creditors' rights and remedies generally and (b) general principles of equity (regardless of whether enforcement is sought in a proceeding at law or in equity).

"ERISA" means the Employee Retirement Income Security Act of 1974.

"<u>Escrow Agent</u>" means a U.S., Hong Kong or Singapore branch of an internationally recognized banking institution, as mutually agreed upon by Seller and Buyer.

"Escrow Agreement" means an escrow agreement by and among Buyer, Seller and the Escrow Agent in form and substance reasonably acceptable to Buyer and Seller pursuant to which the Escrow Agent shall agree to hold the Deposit Amount in Dollars until such time as the Deposit Amount, Termination Fee or Expense Reimbursement Amount becomes payable to Buyer or Seller pursuant to the terms hereof.

"<u>Estimated Total Equity Value</u>" means One Billion, Eight Hundred Twenty-Eight Million, One Hundred Eighty-Eight Thousand, Two Hundred Fifty-Seven Dollars (\$1,828,188,257).

"Expense Reimbursement Amount" means Five Million Dollars (\$5,000,000).

"Fundamental Representations" means the representations and warranties set forth in Sections 3.1 (*Title to Purchased Interests*), 3.2 (*Organization*), 3.3 (*Authorization of Agreement*), 3.4(a)(i) (*Conflicts*), 4.1 (*Organization*), 4.2 (*Authorization of Agreement*), 4.3(a)(i) (*Conflicts*), 4.4 (*Capitalization; Subsidiaries*), 4.11(j) (*Tax Matters*), 4.19 (*Financial Advisor*), 5.1 (*Organization*), 5.2 (*Authorization of Agreement*) and 5.5 (*Financial Advisor*).

"Fraud" means a claim for actual and intentional fraud based on any representation contained in Article III, IV or V of this Agreement and requires (a) a material false statement or that a material omitted fact be necessary in order to make such representation, in the light of the circumstances under which it was made, not misleading, (b) that the party making such representation, through its "Knowledge" individuals, had actual knowledge of the inaccuracy of such material statement or that such material omitted fact was necessary in order to make such representation, in the light of the circumstances under which it was made, not misleading, (c) the party making such representation had the intent to deceive the other party or to induce the other party to enter into this Agreement and (d) the other party acted in reliance on such representation and suffered injury as a result of such reliance. For the avoidance of doubt, "Fraud" does not include any claim for equitable fraud, promissory fraud, unfair dealings fraud, or any torts based on negligence or recklessness.

"Government Sponsored Health Care Program" means any plan or program providing health care benefits, whether directly through insurance or otherwise, that is funded directly, in whole or part, by a Governmental Entity, whether pursuant to one or more contracts

with the applicable Governmental Entity or otherwise, including Medicare, state Medicaid programs, the TRICARE program, Medicare Advantage and managed Medicaid.

"Governmental Entity" means any government or governmental, judicial, administrative or regulatory body thereof, or political subdivision thereof, whether domestic, foreign, federal, national, state, provincial, local or supranational, or any agency, commission, instrumentality or authority thereof, any self-regulatory or quasi-governmental authority or any court, tribunal or arbitrator (public or private).

"Hazardous Substances" means any toxic, carcinogenic, reactive, corrosive, ignitable or flammable chemical or chemical compound, or hazardous or toxic substance, material or waste, or pollutant or contaminant, whether solid, liquid or gas, in each case, that is listed, or is subject to regulation, control or remediation under, any Environmental Laws, including petroleum and petroleum byproducts, asbestos, polychlorinated biphenyls and toxic mold.

"Health Care Laws" means (i) any and all applicable federal, state and local Laws of any applicable Governmental Entity concerning health care or insurance fraud and abuse, including, as applicable, the federal Anti-Kickback Statute (42 U.S.C. § 1320a-7b(b) and 41 U.S.C. §§ 51-58), the civil False Claims Act (31 U.S.C. § 3729 et seq.), the Exclusion Laws (42 U.S.C. §§ 1320a-7 and 1320a-7a), the Program Fraud Civil Remedies Act (31 U.S.C. §§ 3801-3812), the Civil Monetary Penalties Law (42 U.S.C. §§ 1320a and 1320a-7b, and the regulations promulgated pursuant to such statutes; (ii) the federal Food, Drug & Cosmetic Act (21 U.S.C. §§ 301 et seq.), the Federal Health Care Fraud Law (18 U.S.C. § 1347) and all federal and state Laws, as applicable, concerning pharmacology and dispensing medicines or controlled substances, and the regulations promulgated thereunder; (iii) any and all applicable federal, state and local Laws concerning privacy and data security for patient information, including the Health Insurance Portability and Accountability Act of 1996 (42 U.S.C. §§ 1320d-1329d-8), as amended, and all federal and state laws concerning medical record retention, privacy, security, patient confidentiality and informed consent, and the regulations promulgated thereunder; (iv) Medicare (Title XVIII of the Social Security Act), as amended and the regulations promulgated thereunder, including, specifically, conditions of participation for skilled nursing facilities; (v) Medicaid (Title XIX of the Social Security Act) and the regulations promulgated thereunder; (vi) the Medicare Prescription Drug, Improvement, and Modernization Act of 2003 (Pub. L. No. 108-173) and the regulations promulgated thereunder; (vii) the Patient Protection and Affordable Care Act (Pub. L. 111-148) as amended by the Health Care and Education Reconciliation Act of 2010 (Pub. L. 111-152); (viii) quality, safety and accreditation standards and requirements of all applicable state Laws or regulatory bodies; (ix) federal, state and local Laws regulating the ownership, operation or licensure of a health care facility or business, or assets used in connection therewith, including hospitals, skilled nursing facilities, assisted living facilities, independent living facilities and memory care facilities; (x) federal, state and local Laws relating to the provision of management or administrative services in connection with the practice of a health care profession, employment of professionals by non-professionals, professional fee splitting, patient brokering, patient or program charges, claims submission, record retention, certificates of need, certificates of operations and authority; (xi) federal and state Laws with respect to financial relationships between referral sources and referral recipients, including, but not limited to the federal Stark Law (42 U.S.C. 1395nn et. seq.) and the

regulations promulgated thereunder; (xii) life safety codes; and (xiii) the Laws of any foreign jurisdiction in which the Company or its Subsidiaries operate that are analogous to those Laws identified in items (i) through (xii) above.

"Income Taxes" means all Taxes based upon, measured by, or calculated with respect to (a) gross or net income or gross or net receipts or profits, including any capital gains and minimum Taxes; and (b) multiple bases, including corporate franchise, doing business or occupation Taxes, if one or more of the bases upon which such Tax may be based upon, measured by, or calculated with respect to, is described in clause (a) above.

"Indebtedness" means, as to any Person, (a) all obligations of such Person for borrowed money (including reimbursement and all other obligations with respect to any surety bonds, letters of credit, bankers' acceptances, in each case, only to the extent drawn, and the Series B Cumulative Redeemable Preferred Interests of NorthStar Realty Healthcare, LLC, whether or not matured), (b) all obligations of such Person evidenced by notes, bonds, debentures or similar instruments, (c) all obligations of such Person to pay the deferred purchase price of property or services, except trade accounts payable and accrued commercial or trade liabilities arising in the ordinary course of business, (d) all interest rate and currency swaps, caps, collars and similar agreements or hedging devices under which payments are obligated to be made by such Person, whether periodically or upon the happening of a contingency (valued at the termination value thereof), (e) all indebtedness created or arising under any conditional sale or other title retention agreement with respect to property acquired by such Person, (f) all obligations of such Person in respect of leases of (or other agreements conveying the right to use) property or other assets which GAAP requires to be classified and accounted for as capital leases as of the date of this Agreement, (g) all indebtedness secured by any Lien on any property or asset owned or held by such Person, and (h) all guarantees by such Person of the Indebtedness of any other Person; provided, however, that Indebtedness shall not include intercompany obligations between the Company and any Company Subsidiary.

"Intellectual Property" means all intellectual property, including trademarks, service marks, trade names, any goodwill associated with the foregoing, Internet domain names, patents, copyrights, trade secrets, and rights in inventions and know-how, and all registrations and applications for any of the foregoing.

"IRS" means the Internal Revenue Service.

"JV Letter Agreement" means the letter agreement to be entered into by and between Buyer and NRF OP SPV on the Closing Date, substantially in the form of Exhibit I hereto.

"Knowledge" means the actual knowledge, after making reasonable inquiry, of (a) Jonathan A. Langer, Robert Gatenio, Daniel D. Raffe, Ronald J. Jeanneault and Ann B. Harrington, in the case of the Seller Group or any member thereof or (b) Zuyu Tan, Xiaochen Li, Tim Zhang and Lin Xu, in the case of Buyer.

"<u>Law</u>" means all foreign, federal, national, supranational, state, provincial and local laws, statutes, codes, ordinances, rules, regulations, resolutions, Orders and rules of law (including common law).

"<u>Lender Reserves</u>" means amounts held in escrow or a segregated account pursuant to the requirements of Indebtedness for borrowed money of the Company or the Company Subsidiaries, as determined on a consolidated basis and in accordance with the Sample Adjustment Calculation and Adjustment Principles.

"<u>Letter of Credit</u>" means a letter of credit issued by the Debt Financing Sources in the amount of Ten Million Dollars (\$10,000,000), in form and substance reasonably acceptable to Buyer and Seller, pursuant to which Seller shall be permitted to draw on in the event the Deposit Amount, Termination Fee or Expense Reimbursement Amount becomes payable to Seller pursuant to the terms hereof.

"<u>Licensing Survey</u>" means any federal Statement of Deficiencies and Plan of Correction and any similar investigation, survey, inspection and plan of correction undertaken or issued by a Governmental Entity and related to the ownership or operation of a Property.

"<u>Lien</u>" means any lien, pledge, mortgage, hypothecation, deed of trust, security interest, claim, lease, license, charge, option, right of first refusal, easement, adverse claim, reversion, servitude, transfer restriction, voting restriction, encumbrance or other similar or dissimilar prohibition, restriction or limitation, including pursuant to any Order.

"Loan Agreements" means one or more loan agreements to be entered into by Buyer and the Debt Financing Sources in connection with the Debt Financing, in form and substance reasonably acceptable to Seller, which shall (a) provide for a principal amount of not less than an amount sufficient to pay the Purchase Price and all other amounts to be paid by Buyer at Closing in connection with the consummation of the Transactions, (b) contain customary conditions to funding, (c) include a commitment of the Debt Financing Sources to fund the Debt Financing upon satisfaction of the conditions to funding, which commitment shall not expire prior to the Outside Date and (d) shall not impose any liabilities or obligations on Seller, the Company or any of their respective Affiliates.

"Loan Documents" means the Loan Agreements and all exhibits, schedules and annexes thereto.

"Material Casualty" means a Casualty for which the aggregate cost to repair the related damage is reasonably expected by Seller to be Five Million Dollars (\$5,000,000) or more in the aggregate.

"<u>Material Condemnation</u>" means a Condemnation for which the condemnation award reasonably expected by Seller is Five Million Dollars (\$5,000,000) or more in the aggregate.

"MOB Sale" means the sale of a portfolio of medical office buildings as described in further detail on Schedule 1.1(a).

- "NDRC" means the National Development and Reform Commission of the PRC or its competent local counterparts.
- "NRF" means NorthStar Realty Finance Corp.
- "NSAM" means NorthStar Asset Management Group Inc.
- "Option Sale" means any sale of a Property made following the exercise of a purchase option pursuant to the agreement set forth on Schedule 1.1(b).
- "Order" means any order, injunction, judgment, decree, stipulation, determination, ruling, writ, assessment or arbitration or other award of a Governmental Entity.
- "Organizational Documents" means, as to any Person, its (a) certificate or articles of incorporation, or similar corporate or other instruments of organization, (b) articles of association, by-laws or other similar instruments, and (c) shareholder agreements, limited partnership agreements, limited liability company agreements or operating agreements and other similar governing corporate documents.
- "<u>Permit</u>" means any approvals, authorizations, consents, licenses, permits, waivers, certifications, franchises, certificates, variances, registrations or other similar authorization issued, or otherwise granted, by a Governmental Entity.

"Permitted Liens" means (a) all Liens disclosed in policies of title insurance, (b) Liens for Taxes, assessments or other governmental charges not yet due, payable or delinquent (or which may be paid without interest or penalties) or the amount or validity of which is being contested in good faith by appropriate proceedings for which the Company or one of its Subsidiaries has established adequate reserves to the extent required by GAAP, (c) mechanics', carriers', workers', repairers', and similar Liens arising or incurred in the ordinary course of business or the amount or validity of which is being contested in good faith by appropriate proceedings for which the Company or one of its Subsidiaries has established adequate reserves to the extent required by GAAP, (d) pledges, deposits or other Liens to the performance of bids, trade contracts (other than for borrowed money), leases or statutory obligations, (e) zoning, entitlement and other land use and environmental regulations by any Governmental Entity, (f) minor survey exceptions and matters as to the Properties which would be disclosed by an accurate survey or inspection of such real property and which do not materially impair the current occupancy or current use of such Property, (g) any Lien affecting the fee interest of any Property created by the applicable tenant(s) of such fee interest, (h) title of a lessor under any lease, (i) any Liens discharged or released at or in connection with Closing, (j) any lien, pledge or security interest granted in connection with any Indebtedness for borrowed money of the Company or any Company Subsidiary and (k) such other imperfections in title, charges, easements, restrictions and encumbrances which do not materially impair the current occupancy or current use of such Property.

"<u>Person</u>" means any individual, corporation, partnership, limited liability company, limited liability partnership, firm, joint venture, association, joint-stock company, trust, unincorporated organization, Governmental Entity or other entity.

- "PRC" means the People's Republic of China (excluding Taiwan, and the Hong Kong and the Macau Special Administrative Regions).
- "PRC Approvals" means the filings with and/or approvals of CIRC and NDRC and the completion of the outbound investment related foreign exchange filing with the relevant bank authorized by SAFE required with respect to the consummation of the Transactions.
- "<u>Pre-Closing Documents</u>" means the documents, agreements and instruments necessary to effect the Pre-Closing Restructuring, each of which shall be in form and substance reasonably acceptable to Buyer.
- "<u>Pre-Closing Restructuring</u>" means the transactions set forth on <u>Exhibit A</u>, as may be revised from time to time by agreement of the Parties prior to the Closing.
- "<u>Pre-Closing Tax Period</u>" means any Tax period that ends on or prior to the Closing Date and that portion of any Straddle Period that ends on the Closing Date.
- "<u>Representatives</u>" means the officers, directors, employees, accountants, consultants, legal counsel, financing sources, agents and other representatives of a Person.
 - "SAFE" means the State Administration of Foreign Exchange of the PRC or its competent local counterparts.
- "Side Letters" means those certain letter agreements entered into by the Parties concurrently with the execution of this Agreement.
 - "Straddle Period" means any Tax period that begins on or prior to the Closing Date and ends after the Closing Date.
- "Subsidiary" means, with respect to any specified Person, any (i) entity of which such specified Person directly or indirectly owns or has the power to vote shares of any capital stock or other ownership interests (x) having voting power to elect a majority of the directors of such corporation or other Persons performing similar functions for such entity or (y) representing (I) 50% or more of the ownership interests in such entity (by value) or (II) an interest in 50% or more of the capital or profits of such entity, or (ii) with respect to which such specified Person acts as sole managing member or sole general partner or in a similar capacity. For the avoidance of doubt, each of the Company and the Company Subsidiaries is, and following the Pre-Closing Restructuring Newco will be, a Subsidiary of JV, and NRF OP SPV is a Subsidiary of Seller.
- "Tax" or "Taxes" means any federal, state, local or foreign taxes, including all net income, gross receipts, license, payroll, employment, excise, severance, escheat or abandoned property, stamp, occupation, premium, windfall profits, environmental, customs duties, equity, franchise, profits, withholding, social security, unemployment, disability, real property, personal property, sales, use, transfer, registration, value added, alternative or add-on minimum, or other tax of any kind whatsoever, including any interest, penalty or addition thereto imposed by a Taxing Authority.

"<u>Taxing Authority</u>" means the IRS or any other Governmental Entity having or purporting to exercise jurisdiction with respect to any Tax.

"<u>Tax Returns</u>" means any and all reports, returns, declarations, claims for refund, elections, disclosures, estimates, or information reports required to be supplied to a Taxing Authority in connection with Taxes, including any schedule or attachment thereto or amendment thereof.

"Termination Fee" means Ten Million Dollars (\$10,000,000).

"<u>Third Party Payor</u>" means any Government Sponsored Health Care Program, insurer, health benefit plan, health maintenance organization, preferred provider organization, employer-sponsored health plan, multi-employer welfare trust, or any other managed care program or third party payor, including any fiscal intermediary or contractor of any of the foregoing.

"<u>Transaction Documents</u>" means the Side Letters, the Escrow Agreement (if applicable), the Letter of Credit (if applicable), the Newco LLC Agreement, the JV LLC Agreement, the JV Letter Agreement, the Pre-Closing Documents and the other documents, certificates and instruments to be executed and delivered to consummate the Transactions.

"<u>Transactions</u>" means the sale and purchase of the Purchased Interests and the performance of the related covenants and agreements contemplated by this Agreement.

"<u>U.S. Tax Person</u>" means a U.S. person (as defined in Section 7701(a)(30) of the Code).

"Willful Breach" means with respect to any breaches or failures to perform any of the covenants or other agreements contained in this Agreement, a material breach that is a consequence of an act or failure to act undertaken by the breaching Person with actual or constructive knowledge (which shall be deemed to include knowledge of facts that a Person acting reasonably should have, based on reasonable inquiry) that such Person's act or failure to act would, or would reasonably be expected to, result in or constitute a breach of this Agreement.

"Working Capital" means the amount (which may be positive or negative) equal to (a) the consolidated current assets of the Company and the Company Subsidiaries, which current assets shall include only the current assets set forth on Exhibit B under the heading "Current Assets" and no other assets minus (b) the consolidated current liabilities of the Company and the Company Subsidiaries, which current liabilities shall include only the current liabilities set forth on Exhibit B under the heading "Current Liabilities" and no other liabilities, in each case of clauses (a) and (b) determined in a manner consistent with the Sample Adjustment Calculation and Adjustment Principles. Notwithstanding anything to the contrary contained herein, in no event shall "Working Capital" include any amount (i) with respect to Indebtedness, (ii) included in the determination of Lender Reserves, (iii) in respect of any liability for any dividends or other distributions with respect to the Newco Interests or any equity interests of the Company or the Subsidiaries of the Company that have been declared but are not yet due and payable as of the Closing Date, or any cash reserved or set aside for purposes of effecting such dividends or distributions, (iv) with respect to any insurance proceeds received or

receivable by the Company or any Company Subsidiary in respect of any Casualty or any liability for the repair or replacement of the applicable Property, (v) with respect to proceeds received by the Company or a Company Subsidiary, or liabilities incurred or payable by the Company or a Company Subsidiary in connection with the sale or Condemnation of a Property prior to Closing or (vi) if either of Estimated Working Capital and Closing Date Working Capital exceed Twelve Million Dollars (\$12,000,000), then the lesser of (A) the amount of such excess and (B) the cash or cash equivalents attributable to such excess.

1.2 <u>Terms Defined Elsewhere</u>. The following terms are defined in the following Sections of this Agreement:

<u>Term</u>	Section
Accountants	2.4(e)(ii)
Adjustment Notice	2.4(c)
Adjustment Principles	2.4(b)
Agreement	Preamble
Antitrust Laws	6.2(b)
Arbitrator	10.1(a)
Asset Management Material Contracts	4.16(a)(xii)
Audited Financial Statements	4.5(a)
Base Amount	9.2(d)
Buyer	Preamble
Buyer Capital Advance	2.9(a)
Buyer Indemnified Parties	8.2
Buyer Objection Notice	2.4(d)
Buyer Review Period	2.4(d)
Cap	8.4
Capital Contribution Advance Guarantee	2.9(b)
CFIUS	6.2(d)(iii)
Claim Notice	8.5(a)
Closing	7.1
Company	Preamble
Company Financial Statements	4.5(b)
Company Interests	4.4(c)
Company Leases	4.10(f)
Company Permits	4.8(b)
Company Subsidiaries	Recitals
Company Subsidiary	Recitals
Company Third Party	4.10(j)
Company Title Insurance Policies	4.10(j)
Company Title Insurance Policy	4.10(j)
Condemnation	4.13
Confidentiality Agreement	6.5
Contract	4.16(a)
Conveyance Taxes	6.6
De Minimis Claims	8.4

<u>Term</u>	<u>Section</u>
Disclosure Schedule	10.8
D-REIT	3.6
Equity Value Adjustment	2.4(a)
Estimated Adjustment Statement	2.4(b)
Estimated Indebtedness	2.4(b)
Estimated Lender Reserves	2.4(b)
Estimated Working Capital	2.4(b)
Existing Environmental Reports	4.15(a)
Extended Closing Date	Definition of Closing Date
Fee Waiver Value	2.4(g)(vi)
Final Adjustment Date	2.4(f)
GAAP	2.4(b)
Governmental Approval	3.4(b)
Griffin Mortgage	4.5(a)
Healthcare Portfolio	Recitals
ICC	10.1(a)
Indemnified Party	8.5(a)
Indemnifying Party	8.5(a)
Initial Closing Date	Definition of Closing Date
Initial Distribution Date	2.4(f)(i)
Intended Tax Treatment	2.6
JV	Preamble
JV HoldCo Interests	Recitals
JV LLC Agreement	2.5(b)
JV OpCo	Preamble
JV OpCo Interests	4.4(a)
Legal Proceeding	3.5
Losses	8.2
Material Company Leases	4.10(g)
Material Contracts	4.16(a)
Newco	Preamble
Newco Interests	4.4(a)
Newco LLC Agreement	2.5(a)
NRF OP	Preamble
NRF OP SPV	Preamble
Operator	4.8(c)
Outside Date	9.1(b)
Parties	Preamble
Party	Preamble
Property	Recitals
Purchase Price	2.2(a)
Purchased Interests	Recitals
Qualifying Income	9.2(d)
REIT	2.6
REIT Requirements	9.2(d)

<u>Term</u>	Section
Rent Roll Summary	4.10(f)
Resolution Period	2.4(e)(i)
Rules	10.1(a)
Sample Adjustment Calculation	2.4(b)
Seller	Preamble
Seller Group	Preamble
Seller Indemnified Parties	8.3
Survival Date	8.1
Tax Guidance	9.2(d)
Threshold	8.4
Total Equity Value	2.4(a)
Unaudited Financial Statements	4.5(b)

- 1.3 <u>Interpretation</u>. Unless the express context otherwise requires:
- (a) the words "hereof", "herein", and "hereunder" and words of similar import, when used in this Agreement, shall refer to this Agreement as a whole and not to any particular provision of this Agreement;
 - (b) terms defined in the singular shall have a comparable meaning when used in the plural, and vice versa;
- (c) the terms "Dollars" and "\$" mean United States Dollars and the terms "Renminbi" and "\pmax" mean the official currency of the PRC;
- (d) references herein to a specific Section, Subsection, Recital, Schedule or Exhibit shall refer, respectively, to Sections, Subsections, Recitals, Schedules or Exhibits of this Agreement;
- (e) wherever the word "include", "includes", or "including" is used in this Agreement, it shall be deemed to be followed by the words "without limitation";
 - (f) references herein to any gender shall include each other gender;
- (g) references herein to any Person shall include such Person's heirs, executors, personal representatives, administrators, successors and assigns; *provided*, *however*, that nothing contained in this clause (g) is intended to authorize any assignment or transfer not otherwise permitted by this Agreement; *provided*, *further*, that, for the avoidance of doubt, references to Seller shall include any Subsidiary of Seller to whom Seller sells, assigns or transfers all or a portion of its JV HoldCo Interests and its rights hereunder with respect thereto in accordance with Section 10.6(c);
- (h) references herein to a Person in a particular capacity or capacities shall exclude such Person in any other capacity;

- (i) references herein to any contract (including this Agreement) mean such contract as amended, supplemented or modified from time to time in accordance with the terms thereof;
- (j) with respect to the determination of any period of time, the word "from" means "from and including" and the words "to" and "until" each means "to but excluding";
- (k) references herein to any Law or any Permit mean such Law or Permit as amended, modified, codified, reenacted, supplemented or superseded in whole or in part, and in effect from time to time;
- (l) references herein to any Law shall be deemed also to refer to all rules and regulations promulgated thereunder;
- (m) any item shall be considered "made available" to Buyer, to the extent such phrase appears in this Agreement, if such item has been posted by the Seller Group or its Representatives in the electronic data room established in connection with the Transactions at least two (2) Business Days prior to the date hereof; and
- (n) the Parties agree that they have been represented by counsel during the negotiation and execution of this Agreement and, therefore, waive the application of any Law, regulation, holding or rule of construction providing that ambiguities in an agreement or other document will be construed against the party drafting such agreement or document.

ARTICLE II

PURCHASE AND SALE OF INTERESTS; PURCHASE PRICE

2.1 <u>Sale and Purchase of Purchased Interests</u>. Upon the terms and subject to the conditions of this Agreement, on the Closing Date, Seller shall, or shall cause its Affiliates to, sell, transfer, assign, convey and deliver, to Buyer, and Buyer shall purchase from Seller (directly or indirectly), the Purchased Interests, which shall represent indirect ownership of the Buyer Ownership Percentage of the Company Interests, free and clear of all Liens other than Liens imposed by Buyer or restrictions on transfer under applicable securities Laws, the JV LLC Agreement, the Newco LLC Agreement or the JV Letter Agreement.

2.2 Purchase Price.

- (a) Subject to the adjustments set forth in <u>Section 2.4</u>, the aggregate consideration for the Purchased Interests at the Closing shall be Three Hundred Thirty Million Dollars (\$330,000,000) (the "<u>Purchase Price</u>").
- (b) Within five (5) Business Days (*provided*, that if Buyer is using commercially reasonable efforts to deliver a fully executed Letter of Credit to Seller or to deliver the Deposit Amount to the Escrow Agent, such period may be extended by Buyer for one (1) additional five (5) Business Day period) of Buyer's execution and delivery of the Loan Agreements, at Buyer's election, (i) Buyer shall deliver or cause to be delivered to Seller the

fully executed Letter of Credit or (ii) the parties to the Escrow Agreement shall execute and deliver the Escrow Agreement and, simultaneously with such execution and delivery, Buyer shall deliver the Deposit Amount to the Escrow Agent to be held and disbursed by the Escrow Agent in accordance with the terms and provisions of the Escrow Agreement.

- (c) At the Closing, (i) Buyer shall pay or cause to be paid the Purchase Price by wire transfer of immediately available funds in Dollars to Seller to the Closing Account and (ii) in the event that the Deposit Amount is not used to fund Buyer's obligations in the foregoing subclause (i), Buyer and Seller shall jointly instruct the Escrow Agent to release the Deposit Amount to Buyer.
- 2.3 <u>Escrow Arrangements</u>. In the event that Buyer exercises its option to deliver the Deposit Amount to the Escrow Agent under <u>Section 2.2(b)(ii)</u>, the Deposit Amount shall be delivered to and held and disbursed by the Escrow Agent in accordance with the terms and provisions of the Escrow Agreement. The Deposit Amount shall not be refundable to Buyer, except as provided in <u>Sections 9.2(b)(ii)</u>, <u>9.2(b)(iii)</u> and <u>9.2(b)(iii)</u> or simultaneously with Closing in accordance with <u>Section 2.2(c)</u>.

2.4 Adjustments to Equity Value.

- (a) The Estimated Total Equity Value shall be adjusted at Closing as follows (each such adjustment, an "Equity Value Adjustment", and the Estimated Total Equity Value as so adjusted at Closing, the "Total Equity Value"):
- (i) The Estimated Total Equity Value shall be increased by the amount of Estimated Working Capital;
- (ii) If the sale, including as a result of a Condemnation or the MOB Sale, of a Property is consummated after the date hereof but before the Closing Date, the Estimated Total Equity Value shall be decreased by the Adjustment Amount for such Property;
- (iii) If the purchase of an Eligible Property is consummated after the date hereof but before the Closing Date, the Estimated Total Equity Value shall be increased by the Adjustment Amount for such Eligible Property;
- (iv) The Estimated Total Equity Value shall be (A) decreased by the amount, if any, equal to the excess of the Estimated Indebtedness over the Company Indebtedness Estimate Amount or (B) increased by the amount, if any, equal to the excess of the Company Indebtedness Estimate Amount over the Estimated Indebtedness;
- (v) The Estimated Total Equity Value shall be increased by the amount, if any, of Estimated Lender Reserves;
- (vi) The Estimated Total Equity shall be (A) decreased by the amount, if any, of any negative adjustment to the Allocated Purchase Price (JV Interest) or (B) increased by the amount, if any, of any positive adjustment to the Allocated Purchase Price (JV Interest), in each case, as a result of the re-valuation of Properties located in the United Kingdom based on the Closing Date Exchange Rate in accordance with the terms hereof; and

(vii) The Estimated Total Equity Value shall be decreased by One Million Dollars (\$1,000,000).

Notwithstanding the foregoing, the Parties agree that (i) to the extent that any of the Equity Value Adjustments arise from or out of any facts, circumstances or conditions or events that overlap, there shall be no duplication of counting toward the calculation of any such Equity Value Adjustments and (ii) prior to each Equity Value Adjustment finally determined pursuant to this <u>Section 2.4</u>, Seller shall contribute to JV in exchange for JV HoldCo Interests a number of Newco Interests that is sufficient to cause Seller's ownership percentage of JV not to be less than 53% (taking into account such Equity Value Adjustment).

- (b) Estimated Adjustment Statement. No later than five (5) Business Days prior to the Closing Date, Seller shall prepare in good faith and deliver or cause to be delivered to Buyer (x) (i) an updated version of Exhibit H reflecting any revaluation of Properties located in the United Kingdom in accordance with the terms hereof and (ii) the updated Company Indebtedness Estimate Amount reflecting any re-valuation of the Company Indebtedness Estimate Amount in accordance with the terms hereof, and (y) an estimate of (i) Closing Date Working Capital (the "Estimated Working Capital"), (ii) Closing Date Indebtedness (the "Estimated Indebtedness"), (iii) Closing Date Lender Reserves (the "Estimated Lender Reserves") and (iv) any Adjustment Amount(s) (such statement, the "Estimated Adjustment Statement"), in the case of each of subclauses (i)-(iii) as of the Determination Time and prepared (A) in accordance with United States generally accepted accounting principles ("GAAP") and (B) on a basis consistent with the sample calculation of Working Capital, Indebtedness and Lender Reserves (the "Sample Adjustment Calculation") set forth on Exhibit B hereto together with any accounting principles, procedures, policies and methods also set forth on such exhibit (collectively, the "Adjustment Principles"); provided, that, in the event of any inconsistency among the foregoing, Exhibit B shall dictate the treatment of such inconsistency followed by GAAP. The Estimated Adjustment Statement will be accompanied by such backup materials and schedules as are reasonably necessary to support the accuracy of the information contained therein. To the extent reasonably requested by Buyer, at Buyer's expense (limited, with respect to Seller, to Seller's reasonable and documented out-of-pocket expenses), Seller will provide reasonable access to members of the Seller Group's accounting and financial staff in connection with Buyer's review of the Estimated Adjustment Statement. Seller will review any comments proposed by Buyer with respect to the Estimated Adjustment Statement and will consider, in good faith, any appropriate changes. Buyer and Seller acknowledge and agree that (A) the exercise by Buyer of the foregoing review right, and any subsequent consultation with Seller in respect thereto, shall not delay or postpone the occurrence of the Closing and, (B) if the Seller and Buyer are unable to agree on any amount set forth in the Estimated Adjustment Statement, without limiting Buyer's rights under <u>Section 2.4(d)</u>, the amount proposed by the Seller shall be used for purposes of the Estimated Adjustment Statement.
- (c) <u>Post-Closing Adjustment</u>. Within sixty (60) calendar days following the Closing Date, Seller shall prepare in good faith and deliver or cause to be delivered to Buyer a notice (the "<u>Adjustment Notice</u>") containing Seller's calculation of (i) Closing Date Working Capital, (ii) Closing Date Indebtedness, (iii) Closing Date Lender Reserves, and (iv) any Adjustment Amount(s), in the case of each of subclauses (i)-(iii) as of the Determination

Time and prepared (A) in accordance with GAAP and (B) on a basis consistent with the Sample Adjustment Calculation and the Adjustment Principles; *provided*, that, in the event of any inconsistency among the foregoing, <u>Exhibit B</u> shall dictate the treatment of such inconsistency followed by GAAP. The Adjustment Notice will be accompanied by such backup materials and schedules as are reasonably necessary to support the accuracy of the information contained therein.

- (d) Within sixty (60) calendar days after the delivery to Buyer of the Adjustment Notice (the "Buyer Review Period"), Buyer shall notify Seller of its agreement or disagreement with the Adjustment Notice. If Buyer in good faith disagrees with the Adjustment Notice, Buyer shall deliver to Seller, prior to the expiration of the Buyer Review Period, a notice (the "Buyer Objection Notice") setting forth (i) the items or amounts with which Buyer disagrees and the basis for such disagreement and (ii) Buyer's proposed corrections to the Adjustment Notice. If Buyer does not deliver a Buyer Objection Notice or if Buyer agrees in writing with the Adjustment Notice, Buyer shall be deemed to agree in all respects with the Adjustment Notice and the items and amounts reflected thereon shall be final and binding upon the Parties. During the Buyer Review Period, the Seller Group shall promptly comply with reasonable written requests by Buyer provided in advance for access, during normal business hours, to each member of the Seller Group's and its respective Affiliates' business, accountants and personnel involved in, and to the supporting information, books and records utilized in, the preparation of the Estimated Adjustment Statement and Adjustment Notice and shall cause each member of the Seller Group's and its respective Affiliates' accountants and personnel to reasonably cooperate with Buyer's review of such items and amounts; provided, that, (A) accountants shall not be obligated to make any work papers (to the extent extant) available to Buyer unless and until Buyer has signed a customary agreement relating to such access to work papers in form and substance reasonably acceptable to such accountants, and (B) Seller shall not be required to provide access to any information to the extent such access could result in the waiver of any legal privilege or work product protection of the Seller or the Company.
 - (e) If a Buyer Objection Notice is delivered in accordance with <u>Section 2.4(d)</u>, the Parties agree that:
- (i) During the thirty (30) calendar day period following the delivery of the Buyer Objection Notice (the "Resolution Period"), Buyer and Seller shall seek in good faith to resolve in writing any disagreement between them with respect to the matters set forth in the Buyer Objection Notice;
- (ii) If, at the end of the Resolution Period, Buyer and Seller have not resolved in writing the matters specified in the Buyer Objection Notice, Buyer and Seller shall jointly refer in writing to the New York City office of Deloitte & Touche LLP (or a firm of accountants of nationally recognized standing reasonably satisfactory to Buyer and Seller) (the "Accountants") for resolution of only those matters in the Buyer Objection Notice which remain unresolved as of the end of the Resolution Period. In performing their review, the Accountants shall apply the Adjustment Principles and determine the accurate application of the Adjustment Principles to only those unresolved items or amounts in the Adjustment Notice as to which Buyer has, in the Buyer Objection Notice, disagreed and such other issues as may be reasonably affected by the items to which Buyer has so disagreed. The Accountants shall be

required to deliver to Buyer and Seller, as promptly as practicable, but no later than thirty (30) calendar days after the Accountants are engaged, a written report setting forth their resolution and, if applicable, their calculation of the disputed items or amounts. In no event shall the Accountants assign a value greater than the greatest value for such item claimed by either Seller in the Adjustment Notice or Buyer in the Buyer Objection Notice or smaller than the smallest value claimed by either Seller in the Adjustment Notice or Buyer in the Buyer Objection Notice. Buyer and the Seller Group shall promptly comply with reasonable requests by the Accountants for information, books, records and similar items; and

- (iii) Upon delivery of the Accountants' written report, such report and the calculations set forth therein shall be final and binding upon Buyer and Seller, and in connection with such Accountants' report, no party shall seek recourse to the courts, other tribunals or venues pursuant to this Agreement or otherwise, other than to collect any amounts due under this Section 2.4. The fees, costs and expenses of the Accountants shall be allocated between the Buyer and Seller based on the inverse of the percentage the Accountants' determination bears to the total amount of the total items in dispute as originally submitted to the Accountants. For example, if the items in dispute equal \$1,000 and the Accountants award \$600 in favor of Seller's position, 60% of the fees, costs and expenses of the Accountants' review would be borne by Buyer and 40% of the fees, costs and expenses would be borne by Seller.
- (iv) The Accountants shall act as experts, not as arbitrators, in resolving such disputed items and amounts. The proceeding before the Accountants shall be an expert determination under applicable Laws governing expert determination and appraisal proceedings. All communications between Seller and Buyer, on the one hand, and the Accountants, on the other hand, shall be in writing with copies simultaneously delivered to the non-communicating party.
- (f) <u>Final Adjustment</u>. Upon the final determination of the Closing Date Working Capital, Closing Date Indebtedness, Closing Date Lender Reserves and the Adjustment Amounts pursuant to this <u>Section 2.4</u> and the final determination of Total Equity Value (the date of such final determinations, the "<u>Final Adjustment Date</u>"):
- (i) Buyer agrees that Newco shall, on the first date on which Newco makes a distribution pursuant to Article 4 of the Newco LLC Agreement following the Final Adjustment Date (the "Initial Distribution Date"), distribute to or for the benefit of Seller an amount of distributions otherwise payable indirectly to Buyer through JV OpCo and JV (other than Special Tax Distributions (as defined in the Newco LLC Agreement) and, unless otherwise elected by Buyer, distributions in respect of Preferred Return (as defined in the Newco LLC Agreement) to JV OpCo) pursuant to Article 4 of the Newco LLC Agreement, an amount equal to the Buyer Ownership Percentage of the amount, if any, by which, the Total Equity Value as finally determined pursuant to this Section 2.4 exceeds the Estimated Total Equity Value, provided, that, if Newco does not distribute such amount to or for the benefit of Seller in full on the Initial Distribution Date, any amount not distributed shall accrue interest at a rate of 7.50% per year from such date until the date such amount (together with any interest accrued thereon) is finally distributed to or for the benefit of Seller;

(ii) Seller agrees that Newco shall, on the <u>Initial Distribution Date</u>, distribute to JV OpCo for the indirect benefit of Buyer an amount of distributions otherwise payable to Seller (other than Special Tax Distributions (as defined in the Newco LLC Agreement), unless otherwise elected by Seller, and distributions in respect of Preferred Return (as defined in the Newco LLC Agreement) to Seller), directly or indirectly, pursuant to Article 4 of the Newco LLC Agreement, an amount equal to the Buyer Ownership Percentage of the amount, if any, by which Total Equity Value as finally determined pursuant to this <u>Section 2.4</u> is less than the Estimated Total Equity Value, *provided*, that, if Newco does not distribute such amount for the indirect benefit of Buyer in full on the Initial Distribution Date, any amount not distributed shall accrue interest at a rate of 7.50% per year from the Initial Distribution Date until the date such amount (together with any interest accrued thereon) is finally distributed for the indirect benefit of Buyer.

Buyer, Seller and each of their respective Affiliates shall, for all Income Tax purposes, treat amounts distributed under Section 2.4(f)(i) to or for the benefit of Seller as having been (i) distributed by Newco to JV OpCo in accordance with Article 4 of the Newco LLC Agreement with respect to Newco interests owned indirectly by Buyer, (ii) distributed by JV OpCo to JV, (iii) distributed by JV to Buyer in accordance with the JV LLC Agreement, and (iv) then paid by Buyer to Seller as an adjustment of the Purchase Price hereunder. Buyer, Seller and each of their respective Affiliates shall, for all U.S. federal, state and local income tax purposes, treat amounts distributed under Section 2.4(f)(ii) to JV OpCo for the indirect benefit of Buyer as having been (i) distributed by Newco to Seller, directly or indirectly, with respect to Newco Interests not owned by JV OpCo in accordance with Article 4 of the Newco LLC Agreement, and (ii) then paid by Seller to Buyer as an adjustment of the Purchase Price hereunder.

- 2.5 <u>Governing Documents</u>. At the Closing, the limited liability company agreements of Newco and JV shall be amended and restated such that:
- (a) the form of limited liability company agreement of Newco attached hereto as <u>Exhibit C</u>, shall be the limited liability company agreement of Newco (the "<u>Newco LLC Agreement</u>"), until thereafter amended in accordance with applicable Law and the applicable provisions of such limited liability company agreement; and
- (b) the form of limited liability company agreement of JV attached hereto as <u>Exhibit D</u>, shall be the limited liability company agreement of JV (the "<u>JV LLC Agreement</u>"), until thereafter amended in accordance with applicable Law and the applicable provisions of such limited liability company agreement.
- 2.6 <u>Tax Treatment</u>. The Parties intend that (a) in the event that the Closing occurs in 2016, (i) the formation of JV by Seller, (ii) the contribution by Seller of the JV OpCo Interests to JV in exchange for the JV HoldCo Interests (including the Purchased Interests), and (iii) the purchase and sale of the Purchased Interests hereunder, shall be treated as an integrated transaction immediately after which Seller is not in control (as defined in section 368(c) of the Code) of JV and thus the transaction described in subclause (ii) of this clause (a) is a taxable exchange in which Seller recognizes gain under Section 1001(a) of the Code and (b) in the event that Seller exercises its rights under <u>Section 10.6(c)</u> to sell, assign or transfer all or a portion of its JV HoldCo Interests to a wholly-owned Subsidiary that is not a disregarded entity for U.S.

federal, state and local income Tax purposes, (i) the formation of JV by NRF OP (directly or indirectly), (ii) the contribution by NRF OP of the JV OpCo Interests to JV in exchange for the JV HoldCo Interests (including the Purchased Interests) and (iii) the sale by NRF OP of the Purchased Interests to an Affiliate pursuant to Section 10.6(c), shall be treated as an integrated transaction immediately after which NRF OP is not in control (as defined in Section 368(c) of the Code) of JV) and thus the transaction described in subclause (ii) of this clause (b) is a taxable exchange in which Seller recognizes gain under Section 1001(a) of the Code (the "Intended Tax Treatment"). The Parties shall file all Tax Returns consistent with the Intended Tax Treatment and shall take no position inconsistent therewith with any Taxing Authority unless otherwise required by a change in applicable Law or a final determination of a Taxing Authority. For its taxable year in which the Pre-Closing Restructuring occurs, each entity classified as a partnership for U.S. federal income tax purposes in which the Company directly or indirectly (other than through a REIT) holds a partnership interest will have a valid election under Section 754 of the Code in effect. JV shall make a timely election to be treated as a real estate investment trust within the meaning of Section 856 of the Code ("REIT") for the taxable year of JV beginning on the date of its formation.

- 2.7 Allocation of Purchase Price. Buyer and Seller agree that for federal, state and local income Tax purposes, the fair market value of the JV HoldCo Interests (including the Purchased Interests) and other relevant items which are treated as consideration received by Seller for federal, state and local income Tax purposes in the exchange of JV OpCo Interests for JV HoldCo Interests pursuant to the Pre-Closing Restructuring shall be allocated amongst the assets of the Company in a manner determined by Seller in accordance with Section 1060 of the Code and the U.S. Treasury Regulations promulgated thereunder. Seller shall provide Buyer a copy of an allocation statement setting forth such allocation and the allocation, in accordance with Treasury Regulation section 1.755-1, of basis adjustments under section 743(b) of the Code amongst the property of each partnership in which the Company holds a partnership interest within thirty (30) days of the final determination of the adjustments, if any, pursuant to Section 2.4.
- 2.8 <u>Withholding</u>. Buyer shall be entitled to deduct and withhold from the consideration otherwise payable pursuant to this Agreement such amounts as it is required to deduct and withhold with respect to the making of such payment under the Code, or any provision of state, local or foreign Tax Law. Assuming Seller delivers to Buyer the certificate described in Section 7.5(b)(viii), Buyer is not aware of any obligation to deduct or withhold on any consideration payable pursuant to this Agreement. In the event Buyer becomes aware of any such obligation, Buyer will promptly notify Seller and cooperate with Seller to reduce or eliminate any such withholding. To the extent that such amounts are so withheld, such withheld and deducted amounts will be treated for all purposes of this Agreement as having been paid to Seller in respect of which such deduction and withholding was made by such Party.

2.9 Capital Advance.

(a) At the Closing, Buyer shall deliver, by wire transfer of immediately available funds in Dollars to the Closing Account, an amount equal to twenty million Dollars (\$20,000,000) (the "<u>Buyer Capital Advance</u>"), to satisfy, in part, the obligations

of the NorthStar Member (on behalf of the JV Member) under Section 3.02(a) of the Newco LLC Agreement.

(b) At the Closing, Seller shall deliver to Buyer a guarantee (the "<u>Capital Contribution Advance</u> <u>Guarantee</u>") from Seller in respect of the NorthStar Member's obligations under the Newco LLC Agreement relating to the Capital Contribution Advance (as defined in the Newco LLC Agreement), substantially in the form of <u>Exhibit G</u> hereto.

ARTICLE III

REPRESENTATIONS AND WARRANTIES OF THE SELLER

Seller hereby represents and warrants to Buyer as follows:

- 3.1 <u>Title to the Purchased Interests</u>. After the completion of the Pre-Closing Restructuring and as of immediately prior to the Closing, Seller (or its Affiliates) will own one hundred percent (100%) of the JV HoldCo Interests, including one hundred percent (100%) of the Purchased Interests, in each case free and clear of Liens other than Liens imposed by Buyer or restrictions on transfer under applicable securities Laws. After completion of the Pre-Closing Restructuring and as of immediately prior to the Closing, neither Seller nor any of its Affiliates will hold any equity or debt interests in JV other than the JV HoldCo Interests. Upon consummation of the Transactions, Buyer will own (i) JV HoldCo Interests representing indirect ownership of the Buyer Ownership Percentage of the Company Interests and, indirectly, (ii) the Buyer Ownership Percentage of the Company Interests, in each case free and clear of Liens other than Liens imposed by Buyer or restrictions on transfer under applicable securities Laws, the JV LLC Agreement, the Newco LLC Agreement or the JV Letter Agreement.
- 3.2 Organization. After completion of the Pre-Closing Restructuring and as of immediately prior to the Closing, Seller will directly own 100% of the equity interests in NRF OP SPV. Seller is a limited partnership duly organized, validly existing and in good standing under the Laws of Delaware. NRF OP SPV is a limited liability company duly organized, validly existing and in good standing under the Laws of Delaware. Each of Seller and NRF OP SPV has the requisite corporate or partnership power and authority to own, lease or operate its properties and to carry on its business as now being conducted. Each of Seller and NRF OP SPV is duly qualified or licensed to do business and is in good standing (with respect to jurisdictions which recognize such concept) in each jurisdiction in which the nature of its business or the ownership, leasing or operation of its properties or assets makes such qualification or licensing necessary except where the failure to be so qualified or licensed or to be in good standing would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on the ability of Seller or NRF OP SPV to perform their respective obligations under this Agreement or the Transaction Documents to which they are a party.
- 3.3 <u>Authorization of Agreement</u>. Each of Seller and NRF OP SPV has all requisite power and authority and has taken all action required to execute and deliver this Agreement and the Transaction Documents to which it is a party and to perform its obligations hereunder and thereunder. This Agreement has been, and upon their execution the Transaction Documents to which Seller or NRF OP SPV is a party shall have been, duly executed and

delivered by Seller or NRF OP SPV, as the case may be, and, assuming that this Agreement and each such Transaction Document constitutes the legal, valid and binding obligation of the other parties hereto and thereto (other than another member of the Seller Group or an Affiliate thereof), this Agreement constitutes, and upon its execution each of the Transaction Documents to which Seller is a party shall constitute, the legal, valid and binding obligation of Seller, enforceable against Seller in accordance with its terms, subject to applicable Equitable Principles.

3.4 Conflicts; Consents of Third Parties.

- (a) The execution and delivery by each of Seller and NRF OP SPV of this Agreement and the Transaction Documents to which it is a party, the performance by each of Seller and NRF OP SPV of this Agreement and the Transaction Documents to which it is a party in accordance with their respective terms, and the consummation of the Transactions, do not and will not, conflict with, or result in any violation or breach of, or default (with or without notice or lapse of time or both) under, give rise to a right of termination, cancellation or acceleration of any material contractual obligation under, result in the creation of any Lien upon any of the Purchased Interests or material Properties under:
- (i) the Organizational Documents of Seller, NRF OP SPV, JV, JV OpCo, Newco or the Company or any of its Subsidiaries;
- (ii) any Contract or Permit to which Seller, NRF OP SPV, JV, JV OpCo, Newco or the Company or any of its Subsidiaries is a party or otherwise bound; or
- (iii) any Law applicable to Seller, NRF OP SPV, JV, JV OpCo, Newco or the Company or any of its Subsidiaries,

except, in the case of (ii) and (iii) where such conflict, violation, breach or default, would not have, and would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on the ability of Seller, NRF OP SPV, JV, JV OpCo, Newco or the Company to perform their respective obligations under this Agreement or the Transaction Documents to which they are a party.

- (b) Except as set forth on <u>Schedule 3.4(b)</u>, no consent, waiver, approval, Order, Permit or authorization of, or declaration or filing with, or notification to, any Governmental Entity (a "<u>Governmental Approval</u>") is required on the part of Seller or any of its Affiliates (other than the Company or any of its Subsidiaries) in connection with the execution and delivery by Seller and NRF OP SPV of this Agreement or the Transaction Documents to which they are a party or the consummation of the Transactions by Seller and its Affiliates (other than the Company or any of its Subsidiaries), except for any such Governmental Approval, the failure of which to make or obtain would not have, and would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on the ability of Seller or NRF OP SPV to perform their respective obligations under this Agreement or the Transaction Documents to which they are a party.
- 3.5 <u>Legal Proceedings</u>. There is no claim, action, suit, arbitration, inquiry, investigation, alternative dispute resolution action or any other judicial or administrative

proceeding of any nature, in Law or equity, by or before a Governmental Entity (each, a "<u>Legal Proceeding</u>"), pending against (or to Seller's Knowledge, threatened in writing against) Seller or NRF OP SPV that would reasonably be expected to have, individually or in the aggregate, a material adverse effect on the ability of Seller or NRF OP SPV to perform their respective obligations under this Agreement or the Transaction Documents to which they are a party. Neither Seller nor NRF OP SPV is subject to any outstanding Order which has had or would reasonably be expected to have, individually or in the aggregate, a material adverse effect on the ability of Seller or NRF OP SPV to perform their respective obligations under this Agreement or the Transaction Documents to which they are a party.

- 3.6 <u>Tax Status</u>. Seller is a U.S. Tax Person, and 100% of the equity interests in Seller are owned (directly or indirectly through other U.S. Tax Persons) by NRF and individuals or estates that are U.S. Tax Persons. To the actual knowledge of Seller and its Affiliates, (a) as of the date hereof, NRF qualifies as a REIT that is a domestically controlled qualified investment entity within the meaning of Code section 897(h) (a "<u>D-REIT</u>") and (b) as of the Closing either NRF will be a D-REIT or 100% of the equity interests in Seller will be owned (directly or indirectly through other U.S. Tax Persons) by one or more U.S. Tax Persons that are classified as (i) D-REITs, (ii) C corporations (within the meaning of U.S. Treasury Regulation section 1.337(d)-7(a)(2)(i)), (iii) individuals, or (iv) estates.
- 3.7 <u>Limitations of Representations and Warranties</u>. Except for the representations and warranties set forth in this <u>Article III</u> and in <u>Article IV</u> and any Transaction Document, neither Seller nor any other Person makes any express or implied representation or warranty with respect to Seller. Seller hereby disclaims any other express or implied representations or warranties. Except for the representations and warranties set forth in this <u>Article III</u> and any Transaction Document, Seller is not making any representations or warranties regarding the Company, JV, their respective Subsidiaries, the Healthcare Portfolio or any pro-forma financial information, financial projections or other forward-looking information or statements with respect thereto. Except for the representations and warranties set forth in this <u>Article III</u> and any Transaction Document, Seller hereby disclaims all liability and responsibility for any representation, warranty, opinion, projection, forecast, statement, memorandum, presentation, advice or information made, communicated or furnished (orally or in writing) to Buyer or its Affiliates or Representatives (including any such opinion, projection, forecast, statement, memorandum, presentation, advice or information made available in any electronic data room hosted by Seller or the Company in connection with the Transactions).

ARTICLE IV

REPRESENTATIONS AND WARRANTIES OF THE COMPANY

Except as set forth on the Disclosure Schedule, the Company hereby represents and warrants to Buyer as follows:

- 4.1 Organization.
- (a) JV is a limited liability company duly organized, validly existing and in good standing under the Laws of the State of Delaware.

- (b) JV OpCo is a limited liability company duly organized, validly existing and in good standing under the Laws of the State of Delaware.
- (c) Newco is a limited liability company duly organized, validly existing and in good standing under the Laws of the State of Delaware.
- (d) The Company is a limited liability company duly organized, validly existing and in good standing under the Laws of the State of Delaware.
- (e) Each of the Company Subsidiaries is a corporation, partnership, limited liability company or other company, as applicable, duly organized, validly existing and in good standing (with respect to jurisdictions which recognize such concept) under the Laws of the jurisdiction of its incorporation, except where the failure to be in good standing would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.
- (f) Each of the Company, JV, Newco and each of their respective Subsidiaries has the requisite power and authority to own, operate or lease the properties and assets owned, operated or leased by it and to conduct its business as it has been and is now being conducted. Each of the Company, JV, Newco and each of their respective Subsidiaries is duly qualified or licensed to do business in each jurisdiction in which the nature of its business or the ownership, leasing or operation of its rights, assets or properties makes such qualification or licensing necessary, except for those jurisdictions where the failure to be so qualified or licensed would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect. The Company has delivered to or made available to Buyer, prior to the execution of this Agreement, true, accurate and complete copies of the Organizational Documents of the Company, JV, JV OpCo and Newco, each as in effect as of the date hereof. Each of the Company, JV, JV OpCo, Newco and each of their respective Subsidiaries is in compliance in all material respects with the terms of their respective Organizational Documents.
- 4.2 <u>Authorization of Agreement</u>. Each of the Company, Newco, JV OpCo and JV has all requisite power and authority and has taken all action required to execute and deliver this Agreement and each Transaction Document to which it is a party and to perform its obligations hereunder and thereunder. This Agreement has been, and upon their execution the Transaction Documents to which the Company, Newco, JV OpCo or JV is a party shall have been, duly executed and delivered by the Company, Newco, JV OpCo or JV, as the case may be, and, assuming that this Agreement and each such Transaction Document constitutes the legal, valid and binding obligation of the other parties hereto and thereto (other than another member of the Seller Group or an Affiliate thereof), this Agreement constitutes, and, upon its execution each of the Transaction Documents to which the Company, Newco, JV OpCo or JV is a party shall constitute, the legal, valid and binding obligation of the Company, Newco, JV OpCo or JV, as the case may be, enforceable against the Company, Newco, JV OpCo or JV in accordance with its terms, subject to applicable Equitable Principles.

4.3 Conflicts; Consents of Third Parties.

- (a) The execution and delivery by each of JV, JV OpCo, Newco and the Company of this Agreement and the Transaction Documents to which it is a party, the performance by each of JV, JV OpCo, Newco and the Company of this Agreement and the Transaction Documents to which it is a party in accordance with their respective terms, and the consummation of the Transactions, do not and will not, conflict with, or result in any violation or breach of, or default (with or without notice or lapse of time or both) under, give rise to a right of termination, cancellation or acceleration of any material contractual obligation under, result in the creation of any Lien upon any of the Purchased Interests or material Properties under:
- (i) the Organizational Documents of the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries;
- (ii) any Contract or Permit to which the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries is a party or otherwise bound; or
- (iii) any Law applicable to the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries,

except, in the case clauses (ii) and (iii), where such conflict, violation, breach or default would not have, individually or in the aggregate, a Company Material Adverse Effect.

(b) No Governmental Approval is required on the part of JV, JV OpCo, Newco, the Company or any of their respective Affiliates in connection with the execution and delivery by Seller, NRF OP SPV, the Company, Newco, JV OpCo and JV of this Agreement or the Transaction Documents to which Seller, NRF OP SPV, the Company, Newco, JV OpCo or JV is a party or the consummation of the Transactions by Seller, NRF OP SPV, the Company, Newco, JV, JV OpCo and their respective Affiliates, except for any such Governmental Approval, the failure of which to make or obtain would not have, and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect or a material adverse effect on the ability of Seller, NRF OP SPV, the Company, Newco, JV OpCo or JV to perform their respective obligations under this Agreement or the Transaction Documents to which Seller, the Company, Newco, JV OpCo or JV is a party.

4.4 Capitalization; Subsidiaries.

- (a) After completion of the Pre-Closing Restructuring and as of immediately prior to the Closing, (i) Seller (or its Affiliates) will directly or indirectly own 100% of the JV HoldCo Interests, (ii) JV will directly own 100% of the equity interests in JV OpCo (the "JV OpCo Interests") and (iii) Seller (or its Affiliates) and JV OpCo will own 100% of the outstanding ownership interests of Newco (the "Newco Interests").
- (b) Except for the JV OpCo Interests, the JV Holdco Interests and the Newco Interests, as contemplated by the Pre-Closing Restructuring, as contemplated by <u>Section 10.6(c)</u>, or as set forth on <u>Schedule 4.4(b)</u>, there are no options, warrants, calls, subscriptions or other rights, agreements, arrangements or commitments of any kind, relating to the issued or unissued ownership interests of JV, JV OpCo, Newco or any of their respective

Subsidiaries, obligating Seller, JV, JV OpCo, Newco, the Company or any of their respective Subsidiaries to issue, transfer or sell or cause to be issued, transferred or sold any shares of capital stock of, or other equity interests in, JV, JV OpCo, Newco or any of their respective Subsidiaries (other than the Company and any Company Subsidiaries) or securities convertible into or exchangeable for such shares or equity interests, or obligating Seller, JV, JV OpCo, Newco the Company or any of their respective Subsidiaries to grant, extend or enter into any such option, warrant, call, subscription or other similar right, agreement, arrangement or commitment. Other than as contemplated by Section 10.6(c), there are no outstanding contractual obligations of Seller, JV, JV OpCo, Newco, the Company or any of their respective Subsidiaries to repurchase, redeem or otherwise acquire any shares of capital stock, or other equity interests in, JV, JV OpCo, Newco or any of their respective Subsidiaries. There are no bonds, debentures, notes, or other Indebtedness having the right to vote on any matters in which JV, JV OpCo, Newco or any of their respective Subsidiaries may vote.

- (c) (i) Immediately after giving effect to the Pre-Closing Restructuring and as of immediately prior to the Closing, Newco will own 100% of the outstanding ownership interests of the Company (the "Company Interests"). (ii) Except for the Company Interests and as contemplated by the Pre-Closing Restructuring or Section 10.6(c), there are no options, warrants, calls, subscriptions or other rights, agreements, arrangements or commitments of any kind, relating to the issued or unissued ownership interests of the Company or any Company Subsidiary, obligating Seller, JV, JV OpCo, Newco, the Company or any of their respective Subsidiaries to issue, transfer or sell or cause to be issued, transferred or sold any shares of capital stock of, or other equity interests in, the Company or any Company Subsidiary or securities convertible into or exchangeable for such shares or equity interests, or obligating Seller, JV, JV OpCo, Newco, the Company or any of their respective Subsidiaries to grant, extend or enter into any such option, warrant, call, subscription or other similar right, agreement, arrangement or commitment.
- (d) There are no voting trusts, stockholder agreements, proxies or other similar agreements or understandings with respect to the voting of any shares of or other equity interests in the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries to which the Company, JV, JV OpCo, Newco or any of their respective Affiliates is a party. None of the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries has granted any preemptive rights, anti-dilutive rights or rights of first refusal or similar rights with respect to any of its shares of capital stock or other equity interests. Except as set forth on Schedule 4.4(d)(ii), each Subsidiary of the Company, JV, JV OpCo or Newco is wholly owned, directly or indirectly, by the Company. Except as set forth on Schedule 4.4(d)(ii), the Company or another Company Subsidiary owns, directly or indirectly, all of the issued and outstanding equity interests of each of the Company Subsidiaries, free and clear of Liens (other than Permitted Liens, limitations on transfer and other restrictions imposed by federal or state securities Laws and the applicable Organizational Documents), and all such equity interests have been duly authorized and validly issued in compliance with all applicable Laws and Organizational Documents and are fully paid, nonassessable (to the extent such concepts are applicable) and are free, and were not issued in violation, of preemptive rights. Except as set forth on Schedule 4.4(d)(iii), the JV OpCo Interests, the JV HoldCo Interests, the Newco Interests and the Company Interests have been duly authorized and validly issued in compliance with all applicable Laws and Organizational Documents and are fully paid, nonassessable (to the extent such concepts are applicable) and are

free, and were not issued in violation, of preemptive rights. Except as set forth on <u>Schedule 4.4(d)(iv)</u>, there are no outstanding contractual obligations of JV, JV OpCo, Newco, the Company or any of their respective Subsidiaries to provide funds to, or make any investment (in the form of a loan, capital contribution or otherwise) in, any Person other than a Company Subsidiary.

- (e) Except for Company Subsidiaries, JV OpCo and Newco, none of the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries owns any shares of capital stock, or other equity interests in, any Person.
- (f) As of immediately prior to the Closing, (i) JV will have no assets and liabilities other than JV OpCo Interests, (ii) JV OpCo will have no assets and liabilities other than Newco Interests, (iii) Newco will have no assets and liabilities other than Company Interests and (iv) the Company will have no assets and liabilities other than with respect to the Healthcare Portfolio.
- (g) All dividends or other distributions on the JV HoldCo Interests, JV OpCo Interests, Newco Interests and the Company Interests which have been authorized or declared prior to the date hereof or pursuant to <u>Section 6.1(c)(ii)</u> have been paid in full (except to the extent such dividends have been declared and are not yet due and payable).

4.5 Financial Statements.

- (a) Schedule 4.5(a) contains true, accurate and complete copies of (i) the audited combined balance sheet of the Borrowers (as defined in that certain Loan Agreement, dated as of December 3, 2014, among Citigroup Global Markets Realty Corp., JPMorgan Chase Bank, National Association, Barclays Bank PLC and Column Financial, Inc., as Lenders, and the entities listed on Schedule I attached thereto, as Borrowers (the "Griffin Mortgage")) contractual basis as of December 31, 2015 and the related combined statements of operations contractual basis and cash flows contractual basis for the year then ended, together with all related notes and schedules thereto, (ii) the audited consolidated balance sheet of Eclipse Investment, LLC and its Subsidiaries as of December 31, 2015 and the related consolidated statements of operations, changes in members' equity and cash flows for the year then ended, together with all related notes and schedules thereto and (iii) the audited consolidated balance sheet of Eclipse Investment, LLC and its Subsidiaries as of December 31, 2014 and the related consolidated statements of operations, changes in members' equity and cash flows for the period May 7, 2014 through December 31, 2014, together with all related notes and schedules thereto (collectively, the "Audited Financial Statements").
- (b) <u>Schedule 4.5(b)</u> contains true, accurate and complete copies of the unaudited consolidated balance sheet of the Healthcare Portfolio as of March 31, 2016 and December 31, 2015 and the unaudited consolidated income statement of the Healthcare Portfolio for the three-month period ended March 31, 2016 and the year ended December 31, 2015, together with all related schedules thereto (collectively, the "<u>Unaudited Financial Statements</u>" and, together with the Audited Financial Statements, the "<u>Company Financial Statements</u>").

- (c) The Company Financial Statements (i) were prepared in accordance with GAAP applied on a basis consistent with the past practices of the Company and the Company Subsidiaries during the periods involved (except as may be indicated in the notes thereto or, in the case of interim financial statements, for normal year-end adjustments, which would not be material in amount), (ii) were prepared on the basis of the books and records of the Company and the Company Subsidiaries and (iii) fairly present (except as may be indicated in the notes thereto or schedules provided therewith) in all material respects, the consolidated financial position and the results of operations of the Healthcare Portfolio as of the dates or for the periods covered thereby (subject, in the case of unaudited interim financial statements, to the absence of notes and normal year-end adjustments).
- (d) The accounting or financial books, records, ledgers and files or other similar information of the Company, JV, JV OpCo, Newco and each of their respective Subsidiaries have been maintained in accordance with good business and accounting practices.
- 4.6 <u>Undisclosed Liabilities</u>. Except (a) to the extent reflected or otherwise specifically reserved against on the Company Financial Statements or referenced in the footnotes thereto, (b) for liabilities and obligations incurred in the ordinary course of business since the Balance Sheet Date, (c) for expenses related to the Transactions and (d) for liabilities and obligations pursuant to or arising under any Contract (including this Agreement) to which the Company or any Company Subsidiary is a party other than liabilities or obligations due to breaches thereunder, since the Balance Sheet Date neither the Company nor any Company Subsidiary has incurred any liabilities or obligations of any nature (whether accrued, absolute, contingent or otherwise) that would be required to be included on a consolidated balance sheet of the Company and the Company Subsidiaries prepared in accordance with GAAP, other than as have not had and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.

4.7 Absence of Certain Changes.

- (a) Since December 31, 2015 through the date hereof, (i) the Company has conducted, in all material respects, its business in the ordinary course, (ii) no Effects have occurred, which have had or could reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, (iii) suffered any casualty loss or damage with respect to the business or any material Property or the material assets of the Company and the Company Subsidiaries, taken as a whole, whether or not such loss or damage shall have been covered by insurance and (iv) Seller has not taken any action that, if taken after the date of this Agreement, without the prior written consent of Buyer, would constitute a breach of Section 6.1.
- (b) The transactions contemplated by the Colony Merger Agreement will not have or result in any Effects which will have, or would reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.

4.8 <u>Compliance with Laws</u>.

(a) (i) Each of Seller and NRF OP SPV and the Company, JV, JV OpCo, Newco and their respective Subsidiaries has complied and is in compliance with all Laws

which affect the business, properties or assets of the Company, JV, JV OpCo, Newco and their respective Subsidiaries, and (ii) no written notice has been received by Seller or NRF OP SPV or the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries, or to the Company's Knowledge, threatened against Seller or NRF OP SPV or the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries alleging any non-compliance with any such Laws, except in each case above, for such non-compliance that has not had and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect. Notwithstanding anything to the contrary in this Section 4.8(a), the provisions of this Section 4.8(a) shall not apply to matters addressed in Sections 4.8(c), 4.11 or 4.15.

- (b) The Company, JV, Newco and their respective Subsidiaries are in possession of all material Permits necessary for the Company and Company Subsidiaries to own or use, lease and, where applicable, operate the Healthcare Portfolio or to carry on their businesses in substantially the same manner as it is being conducted on the date of this Agreement (the "Company Permits") and all of the Company Permits are valid, and in full force and effect, except, in each case, where the failure by the Company, JV, Newco or a Company Subsidiary, as applicable, to possess and maintain any Company Permit in full force and effect has not had and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.
- (c) Except (i) as would not, individually or in the aggregate, reasonably be expected to have a Company Material Adverse Effect, and (ii) with respect to matters addressed in a Licensing Survey, (A) each of the Company and each Company Subsidiary is, and, to the Company's Knowledge, each third party lessee, operator or manager of any Property (each an "Operator") is, in compliance in all material respects with all applicable Health Care Laws relating to the ownership and operation of the Properties, (B) none of the Company, any Company Subsidiary or, to the Company's Knowledge, any Operator, has received any written notice from any Governmental Entity alleging any material violation of any applicable Health Care Law relating to the ownership and operation of the Properties, and (C) to the Company's Knowledge, there is no Legal Proceeding pending against or threatened in writing against the Company, any Company Subsidiary or any Operator alleging any material failure to comply with Health Care Laws relating to the ownership and operation of the Properties.
- (d) Except as would not, individually or in the aggregate, reasonably be expected to have a Company Material Adverse Effect, each Company Subsidiary that is required to be certified for participation in and reimbursement under any material Third Party Payor program that it currently participates in is so certified and, to the extent required, has current provider numbers and provider agreements for each such material Third Party Payor program.
- 4.9 <u>Legal Proceedings</u>. There is no Legal Proceeding pending against (or to the Company's Knowledge, threatened in writing against) the Company, JV, Newco or any of their respective Subsidiaries that would reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect. Neither the Company nor any Company Subsidiary is subject to any outstanding Order which has had or would reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.

4.10 Properties.

- (a) <u>Schedule 4.10(a)(i)</u> sets forth a list of each Property and whether such Property is owned or ground leased. <u>Schedule 4.10(a)(ii)</u> sets forth a list of the address of each facility and real property which, as of the date of this Agreement, is under contract by the Company or a Company Subsidiary for purchase or which (with respect to all or substantially all of such Property) is required under a written agreement to be sold, leased or subleased by the Company or a Company Subsidiary after the date of this Agreement. There are no real properties that the Company or any Company Subsidiary is obligated to sell, buy, lease or sublease at some future date. Neither the Company nor any Company Subsidiary owns or leases any real property which is not set forth on <u>Schedule 4.10(a)</u>.
- (b) The Company or a Company Subsidiary owns good and valid fee simple title (with respect to jurisdictions that recognize such form of title or substantially similar title with respect to all other jurisdictions) or leasehold title (as applicable) to each of the Properties, in each case, free and clear of Liens, except for Permitted Liens.
- (c) Neither the Company nor any Company Subsidiary has received (i) written notice to the effect that any Company Permit or any agreement, easement or other right of an unlimited duration that is necessary to permit the lawful use and operation of the buildings and improvements on any of the Properties or that is necessary to permit the lawful use and operation of all utilities, parking areas, retention ponds, driveways, roads and other means of egress and ingress to and from any of the Properties is not in full force and effect as of the date of this Agreement, (ii) written notice of any uncured violation of any Laws affecting the Properties, except as, individually or in the aggregate, would not reasonably be expected to have a Company Material Adverse Effect or (iii) written notice from any insurance company or underwriter of any defect that would materially and adversely affect the insurability of any of the Properties.
- (d) No Company Permit or any agreement, easement or other right that is necessary to permit the current use and operation of the buildings and improvements on any of the Properties or that is necessary to permit the current use and operation of all parking areas, driveways, roads and other means of egress and ingress to and from any of the Properties has failed to be obtained or is not in full force and effect, and neither the Company nor any Company Subsidiary has received written notice of any outstanding threat of modification or cancellation of any such Permit, except as, individually or in the aggregate, would not reasonably be expected to have a Company Material Adverse Effect.
- (e) <u>Schedule 4.10(e)</u> sets forth the Data Tape, which was prepared in good faith and, to Seller's Knowledge, does not contain in the aggregate any material deficiencies.
- (f) Schedule 4.10(f) (the "Rent Roll Summary") truly and accurately sets forth, as of September 30, 2016, (i) the tenant, (ii) the lease expiration date, (iii) the contractual rent and (iv) the portfolio such Property belongs to, in each case for each Property other than any Property where the Company or any Company Subsidiary is a tenant (other than ground leases). To the Company's Knowledge, no Person other than those identified on the Rent

Roll Summary (and those claiming by, through or under them) is in occupancy of any portion of the Properties set forth on the Rent Roll Summary. Except for discrepancies, errors or omissions that, individually or in the aggregate, would not reasonably be expected to have a Company Material Adverse Effect, the Rent Roll Summary correctly references the corresponding property with respect to each lease or sublease in effect on the date shown therein, the "Company Leases".

- (g) True, accurate and complete copies in all material respects of (i) all ground leases affecting the interest of the Company or any Company Subsidiary in the Properties and (ii) the Company Leases with aggregate annual rent payments to the Company or a Company Subsidiary in excess of Fifteen Million Dollars (\$15,000,000), in each case in effect as of the date of this Agreement (the "Material Company Leases") have been made available to Buyer. Except as individually or in the aggregate, would not reasonably be expected to have a Company Material Adverse Effect, (x) neither the Company nor any Company Subsidiary is and, to the Company's Knowledge, no other party is in breach or violation of, or default under, any Material Company Lease, (y) no event has occurred which would result in a breach or violation of, or a default under, any Material Company Lease by the Company or any Company Subsidiary, or, to the Company's Knowledge, any other party thereto in each case, with or without notice or lapse of time and no tenant under a Material Company Lease is in monetary default under such Material Company Lease, and (z) each Material Company Lease is valid, binding and enforceable in accordance with its terms and is in full force and effect with respect to the Company or a Company Subsidiary and, to the Company's Knowledge, with respect to the other parties thereto, except as may be limited by applicable Equitable Principles; provided, that, for the purposes of clause (x) and (y) above, no tenant will be deemed to be in monetary breach, violation or default under such Material Company Lease if such monetary breach, violation or default under such Material Company Lease if such monetary breach, violation or default under such Material Company Lease if such monetary breach, violation or default under such Material Company Lease if such monetary breach, violation or default has continued for a period of less than sixty (60) days and is with respect to an amount less than Five Million Dollars (\$5,000,000).
- (h) No purchase option, right of first refusal, right of first offer or any other right to purchase or otherwise acquire any Property or any portion thereof has been exercised under any Company Lease for which the purchase has not closed prior to the date of this Agreement.
- (i) Except as set forth in <u>Schedule 4.10(i)</u>, (i) there are no unexpired options to purchase agreements, rights of first refusal, rights of first offer or any other rights to purchase or otherwise acquire any Property or any portion thereof, and (ii) there are no agreements to enter into any contract for sale, ground lease or letter of intent to sell or ground lease any Property or any portion thereof that is owned by the Company or any Company Subsidiary, which, in each case, is in favor of any party other than the Company or a Company Subsidiary (a "<u>Company Third Party</u>").
- (j) The Company and each Company Subsidiary, as applicable, is in possession of title insurance policies or valid marked-up title commitments evidencing title insurance with respect to each Property (each, a "Company Title Insurance Policy" and, collectively, the "Company Title Insurance Policies"). To the Company's Knowledge, each Company Title Insurance Policy is in full force and effect. None of the Company or any Company Subsidiary has made a claim under any Company Title Insurance Policy.

- (k) The Company and Company Subsidiaries have good and valid title to, or a valid and enforceable leasehold interest in, or other right to use, all personal property owned, used or held for use by them (other than property owned by tenants and used or held in connection with the applicable tenancy and other than property owned by any third party managers), except as, individually or in the aggregate, would not reasonably be expected to have a Company Material Adverse Effect. Neither the Company's nor any Company Subsidiary's ownership of or leasehold interest in any such personal property is subject to any Liens, except for Permitted Liens.
- (l) The Company and Company Subsidiaries do not have any Knowledge, and have not received written notice, of (i) any structural defects, or violation of Law relating to any Properties which would have, or would reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, and (ii) any physical damage to any Properties which would have, or would reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect for which there is not insurance in effect covering the cost of the restoration and the loss of revenue.

4.11 Tax Matters.

- (a) Except as would not, individually or in the aggregate, reasonably be expected to have a Company Material Adverse Effect, (i) the Company, JV, JV OpCo, Newco and each of their respective Subsidiaries has timely filed with the appropriate Governmental Entity all Tax Returns required to be filed by the Company, JV, JV OpCo, Newco or their respective Subsidiaries as appropriate, taking into account any extensions of time within which to file such Tax Returns, and all such Tax Returns were true, accurate and complete when filed and (ii) the Company, JV, JV OpCo, Newco and each of their respective Subsidiaries has duly paid (or there has been paid on their behalf) all Taxes required to be paid by them.
- (b) Except as would not, individually or in the aggregate, reasonably be expected to have a Company Material Adverse Effect, (i) there are no current audits, examinations or other proceedings pending with regard to any Taxes of the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries, (ii) none of the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries has received a written notice or announcement of any audits or proceedings with respect to Taxes, (iii) no deficiency for Taxes of the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries has been claimed, proposed or assessed in writing or, to the Seller Group's Knowledge, threatened, by any Governmental Entity, which deficiency has not yet been settled, except for such deficiencies which are being contested in good faith that are set forth on Schedule 4.11(b), and (iv) none of the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries has waived any statute of limitations with respect to Taxes or agreed to any extension of time with respect to any Tax assessment or deficiency for any open tax year. None of the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries has entered into any "closing agreement" as described in Section 7121 of the Code (or any corresponding or similar provision of state, local or foreign income Tax Law).
- (c) Except as would not, individually or in the aggregate, reasonably be expected to have a Company Material Adverse Effect, the Company, JV, JV OpCo, Newco and each of their respective Subsidiaries has complied, in all material respects, with all

applicable Laws, rules and regulations relating to the payment and withholding of Taxes (including withholding of Taxes pursuant to Sections 1441, 1442, 1445, 1446 and 3402 of the Code or similar provisions under any state and foreign Laws) and have duly and timely withheld and, in each case, have paid over to the appropriate Taxing Authorities all amounts required to be so withheld and paid over on or prior to the due date thereof under all applicable Laws.

- (d) There are no material Tax Liens upon any property or assets of the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries except for Permitted Liens.
- (e) None of the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries has requested, has received or is subject to any written ruling of a Governmental Entity or has entered into any written agreement with a Governmental Entity with respect to Taxes.
- (f) There are no Tax allocation or sharing agreements or similar arrangements with respect to or involving the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries, except for customary indemnification provisions contained in any credit or other commercial agreements the primary purpose of which does not relate to Taxes. None of the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries (A) has been a member of an affiliated group filing a consolidated federal income Tax Return (other than a group the common parent of which was Seller) or (B) has any liability for the Taxes of any Person (other than the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries) under Treasury Regulation Section 1.1502-6 (or any similar provision of state, local, or foreign Tax Law) or as a transferee or successor.
- (g) Neither the Company nor any Company Subsidiary has participated in any "listed transaction" within the meaning of Treasury Regulation Section 1.6011-4(b)(2).
- (h) Except as set forth on <u>Schedule 4.11(h)</u>, the Company, JV OpCo and each of their respective Subsidiaries are, and have been since their formation, disregarded entities for federal income Tax purposes. <u>Schedule 4.11(h)</u> sets forth the U.S. federal income tax classification of each Subsidiary of the Company that is not a disregarded entity (including whether such entity is a REIT or a taxable REIT subsidiary).
- (i) Each Subsidiary of the Company that has filed with its Tax Return for any taxable year an election to be a REIT has satisfied all other requirements to qualify as a REIT for each subsequent taxable year for which it has been classified as a corporation for U.S. federal tax purposes. If the Company were a corporation for U.S. federal tax purposes, as of the Closing it would have met the requirements of Section 856(c)(2) and (3) of the Code for the 12-month period made up of the most recently ended calendar quarter and would have met the requirements of Section 856(c)(4) of the Code for the most recently ended calendar quarter and for each of the three calendar quarters immediately preceding such quarter.

- (j) Neither the Company nor any Company Subsidiary will be required to include any item of income (or exclude any item of deduction) in any taxable period ending after the Closing Date as a result of (i) any installment sale or open transaction disposition made on or prior to the Closing Date, (ii) any prepaid amount or deferred revenue received on or prior to the Closing Date, (iii) any cancellation of indebtedness income arising on or prior to the Closing Date, (iv) any change in Tax accounting method prior to the Closing Date under Section 481 of the Code or any comparable provision of state, local or non-U.S. Tax Law, or (v) any "closing agreement" as described in Section 7121 of the Code (or any corresponding or similar provision of state, local or non-U.S. Tax Law).
- (k) This <u>Section 4.11</u> and <u>Section 4.12</u> (to the extent such representations and warranties relate to Taxes) constitute the exclusive representations and warranties of the Company, JV, JV OpCo and Newco with respect to Tax matters. None of the representations and warranties in this <u>Section 4.11</u> or in <u>Section 4.12</u> (to the extent such representations and warranties relate to Taxes), other than those in <u>Section 4.11(j)</u> shall apply to any Tax period (or portion thereof) that begins on or after the Closing Date.
- 4.12 <u>No Employees; No Benefit Plans</u>. None of the Company, JV, Newco or any of their respective Subsidiaries has or has ever had employees. None of the Company, JV, Newco or any of their respective Subsidiaries sponsors, maintains, contributes to or has any obligation to contribute to, or has sponsored, maintained or contributed to or had any obligation to contribute to or has had any liability (contingent or otherwise) with respect to any "employee benefit plan," as defined in Section 3(3) of ERISA, any "single-employer plan," as defined in Section 4001(a)(15) of ERISA or any "multiemployer plan," as defined in Section 3(37) of ERISA.
- 4.13 <u>No Condemnation</u>. As of the date hereof (a) there are no pending or, to the Company's Knowledge, threatened eminent domain, condemnation, expropriation, requisition or similar proceedings against any Property or any portion thereof (each, a "<u>Condemnation</u>") that would have or would reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, and (b) none of Seller, the Company or any Company Subsidiary has received written notice that any Condemnation is contemplated.

4.14 Intellectual Property.

- (a) Each of the Company, JV, JV OpCo, Newco and their respective Subsidiaries does not infringe, misappropriate or otherwise violate or conflict with the Intellectual Property of any other Person. There is no Claim pending or, to the Company's Knowledge, threatened against the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries concerning any of the foregoing, nor has any of them received any written notification that a license under any other Person's Intellectual Property is or may be required to operate its respective business.
- (b) With respect to each item of Intellectual Property that is owned by any of the Company, JV, Newco and any of their respective Subsidiaries, such Intellectual Property is (i) owned free and clear of all exclusive licenses, non-exclusive licenses not granted in the ordinary course of business consistent with past practice, Liens (other than Permitted

Liens), and any obligation to grant any of the foregoing, and (ii) valid, subsisting and enforceable. No Person is engaging, or has engaged in the last six (6) years, in any activity that infringes, misappropriates or otherwise violates or conflicts with any of such Intellectual Property. Each of the Company, JV, Newco and their respective Subsidiaries has taken all commercially reasonable measures to maintain, in all material respects, the confidentiality and value of all confidential Intellectual Property used or held for use in the operation of its respective business.

4.15 Environmental Matters.

- The Company has made available to Buyer the environmental studies, investigations, reports, audits, assessments, licenses, permits and agreements set forth on Schedule 4.15 (collectively, the "Existing Environmental Reports"). Except as set forth in the Existing Environmental Reports: (i) (A) each of Seller and NRF OP SPV and the Company, JV, JV OpCo, Newco and their respective Subsidiaries is and, for the past two (2) years, has been in material compliance with all Environmental Laws which affect the business, properties or assets of the Company, JV, JV OpCo, Newco and their respective Subsidiaries, and (B) no written notice has been received by the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries, or to the Company's Knowledge, threatened against the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries alleging any material non-compliance with any such Environmental Laws; (ii) there has been no material release of Hazardous Substances by Seller or NRF OP SPV or the Company, JV, JV OpCo, Newco or their respective Subsidiaries, or, to the Company's Knowledge, at, in, on, to, or from any Property, that requires any investigation, cleanup, remediation, or remedial or corrective action pursuant to Environmental Law, and none of Seller, NRF OP SPV or the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries, within thirty-six (36) months prior to the date hereof, has made any report or disclosure to any Governmental Entity relating to the release or threatened release of Hazardous Substances to or from any Property; and (iii) there is no Legal Proceeding pending against (or to the Company's Knowledge, threatened in writing against), the Company, JV, JV OpCo, Newco or their respective Subsidiaries arising under or relating to any Environmental Law, and neither the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries is subject to any outstanding Order arising under or relating to any Environmental Law, provided, that, Buyer, for the avoidance of doubt, acknowledges and agrees that the foregoing clauses (i) through (iii) are to the Company's Knowledge with respect to the actions or operations of any Operator including any such actions or operations taken or failed to be taken on behalf of the Company or any of its Subsidiaries.
- (b) Except with respect to <u>Section 4.8(b)</u>, this <u>Section 4.15</u> constitutes the exclusive representations and warranties of the Company, JV and Newco with respect to compliance with or liabilities under Environmental Laws and other environmental matters.

4.16 Material Contracts.

(a) <u>Schedule 4.16(a)</u> sets forth a list or description of each note, bond, mortgage, indenture, lease, sublease, license, purchase or sale order, contract, arrangement, commitment, understanding or agreement or other instrument or obligation (whether written or oral) (each, a "<u>Contract</u>") to which Seller, NRF or NRF OP SPV or the Company, JV, JV OpCo,

Newco or any of their respective Subsidiaries (and for purposes of subclause (xii) only, any Affiliate of Seller) is a party or by which any of their respective properties or assets are bound (collectively, the "Material Contracts") which, as of the date of this Agreement:

- (i) obligates the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries to make non-contingent aggregate annual expenditures (other than principal and/or interest payments or the deposit of other reserves with respect to debt obligations) in excess of eighteen million Dollars (\$18,000,000) and is not cancelable within ninety (90) days without penalty to the Company, JV, Newco or any of their respective Subsidiaries, except for any (A) Company Lease or any ground lease affecting any Property or (B) tenant reimbursements made pursuant to the applicable underlying Company Lease;
- (ii) contains any non-compete or exclusivity provisions with respect to any line of business or geographic area or during any period of time with respect to the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries, or which restricts (or purports to restrict) the ability of the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries from engaging in any line of business or competing in any line of business or with any Person or in any geographic area or during any period of time, in each case, in any line of business or geographic area that is material to the Company and the Company Subsidiaries, except for any Company Lease or any ground lease affecting any Property;
- (iii) constitutes Indebtedness (including any guarantee thereof) in an amount in excess of eighteen million Dollars (\$18,000,000) or any letters of credit or similar instruments issued for the account of the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries or mortgaging, pledging or otherwise placing a Lien securing obligations in excess of eighteen million Dollars (\$18,000,000) on any portion of the assets of the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries, including the Griffin Mortgage, other than any such agreement, indenture, letter of credit or instrument solely between or solely among the Company, JV, JV OpCo, Newco and wholly owned Company Subsidiaries;
- (iv) requires the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries to dispose of or acquire assets or properties (other than in connection with the expiration of a Company Lease or a ground lease affecting a Property or as may accrue in connection with a breach of any Contract) with a fair market value in excess of eighteen million Dollars (\$18,000,000), or provides for any pending or contemplated merger, consolidation or similar business combination transaction involving the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries, except for any Company Lease or any ground lease affecting any Property;
- (v) relates to a joint venture, partnership, strategic alliance, limited liability company or similar arrangement, with a third party;
- (vi) constitutes Indebtedness of a Person (other than a Company Subsidiary) owed to the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries in an amount in excess of eighteen million Dollars (\$18,000,000);

- (vii) relates to the ongoing or scheduled development, construction or capital expenditures, in each case requiring aggregate payments by the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries in excess of eighteen million Dollars (\$18,000,000);
 - (viii) is a Material Company Lease;
- (ix) prohibits the payment of dividends or distributions in respect of the JV OpCo Interests, the JV HoldCo Interests, the Newco Interests or shares of the Company or any Company Subsidiary;
- (x) grants to any Person a right of first refusal, a right of first offer or an option to purchase, acquire, sell or dispose of any Property that, individually or in the aggregate, could involve amounts in excess of eighteen million Dollars (\$18,000,000);
- (xi) contains "earn-out" or other similar contingent purchase price payment obligations, in each case that could result in payments, individually or in the aggregate in excess of five hundred thousand Dollars (\$500,000);
- (xii) relates to the provision of Asset Management Services to JV, JV OpCo, Newco, the Company or any of their respective Subsidiaries (such agreements, the "<u>Asset Management Material Contracts</u>"); or
- (xiii) relates to any outstanding written commitment to enter into any contract or agreement of the type described in subclauses (i) through (xiii) above.

Notwithstanding anything above in subclauses (i) through (xiv), "Material Contracts" shall not include any Contract that (A) is terminable upon thirty (30) days' notice without penalty or premium, (B) will be fully satisfied at or prior to the Closing, (C) provides for aggregate payments by the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries of less than five million Dollars (\$5,000,000) in each calendar year during the remaining term of such Contract, or (D) is a Tax Protection Agreement or ground lease.

- (b) Except as, individually or in the aggregate, has not had and would not reasonably be expected to have a Company Material Adverse Effect:
- (i) each Material Contract (other than the Asset Management Material Contracts) is (A) legal, valid and binding on any of Seller, NRF or NRF OP SPV or the Company, JV, JV OpCo, Newco and any of their respective Subsidiaries that is a party thereto and, to the Company's Knowledge, each other party to such Material Contract and (B) enforceable against, and in full force and effect with respect to, any of Seller, NRF or NRF OP SPV or the Company, JV, JV OpCo, Newco and any of their respective Subsidiaries that is a party thereto and, to the Company's Knowledge, the other parties thereto, except as may be limited by applicable Equitable Principles;
- (ii) each Asset Management Material Contract, is, to the Company's Knowledge (A) legal, valid and binding on the parties thereto and (B) enforceable

against and in full force and effect with respect to the parties thereto, except as may be limited by applicable Equitable Principles;

- (iii) no asset manager has made a claim for indemnification under any Asset Management Material Contract, other than with respect to advancement of expenses in respect of claims in which NRF or any of its Subsidiaries is also named as a party;
- (iv) none of Seller, NRF or NRF OP SPV or the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries that is a party thereto, nor, to the Company's Knowledge, any other party thereto, is in material breach or violation of, or default under or otherwise delinquent in performance under, any Material Contract, and, to the Company's Knowledge, each such other party thereto has performed in all material respects all obligations required to be performed by it under such Material Contract; and
- (v) no event has occurred or not occurred through the action or inaction of Seller, NRF or NRF OP SPV or the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries or, to the Company's Knowledge, any other party thereto, that with notice or lapse of time or both would constitute a material violation, breach, default or delinquency in performance under any Material Contract.
- (c) None of Seller, NRF or NRF OP SPV or the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries or, to the Company's Knowledge, any other party thereto, has received notice of any material breach, violation, default or delinquency in performance under any Material Contract, except for breaches, violations, defaults or delinquencies in performance that would not, individually or in the aggregate, reasonably be expected to have a Company Material Adverse Effect. The Material Contracts shall remain valid and binding in accordance with their terms following the consummation of the transactions contemplated by the Colony Merger Agreement.
- (d) The Company has made available to Buyer, prior to the execution of this Agreement, true and complete copies in all material respects of each written Material Contract and a written summary of each oral Material Contract.
- 4.17 <u>Insurance</u>. The Company and the Company Subsidiaries maintain or cause to be maintained insurance with respect to all of the Properties with financially responsible insurers in such amounts and covering such risks which the Company, acting reasonably, believes are adequate in all material respects for the operation of its business. All premiums due and payable on the insurance policies held by or on behalf of the Company and the Company Subsidiaries relating to the Properties have been paid and no notice of cancellation or termination has been received by the Company or any Company Subsidiary (or the Person that maintains such insurance policies on their behalf) with respect to any such policy.
- 4.18 <u>Affiliated Transactions</u>. Except for Contracts entered into in connection with the Transactions (including the Pre-Closing Restructuring), <u>Schedule 4.18</u> lists every Contract between JV, JV Opco, the Company, Newco or any of their respective Subsidiaries, on the one hand, and Seller or any Affiliate of Seller (other than the Company, JV, JV OpCo or Newco or any of their respective Subsidiaries), on the other.

- 4.19 <u>Financial Advisor</u>. No broker, investment banker, financial advisor or other Person, other than UBS Securities LLC, the fees and expenses of which will be paid by Seller, is entitled to any broker's, finder's, financial advisor's or other similar fee or commission in connection with the Transactions based upon arrangements made by or on behalf of Seller, the Company or any of their respective Affiliates.
- 4.20 <u>Disclaimer of any Representations Regarding Estimates and Projections</u>. In connection with Buyer's investigation of the Healthcare Portfolio, Buyer has received certain projections, including projected statements of operating revenues and income from operations of the Healthcare Portfolio and certain business plan information. Buyer acknowledges that there are uncertainties inherent in attempting to make such estimates, projections and other forecasts and plans and that whether any such projections are in fact achieved will depend upon factors and future events not within the control of the any member of the Seller Group or any of their respective Subsidiaries. Buyer takes full responsibility for making its own evaluation of the adequacy and accuracy of all estimates, projections and other forecasts and plans so furnished to it (including the reasonableness of the assumptions underlying such estimates, projections and forecasts). Accordingly, except for the representations and warranties set forth in <u>Article III</u> and this <u>Article IV</u> and any Transaction Document, no member of the Seller Group makes any representation or warranty with respect to such estimates, projections and other forecasts and plans (including the reasonableness of the assumptions underlying such estimates, projections and forecasts). Buyer further acknowledges that the estimates and projections were compiled based only on unaudited data and operating information provided by the Company or the Company Subsidiaries (or managers of the Healthcare Portfolio), that neither the Seller Group (other than the Company) nor any Affiliate (other than the Company Subsidiaries) or Representative thereof has made an independent verification as to the accuracy or completeness thereof, or the assumptions and estimates therein, and that the presentation of such information does not reflect GAAP.
- 4.21 <u>Limitations of Representations and Warranties</u>. Except for the representations and warranties set forth in <u>Article III</u> and this <u>Article IV</u> and any Transaction Document, neither the Company nor any other Person makes any express or implied representation or warranty with respect to the Company, JV, Newco, their respective Subsidiaries or the Healthcare Portfolio. The Company hereby disclaims any other express or implied representations or warranties. Except with respect to claims of Fraud and for the representations and warranties set forth in <u>Article III</u> and this <u>Article IV</u> and any Transaction Document, the Company hereby disclaims all liability and responsibility for any representation, warranty, opinion, projection, forecast, statement, memorandum, presentation, advice or information made, communicated or furnished (orally or in writing) to Buyer or its Affiliates or Representatives (including any such opinion, projection, forecast, statement, memorandum, presentation, advice or information made available in any electronic data room hosted by the Seller Group in connection with the Transactions).

ARTICLE V

REPRESENTATIONS AND WARRANTIES OF BUYER

Except as set forth on the Disclosure Schedule, Buyer represents and warrants to Seller as follows:

- 5.1 Organization. Taikang Insurance Group Inc. owns, directly or indirectly, one hundred percent (100%) of the capital stock of Buyer, free and clear of Liens other than Liens imposed at Closing in connection with the funding of the loan under the Loan Agreements. Buyer is a Hong Kong limited company duly organized, validly existing and in good standing under the Laws of its jurisdiction of formation. Buyer has the requisite power and authority to own, lease or operate its properties and to carry on its business as now being conducted. Buyer is duly qualified or licensed to do business and is in good standing (with respect to jurisdictions which recognize such concept) in each jurisdiction in which the nature of its business or the ownership, leasing or operation of its properties or assets makes such qualification or licensing necessary, except to the extent that the failure to be so qualified or licensed or to be in good standing would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on the ability of Buyer to perform its obligations under this Agreement or the Transaction Documents to which it is a party. Buyer is not, directly or indirectly, controlled by or under the control of any state-owned or operated Persons.
- 5.2 <u>Authorization of Agreement</u>. Buyer has taken all limited company action required to execute and deliver this Agreement and the Transaction Documents to which it is a party and to perform fully its obligations hereunder and thereunder. This Agreement has been, and upon their execution the Transaction Documents to which Buyer is a party shall have been, duly executed and delivered by Buyer and, assuming that this Agreement and each such Transaction Document constitutes the legal, valid and binding obligation of the other parties hereto and thereto, this Agreement constitutes, and upon its execution each of the Transaction Documents to which Buyer is a party shall constitute, the legal, valid and binding obligation of Buyer, enforceable against Buyer in accordance with its terms, subject to applicable Equitable Principles.
- 5.3 <u>Conflicts; Consents of Third Parties</u>. Assuming the making of the filings and the receipt of the consents described in <u>Schedule 5.3</u>, the execution and delivery by Buyer of this Agreement and the Transaction Documents to which it is a party, the performance by Buyer of this Agreement and the Transaction Documents to which it is a party in accordance with their respective terms, and the consummation of the Transactions, do not and will not, conflict with, or result in any violation or breach of, or default (with or without notice or lapse of time or both) under, or give rise to a right of termination, cancellation or acceleration of any material contractual obligation under:
 - (a) the Organizational Documents of Buyer;
 - (b) any Contract or Permit to which Buyer is a party or otherwise bound; or

(c) any Law applicable to Buyer,

except, in the case of clauses (b) and (c), where such conflict, violation or default would not, and would not reasonably be expected to materially impair or delay the ability of Buyer to consummate the Transactions.

- 5.4 <u>Legal Proceeding</u>. There is no Legal Proceeding pending against (or to Buyer's Knowledge, threatened in writing against) Buyer or any of its Affiliates that would reasonably be expected to have, individually or in the aggregate, a material adverse effect on the ability of Buyer to perform its obligations under this Agreement or the Transaction Documents to which it is a party. Buyer is not subject to any outstanding Order which would, or would reasonably be expected to, materially impair or delay the ability of Buyer to consummate the Transactions.
- 5.5 <u>Financial Advisor</u>. Except for fees and commissions that will be paid directly by Buyer, no Person retained by or on behalf of Buyer or any of its Affiliates and no other Person, by reason of any action of Buyer, is entitled to any brokerage commissions, finders' fees or similar compensation in connection with the Transactions.
- 5.6 <u>Purchase for Investment</u>. Buyer is purchasing the Purchased Interests for its own account for investment and not for resale or distribution in any transaction that would be in violation of the securities Laws of the United States of America or any state thereof, *provided*, for the avoidance of doubt, that, as an Affiliate of an asset management company, Buyer may use multiple accounts managed by Buyer or one of its Affiliates to purchase the Purchased Interests or otherwise in connection with the Transactions. Buyer is an "accredited investor" as defined in Rule 501 of Regulation D promulgated under the Securities Act of 1933.
- 5.7 <u>Financial Ability</u>. Subject to receipt of the PRC Approvals, Buyer has, or will at Closing have, sufficient cash in Dollars readily available to enable it to consummate the Transactions, and there is not, nor will there be, any restriction on the use of such cash for such purpose.
- 5.8 <u>Sophisticated Party</u>. Buyer (a) is a sophisticated party with respect to the Purchased Interests, (b) is able to bear the financial risk associated with purchasing the Purchased Interests on the terms set forth in this Agreement, and (c) has such knowledge and experience in financial and business matters, and has made transactions of a similar nature, so as to be aware and capable of evaluating the merits and potential risks and uncertainties inherent in transactions of the type contemplated in this Agreement.
- 5.9 <u>Compliance with Laws</u>. Buyer and each of its Affiliates has complied with and is in compliance with any Orders or Laws which could affect the Transactions, except for non-compliance which would not have, and would not be reasonably be expected to have, a material adverse effect on the ability of Buyer to perform its obligations under this Agreement or the Transaction Documents to which it is a party.
- 5.10 <u>Investigation; Limitation on Representations and Warranties; Disclaimer of Other Representations and Warranties</u>. Buyer has conducted its own independent review and analysis of the business, operations, assets, technology, liabilities, results of operations, financial

condition and prospects of the Company, JV and the Healthcare Portfolio and acknowledges that Buyer has been provided access to personnel, properties, premises and records related to the Healthcare Portfolio for such purposes. In entering into this Agreement, Buyer has relied upon its independent investigation and analysis of the Company, JV, their respective Subsidiaries and the Healthcare Portfolio, and Buyer acknowledges and agrees that it has not been induced by and has not relied upon any representations, warranties or statements, whether express or implied, made by any member of the Seller Group, or any of their respective Affiliates, Subsidiaries, shareholders, controlling persons, trustees or Representatives that are not expressly set forth in Article III or Article IV of this Agreement or in a Transaction Document, whether or not such representations, warranties or statements were made in writing or orally. Buyer acknowledges and agrees that, except for the representations and warranties expressly set forth in Article III and Article IV of this Agreement and the Transaction Documents, (a) the Seller Group makes no, and has not made, any representations or warranties relating to itself or its business, the Healthcare Portfolio or otherwise in connection with the Transactions and Buyer is not relying on any representation or warranty except for those expressly set forth in Article III and Article IV of this Agreement or in a Transaction Document, (b) no Person has been authorized by the Seller Group to make any representation or warranty relating to itself or its business, the Healthcare Portfolio or otherwise in connection with the Transactions, and if made, such representation or warranty must not be relied upon by Buyer as having been authorized by such party and (c) any estimates, projections, predictions, data, financial information, memoranda, presentations or any other materials or information provided or addressed to Buyer or any of its Representatives are not and shall not be deemed to be or include representations or warranties unless any such materials or information is the subject of an express representation or warranty set forth in Article IV of this Agreement or in a Transaction Document.

5.11 <u>No Other Representations or Warranties</u>. Except for the representations and warranties set forth in this <u>Article V</u>, neither Buyer nor any other Person makes any express or implied representation or warranty with respect to Buyer. Buyer hereby disclaims any other express or implied representations or warranties.

ARTICLE VI

COVENANTS AND AGREEMENTS

- 6.1 Conduct of the Healthcare Portfolio Pending the Closing.
- (a) Between the date of this Agreement and the Closing Date, except as (i) set forth in Schedule 6.1, (ii) expressly contemplated or permitted by any other provision of this Agreement, (iii) required by Law or (iv) consented to in writing by Buyer, which consent shall not be unreasonably withheld, delayed or conditioned, each of the Company, JV, JV OpCo, Newco and each of their respective Subsidiaries agrees to, and Seller shall cause each of the Company, JV, JV OpCo, Newco and each of their respective Subsidiaries to, use its reasonable best efforts to conduct its business in the ordinary course of business and in a manner consistent with past practice, and each of the Company, JV, JV OpCo, Newco and each of their respective Subsidiaries agrees to, and Seller shall cause each of the Company, JV, JV OpCo, Newco and each of their respective Subsidiaries to, use its reasonable best efforts to, (A) preserve substantially intact its existing material rights, assets and properties, (B) preserve substantially

intact its business organization and (C) maintain and preserve intact its current relationships with material customers, suppliers, distributors, creditors and other Persons with which the Company, JV, JV OpCo, Newco or any of their respective Subsidiaries has significant business relations.

- (b) Between the date of this Agreement and the Closing Date or the date, if any, on which this Agreement is terminated in accordance with Section 9.1, except as (i) set forth on Schedule 6.1, (ii) expressly required by or necessary to perform any of the Seller Group's obligations under this Agreement, (iii) required by Law or (iv) consented to in writing by Buyer, which consent shall not be unreasonably withheld, delayed or conditioned, each of the Company, JV, Newco and each of their respective Subsidiaries agrees not to, and Seller shall cause the Company, JV, Newco and their respective Subsidiaries not to, (A) take or permit any action that, if taken after the Closing Date would constitute a "Major Decision" as such term is defined in the Newco LLC Agreement or (B) amend the Organizational Documents of the Company, JV, JV OpCo or Newco.
- (c) Notwithstanding the foregoing, subject to Sections 6.1(d) and 6.10(c), prior to the Closing, the Company, JV, JV OpCo, Newco and/or any of their respective Subsidiaries may use all available cash to (i) repay in the ordinary course of business any Indebtedness when due in accordance with its terms or (ii) declare or pay any dividends or other distributions with respect to the JV OpCo Interests, the JV HoldCo Interests, the Newco Interests or any equity interests of the Company or the Subsidiaries of the Company, *provided*, that, any dividends or other distributions declared with respect to the Newco Interests, the JV OpCo Interests, the JV HoldCo Interests or any equity interests of the Company or the Subsidiaries of the Company shall be paid prior to the Closing.
- (d) Between the Determination Time and the effective time of the consummation of the Transactions, each of the Company, JV, Newco and each of their respective Subsidiaries agrees not to, and Seller shall cause the Company, JV, Newco and their respective Subsidiaries not to, (i) declare or pay any dividends or other distributions with respect to the Newco Interests, the JV HoldCo Interests, the JV OpCo Interests or any equity interests of the Company or the Subsidiaries of the Company or JV, (ii) incur any Company Indebtedness or (iii) take any action outside of the ordinary course of business that will modify Lender Reserves or Working Capital in any way.

6.2 Efforts; Consents and Approvals.

(a) Subject to the terms and conditions of this Agreement, Buyer and Seller shall use their reasonable best efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable under applicable Law to consummate the Transactions as promptly as practicable, including (i) preparing and filing as promptly as practicable with any Governmental Entity all documentation to effect all necessary filings, notices, petitions, statements, registrations, submissions of information, applications and other documents, (ii) obtaining and maintaining all approvals, consents, registrations, permits, authorizations and other confirmations required to be obtained from any Governmental Entity that are necessary, proper or advisable to consummate the Transactions and (iii) taking all actions as may be required by any such Governmental Entity to obtain such authorization.

- (b) Buyer, on the one hand, and the Seller Group, on the other hand, shall, in connection with the efforts referenced in this Section 6.2 to obtain all authorizations for the Transactions under applicable competition or investment Law (collectively, the "Antitrust Laws"), including, for the avoidance of doubt, the PRC Approvals, use its reasonable best efforts to (i) cooperate in all respects with each other in connection with any filing or submission and in connection with any investigation or other inquiry, including any proceeding initiated by a private party, (ii) keep the other Party reasonably informed of any communication received by such Party from, or given by such Party to, any U.S. or foreign Governmental Entity and of any communication received or given in connection with any proceeding by a private party, in each case, regarding the Transactions, and (iii) permit the other Party to review any communication given by it to, and consult with each other in advance of any meeting or conference with any Governmental Entity or, in connection with any proceeding by a private party, with any other Person, and to the extent permitted by such applicable Governmental Entity or other Person, give the other Party the opportunity to attend and participate in such meetings and conferences.
- (c) In furtherance and not in limitation of the covenants of the Parties contained in Section 6.2(a), if any objections are asserted with respect to the Transactions under any Antitrust Law or if any suit or investigation is instituted (or threatened to be instituted) by any applicable Governmental Entity or any private party challenging the Transactions as violative of any Antitrust Law or which would otherwise prevent, impede or delay the consummation of the Transactions, Buyer and the Seller Group shall use their respective reasonable best efforts to cooperate with each other and resolve any such objections or suits so as to permit consummation of the Transactions.
- (d) In furtherance and not in limitation of the covenants of the Parties contained in <u>Section 6.2(a)</u> and <u>Section 6.2(b)</u>, Buyer, on the one hand, and the Seller Group, on the other hand, shall:
- (i) make all appropriate filings required in connection with the PRC Approvals as promptly as practicable following the date hereof and in any event, no later than five (5) Business Days following the date hereof, and supply as promptly as practicable any additional information and documentary material that may be requested pursuant to applicable Law in connection with the PRC Approvals;
- (ii) use reasonable best efforts to take, or cause to be taken, all actions and to do, or cause to be done, and to assist and cooperate with the other Parties hereto in doing, all things necessary, proper or advisable to obtain the PRC Approvals as soon as practicable and to avoid any impediment to the consummation of the Transactions under any applicable Laws or Orders, including using reasonable best efforts to take all such actions as may be necessary to resolve such objections, if any, as any PRC Governmental Entity or Person may assert under applicable Laws or Orders with respect to the Transactions; and
- (iii) in the event that Buyer is requested by a Governmental Entity to make a filing or submission to the Committee on Foreign Investment in the United States ("<u>CFIUS</u>"), (A) reasonably cooperate with each other in connection with any such filing or submission, (B) keep the other Party informed of any communication received by such Party from, or given by such Party to, CFIUS regarding the Transactions, and (C) permit the other

Party to review any communication given by it to, and consult with each other in advance of any meeting or conference with CFIUS, and to the extent permitted by CFIUS, give the other Party the opportunity to attend and participate in such meetings and conferences.

- (e) The Parties shall reasonably cooperate with each other and use their respective reasonable efforts to take such actions as the other may reasonably request to obtain (i) any consents necessary or desirable in connection with the Transactions pursuant to any Material Contracts, or (ii) any other consents from any third parties (excluding, for the avoidance of doubt, any Governmental Entity) as may be reasonably required to consummate the Transactions, *provided*, that, notwithstanding the foregoing, Buyer shall not be required to incur any expenses or liabilities, including, without limitation, the incurrence of interest or the payment of additional fees or penalties, make any payment or expenditures or agree to any obligation, undertaking or limitation of rights in order to obtain the consents set forth in this Section 6.2(e); provided, further, that the consummation of the Transactions shall not be delayed by, or conditioned on, the receipt or failure to receive of any such consents. For the avoidance of doubt, following the Closing, the Parties shall continue to reasonably cooperate with each other and use their respective reasonable efforts to take such actions as the other may reasonably request with respect to any of the consents set forth in this Section 6.2(e) that have not been received as of the Closing.
- 6.3 <u>Pre-Closing Restructuring</u>. Subject to and upon the terms and conditions of this Agreement, prior to or at the Closing, Seller shall take, and shall cause the Company, NRF OP SPV, JV, JV OpCo, Newco and their respective Subsidiaries to take, such actions as are necessary to give effect to the Pre-Closing Restructuring.

6.4 Financing.

- (a) Buyer shall (and shall cause each of its Affiliates to) use commercially reasonable efforts to as promptly as practicable after the date hereof, (i) secure the Debt Financing and execute the Loan Agreements and (ii) deliver the fully executed Letter of Credit to Seller or the Deposit Amount to the Escrow Agent, as applicable and in accordance with Section 2.2(b).
- (b) Following the execution of the Loan Agreements, Buyer shall (i) as promptly as reasonably practicable provide Seller with the fully executed Loan Agreements, (ii) as promptly as reasonably practicable after the occurrence thereof notify Seller of any breach or default by any party to any Loan Document that is not curable prior to the Outside Date, (iii) notify Seller, as promptly as reasonably practicable after Buyer's receipt of any written notice or other written communication from any Debt Financing Source with respect to any actual or threatened breach, default, termination or repudiation by such Debt Financing Source of its obligations or commitments under any Loan Document, and (iv) use commercially reasonable efforts to (and shall cause each of its Affiliates to use commercially reasonable efforts to) (A) maintain in effect the Debt Financing and the Loan Documents, (B) satisfy in all material respects on a timely basis all conditions applicable to Buyer set forth in the Loan Documents including, pledging or causing its Affiliates to pledge all collateral required under the Loan Documents and (C) enforce all rights of Buyer under the Loan Documents with respect to the

funding of the Debt Financing Sources' obligations to the extent it would be commercially reasonable to do so.

- (c) In connection with the Debt Financing, (i) no member of the Seller Group nor any of their respective Affiliates shall be required to become subject to any obligations or liabilities with respect thereto, and (ii) nothing shall obligate any member of the Seller Group or any of their respective Affiliates to provide any certificates or declare or make any determinations with respect thereto. Buyer acknowledges and agrees that no member of the Seller Group nor any of their respective Affiliates shall have any responsibility for, or incur any liability to any Person under or in connection with, the Loan Documents or the Debt Financing and Buyer shall indemnify and hold harmless each member of the Seller Group and their respective Affiliates and Representatives from and against any and all Losses suffered or incurred by them in connection with the arrangement of the Debt Financing.
- Group, nor any of their respective Affiliates, including after the consummation of the transactions contemplated by the Colony Merger Agreement, any successor entity of any of the foregoing, shall issue or cause the publication of any press release or other public announcement with respect to the Transactions or this Agreement or the Transaction Documents without the prior written consent of the other Party, unless such Party determines, after consultation with outside counsel, that it is required by applicable Law or by any listing agreement with or the listing rules of a national securities exchange or trading market or by the CIRC to issue or cause the publication of any press release or other announcement with respect to the Transactions or this Agreement or the Transaction Documents, in which event such Party shall provide a reasonable opportunity to the other Party to review and comment upon such press release or other public announcement and shall give due consideration to all reasonable additions, deletion or changes suggested thereto; *provided*, *however*, that each Party and their respective Affiliates may make statements that are not inconsistent with previous press releases, public disclosures or public statements made by Buyer or a member of the Seller Group in compliance with this Section 6.5. Except to the extent modified by the foregoing provisions, the Confidentiality Agreement dated as of December 10, 2015, between NRF and Buyer (the "Confidentiality Agreement") shall (i) prior to the occurrence of the Closing, remain in full force and effect, notwithstanding the execution and delivery, or any termination, of this Agreement and (ii) upon the Closing, be terminated.
- 6.6 <u>Conveyance Taxes</u>. Buyer and the Seller Group will reasonably cooperate in the preparation, execution and filing of Tax Returns, questionnaires, applications or other documents regarding any real property transfer or gains, sale, use, value added and stamp Taxes, any recording, registration and other fees and similar Taxes ("<u>Conveyance Taxes</u>") which become payable in connection with the Transactions. Conveyance Taxes payable in connection with the Transactions (other than the sale and purchase of the Purchased Interests hereunder) shall be the sole responsibility of the Company. Conveyance Taxes, if any, payable solely as a result of the sale and purchase of the Purchased Interests hereunder shall be borne 50% by Seller and 50% by Buyer.
- 6.7 <u>Expenses</u>. Except as otherwise specifically provided herein, Buyer, on the one hand, and Seller, on the other hand, shall bear their respective expenses incurred in

connection with the preparation, execution and performance of this Agreement and the Transactions, including all fees and expenses of agents, representatives, counsel and accountants.

- 6.8 <u>Further Assurances</u>. Subject to the terms and conditions of this Agreement, each of the Parties shall use reasonable best efforts to take or cause to be taken all actions and to do or cause to be done all things reasonably necessary, proper or advisable under applicable Laws to consummate and make effective the Transactions, including (a) contesting any legal proceeding relating to the Transactions and (b) executing any additional instruments necessary to consummate the Transactions.
- 6.9 <u>Managers</u>. Prior to or at the Closing, Seller shall take, and shall cause its Subsidiaries to take, such actions as are necessary and appropriate so that, effective as of the Closing, (i) the board of managers of Newco shall be comprised of the individuals set forth on <u>Schedule 6.9(a)</u> and (ii) the board of directors of JV shall be comprised of the individuals set forth on <u>Schedule 6.9(b)</u>.

6.10 Casualty and Condemnation.

- (a) If, prior to the Closing Date, a Property or any portion thereof suffers a Material Casualty or the Company or any Company Subsidiary receives a notice or the Seller Group obtains Knowledge of a Material Condemnation, Seller shall, or shall cause the Company and the Company Subsidiaries to, promptly (a) notify Buyer of that event and provide Buyer with details of the extent of the Material Casualty or Material Condemnation, (b) make and diligently pursue a claim under any insurance policy covering such Property with respect to such Material Casualty or Material Condemnation and keep Buyer apprised of any material developments concerning such claim, (c) with respect to any Property or any portion thereof that suffers a Material Condemnation, diligently pursue any claim or rights to a condemnation award or proceeds relating to such Property or portion thereof and (d) with respect to any Property or any portion thereof that suffers a Material Casualty, use any proceeds from any such insurance policy to repair any damage to such Property or portion thereof.
- (b) Buyer shall be bound to purchase the Purchased Interests for the Purchase Price (after taking into account the Equity Value Adjustments contemplated pursuant to <u>Section 2.4</u>) as required by the terms hereof without regard to the occurrence or effect of any such Casualty or Condemnation.
- (c) If, prior to the Closing, a Property or any portion thereof suffers a Casualty, the Company and/or any of its Subsidiaries shall retain any proceeds received from any insurance policy covering such Property or portion thereof and shall in no event declare or pay any dividends or other distributions with respect to any equity interests of the Company or the Subsidiaries of the Company in respect of such proceeds.
- 6.11 Advice of Changes Seller (on behalf of the Seller Group) and Buyer will promptly advise each other of (a) any event to the Knowledge of either party which would render any representation or warranty contained in this Agreement, if made on or as of the date of the Closing, materially untrue or inaccurate, (b) any Effect that has or would reasonably be expected to have a Company Material Adverse Effect or (c) any failure of any member of the Seller Group

or Buyer to materially comply with or satisfy any covenant, condition or agreement to be complied with or satisfied by it hereunder. Notwithstanding the disclosure to Buyer or Seller, as applicable of any matter pursuant to this <u>Section 6.11</u>, the disclosing party shall not be relieved of any liability for, nor shall the providing of such information by the disclosing party to Buyer or Seller, as applicable, be deemed a waiver of, (i) the breach of any representation, warranty, covenant or agreement of any member of the Seller Group or Buyer contained in this Agreement or (ii) the related indemnification obligations of Seller or Buyer under <u>Article VIII</u>.

ARTICLE VII

CLOSING

- 7.1 <u>Closing</u>. The closing of the purchase and sale of the Purchased Interests (the "<u>Closing</u>") shall take place on the Closing Date at the offices of Paul, Weiss, Rifkind, Wharton & Garrison LLP, 1285 Avenue of the Americas, New York at 10:00 a.m. Eastern Time, or such earlier or later time on the Closing Date as may be agreed in writing by Seller and Buyer.
- 7.2 <u>Conditions to Each Party's Obligations to Effect the Closing</u>. The respective obligations of each Party to effect the Transactions will be subject to the satisfaction on or prior to the Closing Date of the following conditions, any and all of which may be waived in whole or in part by Buyer and Seller (on behalf of the Seller Group), as the case may be, to the extent permitted by applicable Law:
- (a) <u>Governmental and Regulatory Approvals</u>. The consents, approvals and actions of, filings with, and notices to, any Governmental Entity set forth on <u>Schedule 7.2(a)</u> shall have been obtained or made and remain in full force and effect.
- (b) <u>No Injunction or Restraint</u>. No preliminary or permanent injunction or Order issued by any court of competent jurisdiction or other Governmental Entity in PRC or a jurisdiction material to the business of the Healthcare Portfolio shall be in effect preventing or prohibiting the consummation of the Transactions.
- 7.3 <u>Conditions to Obligations of the Seller Group</u>. The obligations of the Seller Group to effect the Transactions are also subject to the satisfaction or waiver by Seller on or prior to the Closing Date of each of the following additional conditions:
- (a) Representations and Warranties. Each of the representations and warranties of Buyer set forth in this Agreement shall be true and correct in all respects at and as of the Closing Date, as though made on and as of the Closing Date (except to the extent that such representations and warranties are expressly limited by their terms to another date, in which case such representations and warranties shall be true and correct in all respects as of such other date), except where the failure of such representations and warranties to be true and correct (without giving effect to any "materiality" qualifications or exceptions set forth therein) would not, individually or in the aggregate, reasonably be expected to have a material adverse effect on the ability of Buyer to perform its obligations under this Agreement or the Transaction Documents to which it is a party.

- (b) <u>Performance of Obligations</u>. Buyer shall have performed or complied in all material respects with all obligations (other than those set forth in <u>Section 6.11</u>) required to be performed or complied with by it under this Agreement at or prior to the Closing Date.
- (c) <u>Closing Deliveries</u>. Buyer shall have made the deliveries required to be made by Buyer under <u>Section 7.5(a)</u>.
- 7.4 <u>Conditions to Obligations of Buyer</u>. The obligations of Buyer to effect the Transactions are also subject to the satisfaction or waiver by Buyer on or prior to the Closing Date of each of the following additional conditions:
- (a) Representations and Warranties. (i) The representations and warranties of (A) Seller in Section 3.1 (*Title to Purchased Interests*) and (B) the Company in Sections 4.4(a) and (c)(i) (Capitalization; Subsidiaries) and 4.7(a)(ii)(Absence of Certain Changes) shall be true and correct in all respects at and as of the Closing Date, as though made on and as of the Closing Date (except to the extent that such representations and warranties are expressly limited by their terms to another date, in which case such representations and warranties shall be true and correct in all respects as of such date), (ii) the representations and warranties of (A) Seller in Sections 3.2 (Organization), 3.3 (Authorization of Agreement), 3.4(a)(i) (Conflicts) and 3.6 (Tax Status) and (B) the Company in Sections 4.1 (Organization), 4.2 (Authorization of Agreement), 4.3(a)(i) (Conflicts), 4.4(b), (c)(ii), (d) and (e)-(g) (Capitalization; Subsidiaries), 4.10(b) (Properties), and 4.19 (Financial Advisor) shall be true and correct in all material respects at and as of the Closing Date, as though made on and as of the Closing Date (except to the extent that such representations and warranties are expressly limited by their terms to another date, in which case such representations and warranties shall be true and correct in all material respects as of such date), except for such representations and warranties that are qualified by qualifications or exceptions as to "materiality" or "Company Material Adverse Effect", in which case such representations shall be true and correct in all respects at and as of the Closing Date, and (iii) the other representations and warranties of members of the Seller Group set forth in this Agreement shall be true and correct in all respects at and as of the Closing Date, as though made on and as of the Closing Date (except to the extent that such representations and warranties are expressly limited by their terms to another date, in which case such representations and warranties shall be true and correct in all respects as of such other date), except where the failure of such representations and warranties to be true and correct (without giving effect to any "materiality" or "Company Material Adverse Effect" qualifications or exceptions set forth therein) would not, individually or in the aggregate, reasonably be expected to have a Company Material Adverse Effect.
- (b) <u>Performance of Obligations</u>. The applicable members of the Seller Group shall have performed or complied in all material respects with all obligations (other than those set forth in <u>Section 6.11</u>) required to be performed or complied with by such members of the Seller Group under this Agreement at or prior to the Closing Date.
- (c) <u>No Company Material Adverse Effect</u>. Since the date of this Agreement, no Company Material Adverse Effect shall have occurred and be continuing.

- (d) <u>Pre-Closing Restructuring Transactions</u>. Seller shall have provided evidence that the Pre-Closing Restructuring shall have been consummated in accordance with <u>Exhibit A</u>.
 - (e) <u>Financing</u>. Buyer shall have executed the Loan Agreements.
- (f) <u>Closing Deliveries</u>. The Seller Group shall have made the deliveries required to be made by the Seller Group under <u>Section 7.5(b)</u>.

7.5 <u>Closing Deliveries</u>.

- (a) <u>Deliveries by Buyer to Seller</u>. At the Closing, Buyer shall deliver to Seller:
 - (i) the Purchase Price by wire transfer of immediately available funds in Dollars to the Closing

Account;

(ii) the Buyer Capital Advance by wire transfer of immediately available funds in Dollars to the

Closing Account;

- (iii) the Deposit Release Instruction executed by Buyer;
- (iv) the JV LLC Agreement executed by Buyer;
- (v) the JV Letter Agreement executed by Buyer; and
- (vi) a certificate of Buyer, dated as of the Closing Date, and signed by an authorized officer of Buyer, in the form set forth on <u>Exhibit E-1</u>.
 - (b) <u>Deliveries by the Seller Group</u>. At the Closing, the Seller Group shall deliver to Buyer:
 - (i) the Deposit Release Instruction executed by Seller;
 - (ii) the JV LLC Agreement executed by NRF OP SPV;
 - (iii) the JV Letter Agreement executed by NRF OP SPV;
 - (iv) the Newco LLC Agreement executed by NRF OP SPV and JV OpCo;
 - (v) the Capital Contribution Advance Guarantee executed by Seller;
 - (vi) evidence of the issuance of the Purchased Interests to Buyer via book-entry format;

- (vii) evidence of the appointment of the two (2) directors of each of the JV, JV OpCo and Newco to be designated by Buyer no later two (2) Business Days prior to the Closing, effective as of the Closing;
- (viii) a certificate signed by Seller, that Seller is not a "foreign person" as defined in Section 1445 of the Code;
- (ix) a certificate of Seller, dated as of the Closing Date, and signed by an authorized officer of the Company, in the form set forth on Exhibit E-2; and
- (x) a certificate of the Company, dated as of the Closing Date, and signed by an authorized officer of the Company, in the form set forth on Exhibit E-3.

ARTICLE VIII

INDEMNIFICATION

8.1 <u>Survival</u>. The representations and warranties of the Parties shall survive the Closing until the date that is thirty (30) days following Buyer's receipt of audited financial statements of the Company and its Subsidiaries for the calendar year 2017 pursuant to Section 6.04 of the Newco LLC Agreement (such date, the "<u>Survival Date</u>"), except that (a) the Fundamental Representations (other than the representations and warranties set forth in <u>Section 4.11(j)</u> (*Tax Matters*)) shall survive indefinitely, and (b) the representations and warranties set forth in <u>Sections 3.6</u> (*Tax Status*) and <u>4.11</u> (*Tax Matters*) shall survive the Closing and shall remain in full force until the date that is ninety (90) days after the expiration of the applicable statutes of limitations for any Taxes arising out of, or resulting from or relating to the inaccuracy of such representation or warranty (taking into account any extensions or waivers thereof). The covenants and agreements of the Parties that are to be performed prior to the Closing shall survive the Closing shall survive, and thus a Claim may be brought in respect of a breach thereof, until the expiration of the applicable statute of limitations in respect of such post-Closing covenants.

8.2 <u>Obligation of Seller to Indemnify</u>.

(a) From and after the Closing Date, subject to the provisions and limitations in this Article VIII, Seller shall indemnify, defend and hold harmless Buyer and its directors, managers, members, officers, employees, Affiliates, agents and representatives (collectively, the "Buyer Indemnified Parties"), from and against all liabilities, Taxes, losses and damages and reasonable attorneys' fees, court costs and other out-of-pocket expenses (collectively, "Losses") suffered or incurred by the Buyer Indemnified Parties to the extent arising out of, or resulting from or relating to (i) the inaccuracy of any of the representations and warranties set forth in Article III, Article IV or made in any certificate delivered pursuant hereto when made or as of the Closing Date (or, in the case of any such representation or warranty that is limited by its terms to the date hereof or another date, the inaccuracy as of such date) (and, other than with respect to the representations and warranty contained in (A) Sections 4.5(c) (Financial Statements), 4.16(a)(ii) (Material Contracts), 4.10(e) (Data Tape) and 4.7(a)(ii) and

- (b) (Absence of Certain Changes) and (B) Section 4.8(a) (Compliance with Laws) solely with respect to any inaccuracy in such representation and warranty which arises during period between the date hereof and Closing, without giving effect to any "materiality" or "Company Material Adverse Effect" qualifications or exceptions set forth therein), (ii) the failure of any member of the Seller Group to perform or fulfill any covenant or agreement of such member of the Seller Group contained in this Agreement, (iii) any Income Taxes imposed on the Company or any Company Subsidiary with respect to any Pre-Closing Tax Period, (iv) the consummation of the transactions contemplated by the Colony Merger Agreement or (v) any liabilities or obligations of Seller or its Affiliates to the extent not related to the Healthcare Portfolio; provided, that, with respect to any Loss suffered or incurred directly or indirectly by the Company and only indirectly suffered by Buyer or any of its Affiliates as a member of JV, the term "Loss" shall mean the percent of such Loss equal to the Buyer's indirect ownership percentage of Newco (which, at Closing, shall be the Buyer Ownership Percentage) at the time of such Loss.
- (b) The right to indemnification, reimbursement or other remedy based upon representations, warranties, covenants and agreements shall not be affected by any investigation conducted with respect to, or any knowledge actually acquired at any time, whether before or after the execution and delivery of this Agreement or the Closing Date, with respect to the accuracy or inaccuracy of or compliance with any such representation, warranty, covenant or agreement.
- 8.3 <u>Obligation of Buyer to Indemnify</u>. From and after the Closing Date, subject to the provisions and limitations in this <u>Article VIII</u>, Buyer shall indemnify, defend and hold harmless Seller and its directors, managers, members, officers, employees, Affiliates, agents and representatives (collectively, the "<u>Seller Indemnified Parties</u>"), from and against all Losses to the extent arising out of, or resulting from or relating to (a) the inaccuracy of any of the representations and warranties set forth in <u>Article V</u> or made in any certificate delivered pursuant hereto when made or as of the Closing Date (or, in the case of any such representation or warranty that is limited by its terms to the date hereof or another date, the inaccuracy as of such date) or (b) the failure of Buyer to perform or fulfill any covenant or agreement of Buyer contained in this Agreement.
- 8.4 <u>Certain Limitations</u>. Notwithstanding the provisions of this <u>Article VIII</u>, after the Closing, (a) the Buyer Indemnified Parties shall not be entitled to recover pursuant to <u>Section 8.2(a)(i)</u> (other than with respect to breaches of Fundamental Representations) until the Losses (excluding all Losses with respect to De Minimis Claims) relating thereto exceed, in the aggregate, Three Million Five Hundred Thousand Dollars (\$3,500,000) (the "<u>Threshold</u>"), *provided*, that, the Buyer Indemnified Parties shall only be entitled to indemnification for the amount of such Losses in excess of the Threshold, (b) in no event shall the Buyer Indemnified Parties be entitled to recover pursuant to <u>Section 8.2(a)(i)</u> (other than with respect to breaches of Fundamental Representations) any amounts in excess of Thirty Five Million Dollars (\$35,000,000) (the "<u>Cap</u>"); *provided*, that, the Cap shall be increased to Seventy Million Dollars (\$70,000,000) (less any amounts recovered to which the Cap applies) with respect to indemnification for breaches of the representations and warranties set forth in <u>Section 4.10(b)</u> (*Title to Properties*), and (c) in no event shall the Buyer Indemnified Parties be entitled to recover pursuant to <u>Section 8.2(a)(i)</u> (other than with respect to breaches of Fundamental

Representations) for any single claim or aggregated claims arising out of substantially the same events or circumstances unless the amount of Losses attributable to such claim or aggregated claims arising out of substantially the same events or circumstances exceed \$100,000 ("De Minimis Claims"). Notwithstanding the foregoing, the maximum aggregate amount recoverable by the Buyer Indemnified Parties for Losses under this Agreement shall be limited to and shall not exceed the Purchase Price actually paid to and received by Seller.

8.5 Indemnification Procedure.

- (a) Any Buyer Indemnified Party or Seller Indemnified Party, as applicable (each, an "Indemnified Party") shall, within the relevant limitation period provided in Section 8.1, promptly (and in any event within 30 days) give the indemnifying party (the "Indemnifying Party") written notice (a "Claim Notice") describing in reasonable detail the facts giving rise to any Claims for indemnification hereunder and shall include in the Claim Notice (if then known) the amount or method of computation of the amount of such Claim and a reference to the provision of this Agreement or any agreement, certificate or instrument delivered pursuant to this Agreement upon which such Claim is based; provided, that, a Claim Notice in respect of any action at law or in equity by or against a third party as to which indemnification will be sought shall be given promptly after the action or suit is commenced, provided, further, that any delay in the provision of, or the failure by an Indemnified Party to provide, a Claim Notice will not affect the Indemnified Party's right to indemnification unless (and then solely to the extent) the Indemnifying Party is actually prejudiced by such delay or failure.
- (b) The Indemnifying Party shall have the right, at its own cost, to participate jointly in the defense of any Claim or demand in connection with which the Indemnified Party has claimed indemnification hereunder, and the Indemnifying Party may elect to take over the defense of such Claim or demand through counsel of its own choosing by so notifying the Indemnified Party within thirty (30) days of receipt of the Indemnified Party's notice of such Claim or demand. If the Indemnifying Party elects to take over the defense, then:
- (i) the Indemnified Party and the Indemnifying Party shall cooperate in the defense or prosecution of such Claim or demand, including by making available, at the Indemnifying Party's expense (or, with respect to any reasonable and documented out-of-pocket expenses incurred by the Indemnifying Party, at the Indemnified Party's expense, if it is ultimately determined that the Indemnified Party is not entitled to indemnification hereunder), all witnesses, pertinent records, materials and information in such party's possession or under its control relating thereto as is reasonably necessary to the defense or prosecution of such Claim or demand and, in the case of the Indemnifying Party, by keeping the Indemnified Party reasonably informed as to the status of such matter and by providing a reasonable opportunity to the Indemnified Party to review and comment upon any pleading and giving due consideration to all reasonable additions, deletion or changes suggested thereto prior to the filing thereof;
- (ii) with respect to any issue involved in such Claim or demand, it shall have the sole right to settle or otherwise dispose of such Claim or demand on such terms as it, in its sole discretion, shall deem appropriate; *provided*, *however*, that the consent of the Indemnified Party to the settlement or disposition of any Claim or demand shall be required if such settlement or disposition (A) shall result in any liability to, or equitable relief

against, the Indemnified Party, or (B) if the Indemnified Party is named in such Claim or demand, does not provide for a complete release of the Indemnified Party, which consent shall not be unreasonably withheld, delayed or conditioned; and

- (iii) the Indemnified Party shall have the right to participate jointly in the defense of such Claim or demand, but shall do so at its own cost not subject to reimbursement under Section 8.2 or 8.3, provided, that, the Indemnifying Party (to the extent the Indemnified Party is entitled to indemnification hereunder) shall bear the reasonable fees, costs and expenses of one counsel (and any specialized counsel, to the extent reasonably required) of the Indemnified Party, if such counsel determines, in writing, that a conflict of interest exists which makes representation by the same counsel as the Indemnifying Party inappropriate.
- (c) If the Indemnifying Party does not elect to take over the defense of a Claim or demand, then the Indemnified Party shall have the right, at the Indemnifying Party's expense (to the extent the Indemnified Party is entitled to indemnification hereunder), to contest, compromise or settle such Claim or demand in the exercise of its reasonable judgment; provided, however, that the consent of the Indemnifying Party to the compromise or settlement of any Claim or demand shall be required if such compromise or settlement (A) shall result in any liability to, or equitable relief against, the Indemnifying Party, or (B) does not provide for a complete release of the Indemnifying Party or the JV or its Subsidiaries, as applicable, which consent shall not be unreasonably withheld, delayed or conditioned.
- (d) The parties shall use commercially reasonable efforts to mitigate and minimize Losses for which indemnification is available under this <u>Article VIII</u> and shall act in good faith in responding to, defending against, settling or otherwise dealing with such Claims. Without limiting the generality of the foregoing, an Indemnified Party shall diligently pursue any available Claims against insurers who may have provided insurance coverage for any Losses and shall use its commercially reasonable efforts to pursue, or to assign to the Indemnifying Party, any Claims or rights it has against any Person that may reasonably be expected to reduce the Losses otherwise incurred by the Indemnified Party. Each Party agrees that it shall cooperate on a commercially reasonable basis with the other Party in the defense of any Claim or action.

8.6 Measure of Indemnification.

- (a) The amount of the Indemnifying Party's liability under this Agreement shall be determined after taking into account any applicable insurance proceeds received by the Indemnified Party including, in the case of Buyer, amounts received or recovered by the Company or any of the Company Subsidiaries under insurance policies or from other sources.
- (b) For all purposes of this <u>Article VIII</u>, "Losses" shall be net of any Tax benefit actually realized by such Indemnified Party or its Affiliates as a result of suffering such Losses.
- (c) Any Claim under <u>Sections 8.2(a)</u> or <u>8.3</u> required to be made on or prior to the expiration of the applicable survival period set forth in <u>Section 8.1</u> and not made, shall be irrevocably and unconditionally released and waived by the Party seeking

indemnification with respect thereto. It is the express intent of the Parties that, if the applicable period for a Claim pursuant to Sections 8.2(a) or 8.3 is shorter than the statute of limitations that would otherwise have been applicable to such Claim, then, by contract, the applicable statute of limitations with respect to such Claim shall be reduced to the shortened survival period contemplated hereby. The Parties further acknowledge that the time periods set forth in Section 8.1 for the assertion of Claims under this Agreement are the result of arms' length negotiation among the Parties and they intend for the time periods to be enforced as agreed by the Parties.

- (d) Notwithstanding anything to the contrary in this Agreement, any amounts payable pursuant to the indemnification obligations under this <u>Article VIII</u> shall be paid without duplication and in no event shall any Party hereto be indemnified under different provisions of this Agreement for Losses that have already been paid or otherwise taken into account under this Agreement. Without limiting the generality of the foregoing, Buyer shall make no Claim for indemnification in respect of any matter that is considered or taken into account in the final determination of any adjustment of the Estimated Total Equity Value pursuant to <u>Section 2.4</u>.
- (e) IN NO EVENT SHALL ANY INDEMNIFYING PARTY BE LIABLE TO ANY INDEMNIFIED PARTY UNDER THIS <u>ARTICLE VIII</u> FOR ANY CONSEQUENTIAL, INDIRECT, INCIDENTAL OR PUNITIVE DAMAGES, INCLUDING LOST PROFITS, LOST REVENUES, BUSINESS INTERRUPTION, COST OF CAPITAL OR LOSS OF BUSINESS REPUTATION OR OPPORTUNITY (BUT EXCLUDING, TO THE EXTENT REASONABLY FORESEEABLE BY THE PARTIES HERETO AS OF THE DATE HEREOF AND THE BASIS FOR SUCH LOSSES IS OF A RECURRING NATURE OR REPRESENTS A REDUCTION IN THE HISTORICAL CASH FLOW OR EARNINGS (INCLUDING BECAUSE OF AN INCREASE IN EXPENSES OR A DECREASE IN REVENUE) OF THE COMPANY OR THE COMPANY SUBSIDIARIES, WHICH REDUCTION WOULD REASONABLY BE EXPECTED TO REFLECT AN IMPAIRMENT OF THE COMPANY'S OR THE COMPANY SUBSIDIARIES' FUTURE CASH FLOWS OR EARNINGS, DIMINUTION IN VALUE), FOR ANY BREACH OR DEFAULT UNDER, OR ANY ACT OR OMISSION ARISING OUT OF OR IN ANY WAY RELATING TO, THIS AGREEMENT OR THE TRANSACTIONS, UNDER ANY FORM OF ACTION WHATSOEVER, WHETHER IN CONTRACT OR OTHERWISE (OTHER THAN INDEMNIFICATION FOR AMOUNTS PAID OR PAYABLE TO THIRD PARTIES IN RESPECT OF ANY THIRD PARTY CLAIM AS FINALLY DETERMINED BY A COURT OF COMPETENT JURISDICTION FOR WHICH INDEMNIFICATION HEREUNDER IS OTHERWISE REQUIRED).
- 8.7 Exclusivity of Indemnity. Subject to the provisions of Sections 2.4, the indemnification provided in this Article VIII shall be the sole and exclusive remedy after the Closing Date for Losses available to the Buyer Indemnified Parties and the Seller Indemnified Parties under this Agreement for misrepresentation or breach of any of the representations, warranties, covenants and agreements contained herein or in any other certificate delivered by the Parties pursuant hereto or any right, claim or action arising under, out of or relating to this Agreement, the negotiation or execution thereof, or the Transactions. Effective as of the

Closing, each Party expressly waives, releases and agrees not to make any claim against Seller or its Affiliates, in the case of Buyer, or Buyer or its Affiliates, in the case of any member of the Seller Group, except for indemnification claims made pursuant to this <u>Article VIII</u>, for the recovery of any costs or damages, whether directly or by way of contribution, or for any other relief whatsoever, under any applicable Laws, whether now existing or applicable or hereinafter enacted or applicable (including claims for breach of contract, failure of disclosure, tortious wrong or violation of securities Laws), *provided*, that, no Party shall be relieved or released from any liabilities or damages for Fraud. For the avoidance of doubt, following the Closing, nothing in this Agreement will limit any remedies available to a Party under the JV LLC Agreement, the JV Letter Agreement or the Newco LLC Agreement after the Closing. This <u>Section 8.7</u> shall survive the Closing indefinitely.

8.8 <u>Tax Treatment of Indemnity Payments</u>. The Parties agree to, unless otherwise prohibited by Law, (i) treat any indemnity payment made under this Agreement as an adjustment to the Purchase Price for all federal, state, local and foreign Tax purposes, and (ii) file all Tax Returns accordingly.

ARTICLE IX

TERMINATION OF AGREEMENT

- 9.1 <u>Termination</u>. This Agreement may not be terminated, except as follows:
 - (a) by mutual agreement of Buyer and Seller;
- (b) by either Buyer or Seller, upon prior written notice, if the Closing shall not have been consummated on or before 11:59 pm New York time on March 31, 2017, *provided*, either Buyer or Seller may extend such date by a three (3) month period (such date, including if and as extended, the "Outside Date") if the extending party is not in breach in any material respect of any of its obligations under this Agreement at the time of such extension; *provided*, *however*, that the Party whose conduct (including, in the case of Seller, the conduct of any member of the Seller Group) has been the cause of, or substantially resulted in, the failure of the Closing to have been consummated on or before the Outside Date may not be the terminating party;
- (c) by either Buyer or Seller, upon prior written notice, if any court of competent jurisdiction or other Governmental Entity in PRC or any jurisdiction material to the business of the Healthcare Portfolio shall have issued a final Order restraining, enjoining or otherwise prohibiting the consummation of the Transactions and such Order shall have become final and non-appealable;
- (d) by Seller, if Buyer fails to consummate the Closing within five (5) Business Days of the date on which Closing is otherwise required to be consummated pursuant to <u>Section 7.1</u>;
- (e) by Seller, if Buyer shall have breached or failed to perform any representation, warranty, covenant or agreement set forth in this Agreement that would give rise to the failure of a condition set forth in Section 7.3(a) or 7.3(b), which breach either (i) is not

curable prior to the Outside Date, or (ii) if curable prior to the Outside Date, has not been cured within the earlier of (A) thirty (30) calendar days after the receipt of notice thereof by Buyer from Seller or (B) three (3) Business Days before the Outside Date; provided, however, that the right to terminate this Agreement under this Section 9.1(e) shall not be available to Seller if any member of the Seller Group is in material breach of any representation, warranty, covenant or other agreement contained in this Agreement;

- (f) by Buyer, if a member of the Seller Group shall have breached or failed to perform any representation, warranty, covenant or agreement set forth in this Agreement that would give rise to the failure of a condition set forth in Section 7.4(a) or 7.4(b), which breach either (i) is not curable prior to the Outside Date, or (ii) if curable prior to the Outside Date, has not been cured within the earlier of (A) thirty (30) calendar days after the receipt of notice thereof by Seller from Buyer or (B) three (3) Business Days before the Outside Date; *provided*, *however*, that the right to terminate this Agreement under this Section 9.1(f) shall not be available to Buyer if Buyer is in material breach of any representation, warranty, covenant or other agreement contained in this Agreement; or
- (g) by Seller, upon written notice at any time following December 8, 2016, if Buyer shall not have (i) delivered to Seller the fully executed Loan Agreements or (ii) delivered (A) to Seller a fully executed and enforceable Letter of Credit for the benefit of Seller or (B) the Deposit Amount to the Escrow Agent in accordance with the terms of the Escrow Agreement.

9.2 Effect of Termination.

(a) <u>Survival After Termination</u>. If this Agreement terminates pursuant to <u>Section 9.1</u> and the Transactions are not consummated, this Agreement shall become null and void and have no further force or effect, all obligations of the parties hereunder shall terminate and there shall be no further liability or obligations hereunder on the part of the parties hereto, *except* that notwithstanding anything in this Agreement to the contrary, the provisions of <u>Section 6.4(c)</u> (*Financing*), <u>Section 6.5</u> (*Public Announcements*), <u>Section 6.7</u> (*Expenses*), <u>Section 8.7</u> (*Exclusivity of Indemnity*), this <u>Section 9.2</u> and <u>Article X</u> (*Miscellaneous*) shall survive any termination of this Agreement; *provided*, that, notwithstanding anything in this Agreement to the contrary but subject to <u>Section 9.2(e)(iii)</u>, no Party shall be relieved or released from any liabilities or damages for Fraud or that arise out of the Willful Breach of any provision of this Agreement prior to termination hereof.

(b) Escrow; Letter of Credit.

(i) If this Agreement is terminated by Seller pursuant to Section 9.1(d) or Section 9.1(e), then, the Seller shall have, as Seller's sole and exclusive remedy, the right to (A) receive the Deposit Amount from the Escrow Agent as liquidated damages (and in such circumstances Buyer shall, if applicable, join with Seller in the Deposit Release Instruction to the Escrow Agent; *provided*, that, if Buyer does not deliver an executed Deposit Release Instruction within five (5) Business Days of such termination, Seller shall be permitted to deliver the Deposit Release Instruction to the Escrow Agent without Buyer's countersignature and the Escrow Agent shall be authorized to release the Deposit Amount to

Seller pursuant to such Deposit Release Instruction) or (B) draw on the Letter of Credit for the amount of the Deposit Amount as liquidated damages, as applicable. Buyer and Seller agree that Seller's damages are difficult to ascertain and that in the event Seller terminates this Agreement in accordance with Section 9.1 and is entitled to receive the Deposit Amount pursuant to this Section 9.2(b)(i), the Deposit Amount is a fair approximation of the Seller's damages.

(ii) If this Agreement is terminated by Buyer pursuant to Section 9.1(f) and prior to such termination Buyer shall have entered into the Loan Agreement and (x) delivered to Seller the Letter of Credit or (y) delivered the Deposit Amount to the Escrow Agent, then (A) Buyer shall have the Deposit Amount returned to it by the Escrow Agent (and in such circumstances Seller shall join with Buyer in the Deposit Release Instruction to the Escrow Agent; provided, that, if Seller does not deliver an executed Deposit Release Instruction within five (5) Business Days of such termination, Buyer shall be permitted to deliver the Deposit Release Instruction to the Escrow Agent without Seller's countersignature and the Escrow Agent shall be authorized to release the Deposit Amount to Buyer pursuant to such Deposit Release Instruction) (if applicable), (B) as promptly as practicable (but in no event later than five (5) Business Days from the date of such termination), Buyer shall receive, as Buyer's sole and exclusive remedy (subject to subclauses (A) and (C) of this Section 9.2(b)(ii)), the Termination Fee from the Seller as liquidated damages and (C) Buyer shall have the right to terminate the Letter of Credit (if applicable). Buyer and Seller agree that Buyer's damages are difficult to ascertain and that in the event Buyer terminates this Agreement in accordance with Section 9.1 and is entitled to receive the Deposit Amount pursuant to this Section 9.2(b)(ii), the Termination Fee is a fair approximation of the Buyer's damages.

(iii) If this Agreement is terminated (A) pursuant to Section 9.1(a) or Section 9.1(g) or (B) by either Seller or Buyer pursuant to Section 9.1(b) or Section 9.1(c), in respect of matters where the Deposit Amount or the Expense Reimbursement Amount is not payable to Seller pursuant to Section 9.2(b)(i) or Section 9.2(c)(i), respectively, then Buyer shall have the right to have the Deposit Amount returned to it by the Escrow Agent (and in such circumstances Seller shall join with Buyer in the Deposit Release Instruction to the Escrow Agent; provided, that, if Seller does not deliver an executed Deposit Release Instruction within five (5) Business Days of such termination, Buyer shall be permitted to deliver the Deposit Release Instruction to the Escrow Agent without Seller's countersignature and the Escrow Agent shall be authorized to release the Deposit Amount to Buyer pursuant to such Deposit Release Instruction) or to terminate the Letter of Credit, as applicable.

(iv) If this Agreement is terminated by either Seller or Buyer pursuant to Section 9.1(b) or Section 9.1(c), in respect of matters where the Expense Reimbursement Amount is payable to Seller pursuant to Section 9.2(c)(i), then, upon (and subject to) receipt of the Expense Reimbursement Amount by Seller, Buyer shall have the right to have the remainder of the Deposit Amount returned to it by the Escrow Agent (and in such circumstances Seller shall join with Buyer in the Deposit Release Instruction to the Escrow Agent provided, that, if Seller does not deliver an executed Deposit Release Instruction within five (5) Business Days of such termination, Buyer shall be permitted to deliver the Deposit Release Instruction to the Escrow Agent without Seller's countersignature and the Escrow Agent shall be authorized to release the remainder of the Deposit Amount to Buyer pursuant to such Deposit Release Instruction) or to terminate the Letter of Credit, as applicable.

- (c) If this Agreement is terminated by either Seller or Buyer pursuant to <u>Section 9.1(b)</u> or <u>Section 9.1(c)</u>:
- (i) solely with respect to (A) in the case of Section 9.1(b), if the failure to close prior to the Outside Date arises from or is related to the failure of Buyer to obtain any of the PRC Approvals prior to the Outside Date, or (B) in the case of Section 9.1(c), an Order from a Governmental Entity in the PRC, and prior to such termination Buyer shall have entered into the Loan Agreements and (x) delivered or cause to be delivered to Seller a fully executed Letter of Credit or (y) delivered the Deposit Amount to the Escrow Agent, then Seller shall have, as Seller's sole and exclusive remedy, the right to (A) receive the Expense Reimbursement Amount from the Escrow Agent as liquidated damages (and in such circumstances Buyer shall, if applicable, join with Seller in the Deposit Release Instruction to the Escrow Agent; provided, that, if Buyer does not deliver an executed Deposit Release Instruction within five (5) Business Days of such termination, Seller shall be permitted to deliver the Deposit Release Instruction to the Escrow Agent without Buyer's countersignature and the Escrow Agent shall be authorized to release the Expense Reimbursement Amount to Seller pursuant to such Deposit Release Instruction) or (B) draw on the Letter of Credit for the amount of the Expense Reimbursement Amount as liquidated damages, as applicable.
- (ii) solely with respect to (A) in the case of Section 9.1(b), if the failure to close prior to the Outside Date arises from or is related to the failure of Seller to obtain any Governmental Approval required on the part of Seller or any of its Affiliates to consummate the Transactions prior to the Outside Date (which, for the avoidance of doubt, the Parties acknowledge, does not include any of the PRC Approvals), or (B) in the case of Section 9.1(c), an Order from a Governmental Entity in the United States or the United Kingdom, then, as promptly as practicable (but in no event later than five (5) Business Days from the date of such termination), and prior to such termination Buyer shall have entered into the Loan Agreement and (x) delivered or cause to be delivered to Seller a fully executed Letter of Credit or (y) delivered the Deposit Amount to the Escrow Agent, then Seller shall pay or cause to be paid, to Buyer by wire transfer of immediately available funds to the account or accounts designated in writing by Buyer, as Buyer's sole and exclusive remedy, the Expense Reimbursement Amount in Dollars as liquidated damages.

Buyer and Seller agree that Seller's and Buyer's damages are difficult to ascertain and that in the event Seller or Buyer terminates this Agreement in accordance with <u>Section 9.1</u> and is entitled to receive the Expense Reimbursement Amount pursuant to this Section 9.2(c), the Expense Reimbursement Amount is a fair approximation of the Seller's or Buyer's damages, as applicable.

(d) Notwithstanding anything to the contrary in this Agreement but solely for the purposes of Section 9.2(b)(i) and Section 9.2(c), the Deposit Amount payable to Seller pursuant to Section 9.2(b)(i) and Section 9.2(c) shall be (i) the Deposit Amount in the event that the Buyer exercises its option to cause the fully executed Letter of Credit to be provided under Section 2.2(b) (in which case the remaining provisions of this Section 9.2(d) shall not apply), and (ii) an amount equal to the lesser of (A) the Deposit Amount (the "Base Amount") and (B) the maximum amount, if any, that can be paid to Seller without causing Seller to fail to meet the requirements of Section 856(c)(2) and (3) of the Code (the "REIT"

Requirements") for such year determined as if the payment of such amount did not constitute Qualifying Income, as determined by independent accountants to Seller (the amount determined by this clause (ii)), in the event that the Buyer exercises its option to deliver the Deposit Amount to the Escrow Agent under Section 2.2(b) (in which case the remaining provisions of this Section 9.2(d) shall apply). Notwithstanding the foregoing, to the extent Seller receives Tax Guidance providing that Seller's receipt of the Base Amount would either constitute Qualifying Income or would be excluded from gross income within the meaning of the REIT Requirements, the Deposit Amount shall, for purposes of $\underline{\text{Section 9.2(b)}(i)}$ and $\underline{\text{Section 9.2(c)}}$ be an amount equal to the Base Amount, and the Escrow Agent shall, upon receiving notice that Seller has received the Tax Guidance, release to Seller the unpaid Base Amount within two (2) Business Days. In the event that Seller is not able to receive the full Base Amount due to the above limitations, the Escrow Agent shall retain the unpaid Base Amount pursuant to the terms of the Escrow Agreement and shall not release any portion thereof to Seller unless and until Seller receives either one or a combination of the following once or more often: (x) a letter from Seller's independent accountant indicating the maximum amount that can be paid at that time to Seller without causing Seller to fail to meet the REIT Requirements (calculated as described above) or (y) Tax Guidance, in either of which events, the Escrow Agent shall release to Seller the lesser of the unpaid Base Amount or the maximum amount stated in the letter referred to in (x) above, within two (2) Business Days after the Escrow Agent has been notified thereof. The obligation of Escrow Agent to pay any unpaid portion of the Deposit Amount shall terminate on December 31 following the date which is five (5) years from the date of this Agreement. Amounts remaining in escrow after the obligation of the Escrow Agent to release the Deposit Amount to Seller terminates shall be returned to Buyer. "Qualifying Income" shall mean income described in Sections 856(c)(2)(A)-(H), 856(c)(3)(A)-(I) and 856(c)(5)(J)(ii) of the Code. "Tax Guidance" shall mean a reasoned opinion from outside counsel or a ruling from the IRS.

(e) Exclusive Remedy.

(i) Buyer and the Seller Group acknowledge that (A) the liquidated damages and other provisions of this <u>Section 9.2</u> are an integral and material part of the Transactions, (B) without these agreements, Buyer and the Seller Group would not enter into this Agreement and (C) any amount payable pursuant to this <u>Section 9.2</u> does not constitute a penalty.

(ii) Notwithstanding anything to the contrary in this Agreement, in the event that this Agreement is terminated and (A) Seller has the right to receive (x) the Deposit Amount pursuant to Section 9.2(b)(i) or (y) the Expense Reimbursement Amount pursuant to Section 9.2(c)(i) or (B) Buyer has the right to receive (x) the Termination Fee pursuant to Section 9.2(b)(ii) or (y) the Expense Reimbursement Amount pursuant to Section 9.2(c)(ii), Seller's right to receive the Deposit Amount or Expense Reimbursement Amount, or Buyer's right to receive the Termination Fee or Expense Reimbursement Amount, as applicable, shall be the sole and exclusive remedy of Seller and its Affiliates against Buyer and its Affiliates or of Buyer and its Affiliates against Seller and its Affiliates, as applicable, for any loss suffered as a result of any breach of this Agreement (other than any provisions that survive termination of this Agreement pursuant to Section 9.2(a)), or the failure of the Transactions to be consummated, whether in equity or at law, in contract, in tort or otherwise, and upon such termination and Seller's receipt of the Deposit Amount or Expense Reimbursement Amount, or Buyer's receipt

of the Termination Fee or Expense Reimbursement Amount, as applicable, none of Buyer or its Affiliates or Seller or its Affiliates, as applicable, shall have any further liability or obligation, including consequential, indirect or punitive damages, relating to or arising out of any breach of this Agreement, or the failure of the Transactions to be consummated, or in respect of any oral representation made or alleged to be have been made in connection herewith or therewith, whether in equity or at law, in contract, in tort or otherwise. For the avoidance of doubt, the foregoing shall not impair the rights of the parties to obtain any remedies pursuant to the Confidentiality Agreement or Seller's right to sue for damages.

(iii) Notwithstanding anything to the contrary in this Agreement, in the event that this Agreement is terminated and prior to such termination Buyer shall not have delivered or cause to be delivered to Seller a fully executed Letter of Credit or delivered the Deposit Amount to the Escrow Agent, neither Buyer or its Affiliates, on the one hand, nor Seller or its Affiliates, on the other, shall have any remedy against the other for any loss suffered as a result of any breach of this Agreement or the failure of the Transactions to be consummated, whether in equity or at law, in contract, in tort or otherwise, including with respect to the right to receive the Deposit Amount, the Termination Fee or the Expense Reimbursement Amount, as applicable.

ARTICLE X

MISCELLANEOUS

10.1 <u>Dispute Resolution; Venue</u>.

(a) Any claim, controversy or dispute arising out of or relating to this Agreement or the Transactions shall be determined by arbitration administered by the International Chamber of Commerce ("ICC") and resolved in accordance with the Rules of Arbitration of the ICC in force at the relevant time as may be amended by this Section 10.1 (the "Rules"). The place of arbitration shall be New York City, New York, USA. The official language of the arbitration shall be English and the tribunal shall consist of three arbitrators (each, an "Arbitrator"). The claimant(s), irrespective of number, shall nominate jointly one Arbitrator; the respondent(s), irrespective of number, shall nominate jointly one Arbitrator, and a third Arbitrator will be nominated jointly by the first two Arbitrators and shall serve as chairman of the arbitration tribunal. In the event the claimant(s) or respondent(s) or the first two Arbitrators shall fail to nominate or agree the joint nomination of an Arbitrator or the third Arbitrator within the time limits specified by the Rules, such Arbitrator shall be appointed promptly by the ICC. The arbitration tribunal shall have no authority to award punitive or punitive-type damages. The award of the arbitration tribunal shall be final and binding upon the disputing parties. Any party to an award may apply to any court of competent jurisdiction for enforcement of such award and, for purposes of the enforcement of such award, the Parties irrevocably and unconditionally submit to the jurisdiction of any court of competent jurisdiction and waive any defenses to such enforcement based on inconvenient forum. Each Party expressly consents and submits to the jurisdiction of any State or Federal court in the County of New York, State of New York, in connection with, arising out of or relating to any arbitration hereunder and waives and agrees not to assert, by way of motion, as a defense or otherwise, in any such

proceeding, that it is not subject personally to the jurisdiction of such court or that the proceeding is brought in an inconvenient forum, or that the venue of the proceeding is improper.

- (b) Notwithstanding the foregoing, the Parties hereby consent to and agree that in addition to any recourse to arbitration as set out in this <u>Section 10.1</u>, any Party may, to the extent permitted under the rules and procedures of the ICC, seek an interim injunction or other form of relief from the ICC as provided for in its Rules. Such application shall also be governed by, and construed in accordance with, the laws of the State of New York.
- (c) By agreeing to arbitration, (i) the Parties do not intend to deprive any court of competent jurisdiction of its ability to issue any form of provisional or equitable remedy, including but not limited to a preliminary injunction, injunction, action for specific performance or attachment in aid of the arbitration, or any interim or conservatory measure and (ii) a request for a provisional remedy or interim or conservatory measure by a Party to a court shall not be deemed a waiver of this agreement to arbitrate.
- 10.2 Notices. Any notice pursuant to this Agreement shall be given in writing by (a) personal delivery, or (b) reputable overnight delivery service with proof of delivery, or (c) United States Mail, postage prepaid, registered or certified mail, return receipt requested, or (d) legible facsimile transmission or as a PDF or similar attachment to an electric mail transmission. Any such notice shall be sent to the intended addressee at the address set forth below, or to such other address or to the attention of such other person as the addressee shall have designated by written notice sent in accordance herewith, and shall be deemed to have been given either at the time of personal delivery, or, in the case of expedited delivery service or mail, as of the date of first attempted delivery at the address and in the manner provided herein, or, in the case of facsimile transmission or by electronic mail attachment, as of the date of the facsimile transmission or electronic mail. Unless changed in accordance with the preceding sentence, the addresses for notices given pursuant to this Agreement shall be as follows:

(a) if to Buyer, to:

Taikang Asset Management Co., Ltd. 10/F, Taikang Life Building 156 Fuxingmennei Street Xicheng District Beijing 100031, China

Attention: Zuyu Tan Yan He

Lin Xu

Email: tanzy02@taikangamc.com.cn

heyan21@taikangamc.com.cn xulin16@taikangamc.com.cn

with copies (which shall not constitute notice) to:

Shearman & Sterling LLP 599 Lexington Avenue New York, New York 10022

Attention: John A. Marzulli, Jr., Esq.

Stephanie Tang, Esq. Lisa M. Brill, Esq.

Facsimile: (646) 848-8590

Email: jmarzulli@shearman.com

stephanie.tang@shearman.com

lbrill@shearman.com

(b) if to any member of the Seller Group, to:

c/o NorthStar Realty Finance Corp.

399 Park Avenue

New York, New York 10022 Attention: Jonathan Langer

Legal Department

Email: langer@nsamgroup.com

legal@nsamgroup.com

with copies (which shall not constitute notice) to:

Paul, Weiss, Rifkind, Wharton & Garrison LLP

1285 Avenue of the Americas New York, NY 10019-6064

Attention: Robert B. Schumer, Esq.

Harris B. Freidus, Esq Ellen N. Ching, Esq.

Facsimile: (212) 492-0097

Email: rschumer@paulweiss.com

hfreidus@paulweiss.com eching@paulweiss.com

10.3 Entire Agreement. This Agreement together with the Confidentiality Agreement, the Transaction Documents and any certificates executed in connection with the consummation of the Transactions, contain the entire agreement among the parties with respect to the sale and purchase of the Purchased Interests and supersede all prior and contemporaneous agreements, negotiations, correspondence, undertakings and communications of the Parties or their representatives, written or oral, with respect thereto. The Parties have voluntarily agreed to define their rights, liabilities and obligations respecting the subject matter hereof exclusively in contract pursuant to the express terms and provisions of this Agreement and the Parties expressly disclaim that they are owed any duties or are entitled to any remedies not expressly set forth in this Agreement. Furthermore, the Parties each hereby acknowledge that this Agreement embodies the justifiable expectations of sophisticated parties derived from arm's-length

negotiations; the parties specifically acknowledge that no Party has any special relationship with another Party that would justify any expectation beyond that of ordinary parties in an arm's-length transaction.

- 10.4 <u>Waivers and Amendments</u>. This Agreement may be amended, superseded, canceled, renewed or extended, and the terms hereof may be waived, only by a written instrument signed by Buyer and Seller (on behalf of any member of the Seller Group) or, in the case of a waiver, by the Party waiving compliance. No delay on the part of any Party in exercising any right, power or privilege hereunder shall operate as a waiver thereof, nor shall any waiver on the part of any Party of any such right, power or privilege, nor any single or partial exercise of any such right, power or privilege, preclude any further exercise thereof or the exercise of any other such right, power or privilege.
- 10.5 <u>Governing Law</u>. This Agreement and all claims and causes of action arising out of the negotiation and execution of this Agreement shall be governed by and construed in accordance with the Laws of the State of New York without regard to any conflict of Laws rules thereof that might indicate the application of the Laws of any other jurisdiction.

10.6 <u>Binding Effect; Assignment.</u>

- (a) This Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors and assigns. Except as provided in Section 10.6(c), this Agreement is not assignable by any Party without the prior written consent of the Seller, in the case of Buyer, or Buyer, in the case of any member of the Seller Group. Notwithstanding the foregoing, Buyer may assign this Agreement to one or more of its Affiliates; *provided*, that, any such assignment shall not relieve Buyer of any of its obligations hereunder; *provided*, *further*, that Buyer shall not make any assignment which could delay the receipt of any PRC Approvals or affect Buyer's ability to rely on any PRC Approvals obtained prior to such assignment.
- (b) Subject to the terms of the applicable Side Letter, this Agreement shall be effective on the date and upon completion of the recordal process with the NDRC in accordance with PRC laws (such date, the "<u>Effective Date</u>").
- (c) Notwithstanding anything in this Agreement to the contrary, Seller (determined as of the date hereof) shall have the right to sell, assign or transfer all or a portion of its JV HoldCo Interests and its rights with respect thereto hereunder to a wholly-owned Subsidiary for purposes of effecting steps 2, 3 and 5 of the Pre-Closing Restructuring, as set forth in Exhibit A hereto.
- 10.7 <u>Severability of Provisions</u>. If any provision or any portion of any provision of this Agreement shall be held invalid or unenforceable, the remaining portion of such provision and the remaining provisions of this Agreement shall not be affected thereby. If the application of any provision or any portion of any provision of this Agreement to any Person or circumstance shall be held invalid or unenforceable, the application of such provision or portion of such provision to Persons or circumstances other than those as to which it is held invalid or unenforceable shall not be affected thereby.

- 10.8 <u>Disclosure Schedule</u>. Any disclosure in any section of the disclosure schedule delivered by the Company to Buyer on the date hereof concurrently with the entry into this Agreement (the "<u>Disclosure Schedule</u>") shall be deemed disclosed with respect to any other Section of this Agreement (in addition to the Section referenced in such schedule) to the extent that it is reasonably apparent from the wording of such disclosure that such disclosure is applicable to such other Section).
- 10.9 <u>Counterparts</u>. This Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered shall be an original, but all such counterparts together shall constitute one and the same instrument. Each counterpart may consist of a number of copies hereof each signed by less than all, but together signed by all, of the parties hereto. Facsimile signatures or signatures received as a pdf attachment to electronic mail shall be treated as original signatures for all purposes of this Agreement.
- 10.10 <u>No Personal Liability</u>. This Agreement (and each agreement, certificate and instrument delivered pursuant hereto) shall not create or be deemed to create or permit any personal liability or obligation on the part of any partner, member, officer, director, employee, agent, Representative or investor of any Party hereto, except as expressly set forth herein.
- 10.11 <u>No Third Party Beneficiaries</u>. Except as otherwise provided in <u>Section 6.4(c)</u> (*Financing*), <u>Article VIII</u> (*Indemnity*) and <u>Section 10.10</u> (*No Personal Liability*), no provision of this Agreement is intended to, or shall, confer any third party beneficiary or other rights or remedies upon any Person other than the Parties hereto.

[Remainder of page intentionally left blank]

DERWOOD LIMITED

By: /s/ Timothy Zhang

Name: Timothy Zhang

Title: Director

By: /s/ Yan He

Name: Yan He Title: Director

NORTHSTAR REALTY FINANCE LIMITED PARTNERSHIP

By: /s/ Jonathan Langer

Name: Jonathan Langer Title: Chief Executive Officer

NORTHSTAR HEALTHCARE JV HOLDINGS, LLC

By: /s/ Jonathan Langer

Name: Jonathan Langer Title: Chief Executive Officer

NORTHSTAR TK HEALTHCARE REIT, LLC

By: /s/ Jonathan Langer

Name: Jonathan Langer Title: Chief Executive Officer

NORTHSTAR TK HEALTHCARE OPERATING COMPANY LLC

By: /s/ Jonathan Langer

Name: Jonathan Langer Title: Chief Executive Officer

NORTHSTAR HEALTHCARE JV, LLC

By: /s/ Jonathan Langer

Name: Jonathan Langer Title: Chief Executive Officer

NRFC HEALTHCARE HOLDING COMPANY, LLC

By: /s/ Jonathan Langer

Name: Jonathan Langer
Title: Chief Executive Officer

[Signature Page 2 of 2 to Purchase and Sale Agreement]

COLONY NORTHSTAR

UNAUDITED PRO FORMA CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

<u>Index</u>

	<u>Pag</u>
Unaudited Pro Forma Condensed Consolidated Financial Statements	<u>2</u>
Unaudited Pro Forma Condensed Consolidated Balance Sheet as of September 30, 2016	
	<u>3</u>
Unaudited Pro Forma Condensed Consolidated Statement of Operations for the nine months ended September 30, 2016	<u>4</u>
Unaudited Pro Forma Condensed Consolidated Statement of Operations for the year ended December 31, 2015	<u>5</u>
Notes to the Unaudited Pro Forma Condensed Consolidated Financial Statements	6

COLONY NORTHSTAR UNAUDITED PRO FORMA CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

The following unaudited pro forma condensed consolidated financial statements and notes set forth the impact of the Mergers and related transactions on the historical financial condition and results of operations of Colony, NSAM and NRF. The unaudited pro forma condensed consolidated financial statements were prepared using the acquisition method of accounting for business combinations with the Mergers accounted for as a reverse acquisition. In the Mergers, NSAM is the legal acquirer while Colony is considered to be the accounting acquirer for financial reporting purposes. The unaudited pro forma condensed consolidated financial statements give effect to: (i) completion of the Mergers; (ii) the NRF management agreement ceasing to exist; and (iii) completion of certain sales initiatives by NRF, which we refer to as NRF Sales Initiatives, that were executed or are under contract as of the date of this joint proxy statement/prospectus.

The unaudited pro forma condensed consolidated balance sheet consolidates the historical consolidated balance sheets of Colony, NSAM and NRF, giving effect to the Mergers and related transactions and NRF Sales Initiatives as if they had been consummated on September 30, 2016. The unaudited pro forma condensed consolidated statements of operations for the nine months ended September 30, 2016 and year ended December 31, 2015 consolidates the historical consolidated statements of operations of Colony, NSAM and NRF, giving effect to the Mergers and related transactions and NRF Sales Initiatives as if they had been consummated on January 1, 2015, the beginning of the earliest period presented. The unaudited pro forma condensed consolidated financial statements are derived from and should be read in conjunction with the historical consolidated financial statements and notes thereto included in each of Colony's, NSAM's and NRF's respective Annual Reports on Form 10-K for year ended December 31, 2015, as amended, and each of Colony's, NSAM's and NRF's respective Quarterly Reports on Form 10-Q for the nine months ended September 30, 2016, which are incorporated by reference into this joint proxy statement/prospectus. Refer to "Where You Can Find More Information; Incorporation by Reference" beginning on page [—] of this joint proxy statement/prospectus.

The unaudited pro forma condensed consolidated financial statements reflect the assets, liabilities and non-controlling interests of NSAM and NRF at their estimated fair value as of September 30, 2016 based upon a preliminary allocation of the consideration to be received by Colony stockholders, NSAM stockholders and NRF stockholders, respectively, in connection with the Mergers, which we refer to as the merger consideration. A final determination of the fair value and allocation of the merger consideration will be based upon the actual assets, liabilities and non-controlling interests as of the date of completion of the Mergers. The value of the per share consideration will be determined based on the trading price of Colony common stock and NRF preferred stock at the time of the completion of the Mergers. Accordingly, the estimated fair value and allocation of the merger consideration are subject to adjustment and may vary significantly from the actual fair value and allocation of the merger consideration upon completion of the Mergers, if completed. As of the date of this joint proxy statement/prospectus, the Companies expect the closing of the Mergers will occur in January 2017.

The unaudited pro forma condensed consolidated financial statements are presented for illustrative purposes only and are not necessarily indicative of the financial position or results of operations of the Companies had the Mergers and related transactions and NRF Sales Initiatives been completed as of the beginning of the earliest period presented, nor indicative of future results of operations or future financial position of the combined company. The unaudited pro forma condensed consolidated financial statements do not reflect the costs of any integration activities or full benefits that may result from realization of future cost savings from operating efficiencies, revenue or other incremental synergies expected to result from the Mergers. The unaudited pro forma condensed consolidated financial statements reflect management's best estimate based on available information and may be revised as additional information becomes available and as additional analyses are performed.

UNAUDITED PRO FORMA CONDENSED CONSOLIDATED BALANCE SHEET September 30, 2016 (In thousands)

		Historical	(Note 2))	Pro Forma Adjustments			s (Note 4)	Colony NorthSta			
	Colony	NSA	м	NRF			NRF Sales nitiatives(A)	A	Merger djustments ^(B)		Fair Value Adjustment ^(C)		Pro Forma Consolidated
Assets									<u>, </u>	•			
Cash and cash equivalents	\$ 440,173	\$ 8	35,593	\$ 725,3	360	\$	1,317,920	\$	(1,110,991)	(1)(2)(3)	s –	\$	1,458,055
Restricted cash	148,396		26,599	180,0			(4,986)		_		_		350,077
Operating real estate, net	3,294,122		_	7,371,9	996		_		_		1,710,792		12,376,910
Real estate debt investments, net	3,685,654		_	348,5			_		_		544		4,034,737
Real estate debt investments, held for sale	56,357		_		_		_		_		_		56,357
Investments in private equity funds, at fair value	1,507		_	484,8	376		_		_		_		486,383
Investments in unconsolidated ventures	1,004,388	g	97,107	161,7			_		(39,443)	(2)	16,250		1,240,046
Real estate securities, available for sale	23,882		_	526,9			_		_		_		550,848
Securities, at fair value	_	3	88,438		_		_		(35,741)	(4)	_		2,697
Due from affiliates	13,718		9,753	1.8	388		_		(46,939)		_		78,420
Goodwill	680,127		13,328	44,7			_		_		94,806		1,063,028
Intangible assets, net	305,204		3,728	298,9			_		(1,800,000)	(6)	2,653,591		1,661,473
Assets of properties held for sale	216,339		_	2,653,9			(2,533,088)		(2,000,000)		(11,775)		325,435
Other assets	276,774		11,784	565,7			(208,260)		(5,053)	(7)	12,033		683,054
Total assets	\$ 10,146,641	· · · · · · · · · · · · · · · · · · · 	16,330	\$ 13,364,8		\$	(1,428,414)	\$	(3,038,167)		\$ 4,476,241	\$	24,367,520
	0 10,110,011	-	.0,550	15,501,6		_	(1, 120, 111)	=	(5,050,107)		J 1, 17 0,2 11	·	2 1,507,520
Liabilities	r 2.240.024	.		¢ 6000	207	¢	(602.221)	¢			¢ (62.220)		0.416.410
Mortgage and other notes payable	\$ 2,249,834	\$		\$ 6,922,0		\$	(692,231)	\$			\$ (63,220)		8,416,410
Credit facilities and term borrowings	486,176	46	68,679	420,4			_		(889,087)	(8)	28,821		514,998
Convertible senior notes	592,382		_	27,3			_		_		2,085		621,823
Securitization bonds payable	632,828		_	257,8			_		_		_		890,705
Junior subordinated notes, at fair value	_		_	191,1			_		_		_		191,175
Accounts payable and other liabilities	293,544	Š	91,155	205,1	142		7,321		97,786	(9)	177,080		872,028
Due to affiliates—contingent consideration	39,350		_		_		_		_		_		39,350
Due to related parties	_		_	46,9			_		(46,939)		_		_
Intangible liabilities, net	22,791		_	113,9	967		_		(1,800,000)	(6)	1,830,124		166,882
Dividends payable	65,924		_		_		_		_		_		65,924
Liabilities of assets held for sale	112,266		_	1,502,6			(1,408,179)		_		(41,051)		165,695
Derivative liabilities, at fair value	8,677			302,3	316	_	_		_		_		310,993
Total liabilities	4,503,772	55	59,834	9,989,8	367		(2,093,089)		(2,638,240)		1,933,839		12,255,983
Commitments and contingencies													
Redeemable non-controlling interests	_	7	75,149		_		_		_		_		75,149
Equity													
Performance common stock	_		52		_		_		_		_		52
Preferred stock	606,950		_	939,1	118		_		_		30,340		1,576,408
Common stock	1,139		1,890	1,8	307		_		715	(10)	_		5,551
Additional paid-in capital	2,432,716	24	16,171	5,116,1	100		_		(2,539,594)	(11)	2,285,172		7,540,565
Accumulated other comprehensive income (loss)	(23,897)		(210)	(63,7	709)		_		63,919	(11)	_		(23,897)
Retained earnings (accumulated deficit)	(183,585)	(3	88,554)	(2,891,1	153)		384,044		2,112,267	(12)	_		(616,981)
Total stockholders' equity	2,833,323	20	9,349	3,102,1	163		384,044		(362,693)		2,315,512		8,481,698
Non-controlling interests—investments	2,406,753		_	241,0	061		280,631		_		226,890		3,155,335
Non-controlling interests—operating partnership	402,793		1,998	31,7	798				(37,234)	(11)			399,355
Total equity	5,642,869	21	11,347	3,375,0)22		664,675		(399,927)		2,542,402		12,036,388
Total liabilities, redeemable non-controlling interests and equity	\$ 10,146,641	\$ 84	16,330	\$ 13,364,8	389	\$	(1,428,414)	\$	(3,038,167)		\$ 4,476,241	\$	24,367,520

UNAUDITED PRO FORMA CONDENSED CONSOLIDATED STATEMENT OF OPERATIONS Nine Months Ended September 30, 2016 (In thousands, except per share data)

			Hist	torical (Note 2))		Pro Forma Adjustments (Note 5)						Col	lony NorthStar	
	-	Colony		NSAM		NRF		NRF Sales Initiatives ^(D)	A	Merger .djustments ^(E)		Fair Value Adjustment ^(F)			Pro Forma Consolidated
Revenues											•				
Rental and escalation income	\$	254,640	\$	_	\$	527,252	\$	(265,336)	\$	_		\$ (16,371)	(8)	\$	500,185
Hotel operating income		24,830		_		636,283		_		_		_			661,113
Resident fee income		_		_		219,193		_		_		_			219,193
Interest income		291,496		_		115,117		(13,524)		_		_			393,089
Fee income		49,347		276,339		_		_		(139,955)	(1)	_			185,731
Selling commission and dealer manager fees, related parties		_		15,115		_		_		_		_			15,115
Other income		10,071		7,569		14,747		(6,221)		20,130	(1)	_			46,296
Total revenues		630,384		299,023		1,512,592		(285,081)		(119,825)		(16,371))		2,020,722
Expenses															
Management fee		_		_		139,955		_		(139,955)	(1)	_			_
Interest expense		126,635		18,968		362,052		(76,281)		(39,635)	(2)	(1,197	(9)		390,542
Property operating expenses		89,469		_		708,934		(95,278)		_		(543)	(10)		702,582
Commission expense		_		14,025		_		_		_		_			14,025
Other expense—investment and servicing expenses		17,448		5,461		20,933		(1,424)		_		_			42,418
Transaction costs		18,638		32,127		15,590		(186)		(46,114)	(1)	_			20,055
Impairment losses		5,461		_		75,506		_		_		_			80,967
Provision for loan losses		17,412		_		7,974		(3,051)		_		_			22,335
General and administrative expenses															
Compensation expense		80,689		107,547		23,295		_		(9,215)	(1)	_			202,316
Other general and administrative expenses		38,760		31,180		12,708		_		4,003	(1)	_			86,651
Total general and administrative expenses		119,449		138,727		36,003		_		(5,212)		_			288,967
Depreciation and amortization		129,276		7,355		260,287		(33,106)		_		94,621	(11)		458,433
Total expenses		523,788		216,663		1,627,234		(209,326)		(230,916)		92,881			2,020,324
Other income (loss)															
Unrealized gain (loss) on investments and other		_		(10,197)		(269,052)		_		10,475	(3)	_			(268,774)
Realized gain (loss) on investments and other		68,114		(874)		(11,768)		(36,914)		_		_			18,558
Other gain (loss), net		18,270		_		_		_		_		_			18,270
Income (loss) before equity in earnings (losses) of unconsolidated ventures and income tax benefit (expense)		192,980		71,289		(395,462)		(112,669)		121,566		(109,252	_		(231,548)
Equity in (loss) income of unconsolidated ventures		72,226		(5,094)		101,838		(10,799)		(3,526)	(4)	(100,202	,		154,645
Income tax benefit (expense)		865		(9,331)		(12,329)		3,162		, , , , , , , , , , , , , , , , , , ,	(5)	_			(30,325)
Income (loss) from continuing operations	-	266,071		56,864	_	(305,953)	_	(120,306)		105,348		(109,252	_		(107,228)
Redeemable non-controlling interests		_		2,991		_		(120,000)		_		(100,202	,		2,991
Non-controlling interests—investments		130,508		_		(4,423)		(3,602)		(7,373)	(6)	(1,475	(12)		113,635
Non-controlling interests—operating partnership		15,528		533		(3,537)		_		(30,276)			,		(17,752)
Preferred stock dividends		36,066		_		63,178		_		_		_			99,244
Net income (loss) from continuing operations attributable	• •		¢	52 240	¢		¢	(116.704)	¢	142 007		\$ (107.777	_	¢	
to common stockholders	\$	83,969	\$	53,340	\$	(361,171)	\$	(116,704)	\$	142,997		\$ (107,777)	_	\$	(305,346)
Earnings (loss) per share:(G)															
Basic	\$	0.73	\$	0.28	\$	(2.00)								\$	(0.56)
Diluted	\$	0.73	\$	0.28	\$	(2.00)								\$	(0.55)
Weighted average number of shares:(G)															
Basic		112,133		183,251		180,803								_	550,078
Diluted		112,133		185,083		182,664									551,916
		_	_		_										

Refer to accompanying notes to unaudited pro forma condensed consolidated financial statements.

UNAUDITED PRO FORMA CONDENSED CONSOLIDATED STATEMENT OF OPERATIONS Year Ended December 31, 2015 (In thousands, except per share data)

			Histo	rical (Note 2)				I	Pro Fo	orma Adjustment	its (Note 5)				6.1	N. d.C.
	Colo			NSAM		NRF		NRF Sales		Merger Adjustments(E)		Fair Va Adjustme		-	I	ony NorthStar Pro Forma onsolidated
Revenues														•		
Rental and escalation income	\$ 24	44,823	\$	_	\$	732,425	\$	(381,981)	\$	_		\$	46,630	(8)	\$	641,897
Hotel operating income		55,048		_		784,151		_		_			_			839,199
Resident fee income		_		_		271,394		_		_			_			271,394
Interest income	4	17,305		_		227,483		(45,451)		_			_			599,337
Fee income	(65,813		307,988		_		_		(198,695)	(1)		_			175,106
Selling commission and dealer manager fees, related parties		_		126,907		_		_		_			_			126,907
Other income		11,382		926		29,466		(10,339)		36,484	(1)		_			67,919
Total revenues	75	94,371		435,821		2,044,919		(437,771)		(162,211)			46,630	_		2,721,759
Expenses														-		
Management fee		15,062		_		198,695		_		(198,695)	(1)		_			15,062
Interest expense		33,094		778		495,086		(109,499)		(36,035)			(2,673)	(9)		480,751
Property operating expenses		17,713		_		915,701		(150,954)					453			882,913
Commission expense		_		117,390		_		_		_			_			117,390
Other expense—investment and servicing expenses		23,369		1,657		26,607		(11,992)		_			_			39,641
Transaction costs		38,888		9,665		31,427		(1,833)		_			_			78,147
Impairment losses		11,192		_		31,951		_		_			_			43,143
Provision for loan losses		37,475		_		4,201		(1,961)		_			_			39,715
General and administrative expenses		.,				,		(=,===)								30,120
Compensation expense	,	84,506		125,817		41,437		_		18,740	(1)		_			270,500
Other general and administrative expenses		38,238		33,386		16,658		_		4,901			_			93,183
Total general and administrative expenses		22,744		159,203	_	58,095				23,641				-		363,683
Depreciation and amortization		40,977		1,863		456,916		(148,301)				1	57,185	(11)		608,640
Total expenses		40,514	_	290,556	_	2,218,679	_	(424,540)	_	(211,089)			54,965	-		2,669,085
Other income (loss)		.0,01		250,550	_	2,210,070		(12.,5.0)		(211,000)			3 1,5 05	-		2,000,000
Unrealized gain (loss) on investments and other				(4,274)		(204,112)		_		3,745	(3)					(204,641)
Realized gain (loss) on investments and other		8,962		(4,2/4)		14,407		1,709		3,743	(-)					25,078
Gain on remeasurement of consolidated investment entities,						14,407		1,703								
net		41,486		_		_		_		_			_			41,486
Other gain (loss), net Income (loss) before equity in earnings (losses) of		(5,170)	_		_		_						_	_		(5,170)
unconsolidated ventures and income tax benefit (expense)	19	99,135		140,991		(363,465)		(11,522)		52,623		(1	08,335)			(90,573)
Equity in income of unconsolidated ventures	4	47,605		1,625		219,077		(77,851)		(4,443)	(4)		_			186,013
Income tax benefit (expense)		9,296		(21,869)		(14,325)		8,565		9,711	(5)			_		(8,622)
Income (loss) from continuing operations	2	56,036		120,747		(158,713)		(80,808)		57,891		(1	08,335)			86,818
Non-controlling interests—investments	;	86,123		_		(20,802)		7,896		(13,522)	(6)		8,075	(12)		67,770
Non-controlling interests—operating partnership		19,933		953		(3,206)		_		(23,600)	(7)		_			(5,920)
Preferred stock dividends		42,569				84,238								_		126,807
Net income (loss) from continuing operations attributable to common stockholders		07,411	\$	119,794	\$	(218,943)	\$	(88,704)	\$	95,013		\$ (1	16,410)	-	\$	(101,839)
Earnings (loss) per share:(G)																
Basic	\$	0.96	\$	0.61	\$	(1.25)									\$	(0.19)
Diluted	\$	0.96	\$	0.60	\$	(1.25)									\$	(0.19)
Weighted average number of shares:(G)			-		_											
Basic	1:	10,931		188,706		174,873										548,328
Diluted		10,931	_	193,119	_	176,345										549,781
Diraced		,	_	100,110	_	1, 0,040										3 /3,/01

⁽a) On April 2, 2015, Colony became an internally managed company by acquiring its manager, Colony Financial Manager, LLC, a wholly owned subsidiary of Colony Capital, LLC, as part of a combination transaction. Prior to such time, Colony was externally managed. The condensed consolidated pro forma statement of operations for the year ended December 31, 2015 does not adjust for activities prior to such combination transaction.

Refer to accompanying notes to unaudited pro forma condensed consolidated financial statements.

Notes to the Colony NorthStar Unaudited Pro Forma Condensed Consolidated Financial Statements

Note 1. Description of the Mergers

On June 2, 2016, a merger agreement was entered into among Colony, NSAM and NRF which provides for the mergers of Colony, NSAM and NRF with and into Colony NorthStar, as the publicly-traded company for the combined organizations. Upon the closing of the Mergers, Colony stockholders will own approximately 33.25%, NSAM stockholders will own approximately 32.85% and NRF stockholders will own approximately 33.90% of Colony NorthStar, on a fully diluted basis, excluding the effect of certain equity-based awards issuable in connection with the Mergers. Prior to the closing of the Mergers, NSAM expects the NSAM board or a duly authorized committee thereof to declare a special cash dividend in the amount of \$228 million to NSAM common stockholders.

Each share of Colony class A and class B common stock issued and outstanding immediately prior to the effective time of the Mergers will be canceled and converted into the right to receive 1.4663 shares of Colony NorthStar class A and class B common stock, respectively. Concurrently, Colony OP will issue additional partnership units to equal the number of operating partnership units outstanding on the day prior to the closing of the Mergers multiplied by the exchange ratio of 1.4663.

Each share of NSAM common stock and NSAM performance common stock issued and outstanding immediately prior to the effective time of the Mergers will be canceled and converted into the right to receive one share of Colony NorthStar class A common stock and Colony NorthStar performance common stock, respectively.

In connection with the merger of NRF LP with and into NRF and related reorganization transactions: (i) each NRF LTIP unit outstanding as of immediately prior to such effective time will be deemed to be fully vested and converted into one share of NRF common stock; (ii) each partnership unit in NRF LP designated as a partnership common unit outstanding as of immediately prior to such effective time (other than those held by NRF) will be converted into one share of NRF common stock; and (iii) each other interest in NRF LP held by NRF will no longer be outstanding and will automatically be canceled and will cease to exist.

In connection with the merger of New Parent Merger Sub with and into NRF: (i) each share of NRF common stock issued and outstanding as of immediately prior to the effective time of such merger automatically will be canceled and converted into the right to receive one share of New NRF Parent common stock; and (ii) each share of each series of NRF preferred stock will be converted into the right to receive one share of a corresponding series of New NRF Parent preferred stock.

At the effective time of the NRF merger, each share of New NRF Parent common stock will be converted into the right to receive 1.0996 shares of Colony NorthStar class A common stock.

Each share of each series of the Colony and NRF preferred stock (New NRF Parent preferred stock) issued and outstanding immediately prior to the effective time of the Mergers will be canceled and converted into the right to receive one share of a corresponding series of Colony NorthStar preferred stock, having substantially the same preferences, conversion and other rights, voting powers, restrictions, limitations as to dividends, qualifications and terms and conditions of redemption.

In connection with the strategic initiatives of NRF, NRF continues to execute a series of sales initiatives which include: (i) sales of certain real estate assets; (ii) sales of certain limited partnership interests in real estate private equity funds; and (iii) sales and/or accelerated repayments of certain commercial real estate, or CRE, debt and securities investments.

Under the merger agreement, NRF is required to use reasonable best efforts to continue certain agreed upon sales initiatives. In addition, in connection with the Mergers, NSAM and NRF plan to repay their respective outstanding corporate borrowings, which we refer to, collectively, as the NorthStar corporate borrowings (excluding NRF's \$280 million of trust preferred securities with maturities beginning in 2035), contemporaneous with the closing of the Mergers.

The completion of the Mergers is subject to, among other things, regulatory approvals and the receipt of Colony, NSAM and NRF stockholder approval. As of the date of this joint proxy statement/prospectus, the Mergers are expected to be completed in January 2017.

Note 2. Basis of Presentation

The unaudited pro forma condensed consolidated financial statements relating to the mergers of Colony, NSAM and NRF are prepared as of and for the nine months ended September 30, 2016 and for the year ended December 31, 2015, in accordance with Article 11 of Regulation S-X and, in the opinion of management, reflect all necessary adjustments. Accordingly, the historical financial information of Colony, NSAM and NRF has been adjusted to give pro forma effect to all significant events that are: (i) directly attributable to the Mergers and related transactions and NRF Sales Initiatives; (ii) factually supportable; and (iii) with respect to the unaudited pro forma condensed consolidated statements of operations, expected to have a continuing impact on the results of the combined company.

The unaudited pro forma condensed consolidated balance sheet as of September 30, 2016 is presented as if the Mergers and related transactions and NRF Sales Initiatives had become effective on September 30, 2016. The unaudited pro forma condensed consolidated statements of operations for the nine months ended September 30, 2016 and year ended December 31, 2015 are presented as if the Mergers and related transactions and NRF Sales Initiatives had become effective on January 1, 2015, the beginning of the earliest period presented.

Certain amounts in the historical consolidated financial statements of Colony, NSAM and NRF have been reclassified to conform to the presentation of the combined company in the unaudited pro forma condensed consolidated financial statements. Discontinued operations reported in NRF's historical consolidated statement of operations for the year ended December 31, 2015 have been excluded from the unaudited pro forma condensed consolidated statement of operations for the year ended December 31, 2015.

Significant transactions between NSAM and NRF during the nine months ended September 30, 2016 and year ended December 31, 2015 have been eliminated in the unaudited pro forma condensed consolidated financial statements as if NSAM and NRF were consolidated affiliates during these periods. There were no transactions between Colony and either NSAM or NRF during the periods presented.

The Mergers are accounted for under the acquisition method for business combinations as a reverse acquisition pursuant to Accounting Standards Codification Topic 805, Business Combinations. In the Mergers, NSAM is the legal acquirer while Colony is considered to be the accounting acquirer for financial reporting purposes. The consideration to be transferred establishes a new accounting basis for the assets acquired, liabilities assumed and non-controlling interests of NSAM and NRF, measured at their respective fair value as of the date the Mergers are consummated. Accordingly, the unaudited pro forma condensed consolidated financial statements include adjustments to record the assets, liabilities and non-controlling interests of NSAM and NRF at their estimated fair value, which are preliminary and subject to revision. To the extent fair value of the merger consideration exceeds fair value of net assets acquired, any such excess represents goodwill. Alternatively, if fair value of net assets acquired exceeds fair value of the merger consideration, the transaction could result in a bargain purchase gain that is recognized immediately in earnings and attributable to Colony NorthStar common stockholders. Adjustments to estimated fair value of identifiable assets and liabilities of NSAM and NRF, as well as adjustments to the merger consideration may change the determination and amount of goodwill and/or bargain purchase gain and may impact depreciation, amortization and accretion based on revised fair value of assets acquired and liabilities assumed. The actual value of the merger consideration will depend upon the market price of the Colony common stock and the NRF preferred stock at the time of closing of the Mergers. The final fair value and allocation of merger consideration will be determined upon completion of the Mergers, with the allocation to be finalized as soon as practicable within the measurement period of no later than one year following the closing date of the Mergers. The final acquisition accounting

The unaudited pro forma condensed consolidated financial statements are presented for illustrative purposes only and are not necessarily indicative of the financial position or results of operations of the Companies had the Mergers and related transactions and NRF Sales Initiatives been completed as of the beginning of the earliest period presented, nor indicative of future results of operations or future financial position of the combined company. The unaudited pro forma condensed consolidated financial statements do not reflect the costs of any integration activities or full benefits that may result from realization of future cost savings from operating efficiencies, revenue or other incremental synergies expected to result from the Mergers. Additionally, the unaudited pro forma condensed consolidated financial statements should be read in connection with the historical consolidated financial statements and notes thereto included in each of Colony's, NSAM's and NRF's respective Annual Reports on Form 10-K for the year ended December 31, 2015, as amended, and each of Colony's, NSAM's and NRF's respective Quarterly Reports on Form 10-Q for the nine months ended September 30, 2016. The unaudited pro forma condensed consolidated financial statements represent management's best estimate based on available information and may be revised as additional information becomes available and as additional analyses are performed.

Note 3. Pro Forma Merger Consideration

Pursuant to the merger agreement, NSAM common stock will first be converted into Colony NorthStar common stock through the Redomestication merger. NSAM will then acquire 100% of the common stock and preferred stock of Colony and NRF through the conversion of all such outstanding shares, based on pre-determined exchange ratios, into shares of Colony NorthStar.

As the Mergers are accounted for as a reverse acquisition, the fair value of the consideration transferred is measured based upon (i) the number of shares of common stock Colony, as the accounting acquirer, would theoretically have to issue to the stockholders of NSAM and NRF to achieve the same ratio of ownership in Colony NorthStar upon completion of the Mergers; and (ii) applying the Colony class A common stock price.

As a result, the implied shares of Colony common stock issued in consideration was computed based on the number of outstanding shares of NSAM and NRF common stock prior to the Mergers divided by the exchange ratios of 1.4663 and 1.3335, respectively.

Substantially all NSAM and NRF equity awards will vest upon consummation of the Mergers. As Colony NorthStar is obligated to issue Colony NorthStar common stock upon consummation of the Mergers and settlement of these equity awards relate to pre-combination services, these equity awards form part of the outstanding shares of NSAM and NRF common stock that are used to determine the merger consideration.

Any NSAM and NRF equity awards that do not vest by their terms upon consummation of the Mergers will be assumed by Colony NorthStar through conversion into comparable Colony NorthStar equity awards with substantially the same terms. Accordingly, these equity awards are not included in the outstanding shares of NSAM and NRF common stock that are used to determine the merger consideration and will be recognized as compensation expense in the post-combination period.

The pro forma merger consideration is estimated as follows (in thousands, except per share):

	 NSAM	 NRF	 Total
Outstanding shares of common stock prior to the Mergers	189,487	183,251	
Exchange ratio(i)	1.4663	1.3335	
Implied shares of Colony common stock issued in consideration	129,228	137,421	266,649
Price per share of Colony common stock(ii)	\$ 18.90	\$ 18.90	\$ 18.90
Fair value of implied shares of Colony common stock issued in consideration	\$ 2,442,409	\$ 2,597,257	\$ 5,039,666
Fair value of Colony NorthStar preferred stock to be issued(iii)	_	969,458	969,458
Total pro forma merger consideration	\$ 2,442,409	\$ 3,566,715	\$ 6,009,124

⁽i) Represents (a) the pre-determined exchange ratio of one Colony share for 1.4663 Colony NorthStar shares; and (b) the derived exchange ratio of one Colony share for 1.3335 NRF shares based on the pre-determined exchange ratio of one NRF share for 1.0996 Colony NorthStar shares.

⁽iii) Fair value of Colony NorthStar preferred stock to be issued is estimated based on the shares of NRF preferred stock outstanding as of September 30, 2016 multiplied by the closing price, which represents the clean price, of the respective series of NRF preferred stock as of November 9, 2016, as follows (in thousands, except per share):

	Number of Shares Outstanding	P	rice Per Share	F	air Value
NRF Preferred Stock					
Series A 8.75%	2,467	\$	24.50	\$	60,442
Series B 8.25%	13,999		23.99		335,836
Series C 8.875%	5,000		24.86		124,300
Series D 8.50%	8,000		25.06		200,480
Series E 8.75%	10,000		24.84		248,400
Fair value of Colony NorthStar preferred stock to be issued				\$	969,458

The following table presents a summary of the preliminary purchase price allocation of the pro forma merger consideration to the assets acquired, liabilities assumed and non-controlling interests of NSAM and NRF based on their respective estimated fair value as of September 30, 2016, after adjusting for NRF Sales Initiatives. Based on current estimates, the consideration to be transferred is in excess of the estimated fair value of assets and liabilities for both NSAM and NRF thereby resulting in goodwill that is recognized as a fair value adjustment in the unaudited pro forma condensed consolidated balance sheet (in thousands):

	NSAM	NRF	Total
Pro forma merger consideration ⁽ⁱ⁾	\$ 2,442,409	\$ 3,566,715	\$ 6,009,124
Preliminary allocation of pro forma merger consideration:			
Assets acquired	2,953,665	13,886,740	16,840,405
Liabilities assumed	(765,735)	(9,624,716)	(10,390,451)
Redeemable non-controlling interests	(75,149)	_	(75,149)
Non-controlling interests—investments	 	(748,582)	 (748,582)
Fair value of net assets acquired(ii)	 2,112,781	 3,513,442	 5,626,223
Preliminary pro forma goodwill	\$ 329,628	\$ 53,273	\$ 382,901

⁽ii) The pro forma merger consideration was determined based on the closing price of Colony common stock of \$18.90 on November 9, 2016.

- (i) Refer to the table Note 3 above, *Total pro forma merger consideration*.
- (ii) Refer to fair value of net assets acquired in Note 4.C, Adjustments to Unaudited Pro Forma Condensed Consolidated Balance Sheet—Fair Value Adjustments.

The pro forma merger consideration does not purport to represent the actual value of the merger consideration, which will be measured on the closing date of the Mergers at the then-current market price per share of Colony common stock and NRF preferred stock. The exchange ratios are not subject to adjustments to account for fluctuations in the share prices of Colony, NSAM or NRF common stock between now and the closing date of the Mergers. Therefore, the implied value of the merger consideration, and consequently, the resulting goodwill, may vary significantly due to movements in the Colony common stock and NRF preferred stock. For example, each 10% increase or decrease in the price of Colony common stock on the closing date of the Mergers from the price of Colony common stock assumed in these unaudited pro forma condensed consolidated financial statements would result in an increase or decrease in the estimated merger consideration of \$504.0 million and a corresponding increase or decrease in goodwill or result in a bargain purchase gain to the extent that merger consideration does not exceed the fair value of net assets acquired, measured separately for each acquired entity.

The estimated fair value and preliminary allocation of merger consideration are subject to revisions and will be finalized upon completion of the Mergers.

Note 4. Adjustments to the Unaudited Pro Forma Condensed Consolidated Balance Sheet

A. NRF Sales Initiatives

The following table presents a summary of the pro forma adjustments to the unaudited condensed consolidated balance sheet as of September 30, 2016 related to NRF Sales Initiatives. Such adjustments eliminate the net assets of those asset sales under contract but not yet sold as of November 4, 2016 as presented in NRF's Quarterly Report on Form 10-Q for the nine months ended September 30, 2016 (in thousands):

	anufactured Housing ⁽¹⁾	edical Office Buildings ⁽²⁾		Multifamily ⁽³⁾		vate Equity Portfolio ⁽⁴⁾	Healthcare Joint Venture		al NRF Sales Initiatives
Assets									
Cash and cash equivalents(6)	\$ 613,750	\$ 114,828	\$	44,670	\$	204,672	\$	340,000	\$ 1,317,920
Restricted cash	_	(945)		(4,041)		_		_	(4,986)
Assets of properties held for sale(7)	(1,592,357)	(807,731)		(133,000)		_		_	(2,533,088)
Other assets	 (276)	 (2,632)		(680)		(204,672)			(208,260)
Total assets	\$ (978,883)	\$ (696,480)	\$	(93,051)	\$	_	\$	340,000	\$ (1,428,414)
Liabilities									
Mortgage and other notes payable	\$ _	\$ (692,231)	\$	_	\$	_	\$	_	\$ (692,231)
Accounts payable and other liabilities	(1,473)	(8,161)		(3,045)		_		20,000	7,321
Liabilities of properties held for sale ⁽⁷⁾	 (1,281,438)	 (19,229)		(107,512)					(1,408,179)
Total liabilities	(1,282,911)	(719,621)		(110,557)		_		20,000	(2,093,089)
Equity									
Stockholders' equity	325,924	49,182		18,938		_		(10,000)	384,044
Non-controlling interests—investments	(21,896)	 (26,041)		(1,432)				330,000	280,631
Total equity	304,028	 23,141	_	17,506				320,000	664,675
Total liabilities and equity	\$ (978,883)	\$ (696,480)	\$	(93,051)	\$		\$	340,000	\$ (1,428,414)

⁽¹⁾ In May 2016, NRF entered into an agreement to sell its manufactured housing portfolio for \$2.0 billion with \$1.3 billion of related mortgage financing (recorded in liabilities of properties held for sale) expected to be assumed by the buyer as part of the transaction. NRF expects to receive \$614.8 million of net proceeds, including a \$50.0 million deposit made by the buyer. The transaction is expected to close in the first quarter 2017.

⁽²⁾ In September 2016, NRF entered into a definitive agreement to sell a portfolio of medical office buildings for \$837.9 million with \$692.2 million of related mortgage financing expected to be paid off as part of the transaction. NRF expects to receive \$114.8 million of net proceeds. The transaction is expected to close in the fourth quarter 2016.

⁽³⁾ To date through November 2016, the Company sold ten multifamily properties for \$307.0 million with \$210.0 million of mortgage financing assumed as part of the transaction. The above adjustment represents the remaining six properties sold subsequent to September 30, 2016. The Company received \$85.0 million of net proceeds from such sales.

⁽⁴⁾ In September 2016, NRF sold a portfolio of PE Investments for a gross sales price of \$317.6 million with \$44.7 million of deferred purchase price assumed as part of the transaction. NRF received \$33.9 million of net proceeds and will receive the remaining \$204.7 million of net proceeds in the fourth quarter 2016.

⁽⁵⁾ In November 2016, NRF entered into an agreement to sell a non-controlling interest in its healthcare portfolio for net proceeds of approximately \$340 million.

⁽⁶⁾ Proceeds from such sales are net of cash and cash equivalent balances as of September 30, 2016, as applicable.

(7) The following table presents the major classes of long-lived assets and liabilities classified as held for sale as of September 30, 2016 (in thousands):

			Assets			Liabilities										
Description	Ol	perating Real Estate ⁽ⁱ⁾		ntangible Assets ⁽ⁱⁱ⁾		Other Assets ⁽ⁱⁱⁱ⁾		Total	Mortgage and Other Notes Payable			Intangible Liabilities	L	Other iabilities ^(iv)		Total
Manufactured housing communities	\$	1,441,656	\$	23,983	\$	126,718	\$	1,592,357	\$	1,255,454	\$	_	\$	25,984	\$	1,281,438
Medical office buildings		742,485		63,818		1,428		807,731		_		19,229		_		19,229
Multifamily		133,000		_		_		133,000		107,512				_		107,512
Total	\$	2,317,141	\$	87,801	\$	128,146	\$	2,533,088	\$	1,362,966	\$	19,229	\$	25,984	\$	1,408,179

(i) Operating real estate is comprised of the following (in thousands):

Operating real estate - held-for-sale	 anufactured Housing ommunities	 dical Office Buildings	М	ultifamily	Total
Land and improvements	\$ 1,453,007	\$ 65,097	\$	15,333	\$ 1,533,437
Buildings and improvements	141,593	723,719		128,517	993,829
Furniture, fixtures and equipment	 7,814	 29		329	8,172
Subtotal	1,602,414	788,845		144,179	2,535,438
Less: accumulated depreciation	 (160,758)	 (46,360)		(11,179)	(218,297)
Total	\$ 1,441,656	\$ 742,485	\$	133,000	\$ 2,317,141

- (ii) Represents the carrying value of in-place and above-market leases, net of amortization.
- (iii) Includes cash and cash equivalents, restricted cash, accounts, notes and other receivables.
- (iv) Includes interest, taxes and accounts payable.

B. Merger Adjustments

- (1) Includes an adjustment related to the payment of the NSAM special dividend of \$228.0 million. Refer to footnote 12.
- (2) Includes an adjustment for the expected disposition of NSAM's investment in Island Hospitality Management Inc., which, for the purpose of this pro forma adjustment, is assumed to be at its carrying value of \$39.4 million. NSAM's investment in Island Hospitality Management Inc. is expected to be sold in connection with the Mergers.
- 3) The following table presents a summary of merger-related adjustments in connection with the pay down of NorthStar corporate borrowings (in thousands):

<u>Adjustments to cash and cash equivalents related to the pay down of the NorthStar corporate borrowings:</u>	NSAM	NRF	Total
Principal pay down of NorthStar corporate borrowings (refer to footnote 8)	\$ (497,500)	\$ (425,000)	\$ (922,500)
Prepaid interest (refer to footnote 7)	_	4,328	4,328
Interest payable (refer to footnote 9)	(4,262)		(4,262)
Total	\$ (501,762)	\$ (420,672)	\$ (922,434)

- (4) Represents the elimination of the carrying value of NSAM's ownership of 2.7 million shares of NRF common stock. Refer to footnotes 10 and 11.
- (5) Represents the elimination of the management fee receivable (payable) between NSAM and NRF, respectively.
- (6) Represents the elimination of the fair value of the management contract value between NSAM and NRF, which will cease to exist upon completion of the Mergers. Refer to Note 4.C, Fair Value Adjustments for further disclosure.
- (7) The following table presents a summary of merger-related adjustments related to other assets (in thousands):

Adjustments related to other assets:	 NRF
Deferred financing costs	\$ (725) ⁽ⁱ⁾
Prepaid interest	 (4,328) (ii)
Total	\$ (5,053)

- (i) Represents an adjustment to eliminate deferred financing costs related to NRF's corporate revolving credit facility, which is expected to be paid down and terminated in connection with the Mergers. Refer to footnotes 3 and 12 for additional information.
- (ii) Represents an adjustment to eliminate prepaid interest related to NRF's term borrowing, which is expected to be paid down and terminated in connection with the Mergers. Refer to footnote 3 for additional information.

(8) The following table presents a summary of merger-related adjustments related to credit facilities and term borrowings (in thousands):

Adjustments related to credit facilities and term borrowings:	NSAM	NRF	Total
Principal pay down of NorthStar corporate borrowings (refer to footnote 3) $^{(i)}$	\$ (497,500)	\$ (425,000)	\$ (922,500)
Elimination of deferred financing costs	28,822	4,591	33,413
Total	\$ (468,678)	\$ (420,409)	\$ (889,087)

(i) Proceeds from NRF Sales Initiatives are expected to be used for the pay down of NorthStar corporate borrowings. However, to the extent NRF Sales Initiatives are not completed by the closing of the Mergers, Colony has obtained bridge financing for up to \$400 million in addition to the \$850 million available under its credit facility that can be used to pay off such NorthStar corporate borrowings, if necessary. Colony NorthStar does not expect to draw on the bridge financing and therefore, the pro forma adjustments do not reflect any additional borrowing on such credit facility or the bridge financing.

(9) The following table presents a summary of merger-related adjustments related to accounts payable and other liabilities (in thousands):

Adjustments related to accounts payable and other liabilities:	C	Colony	NSAM	 NRF	 Total
Merger-related transaction and other costs(i)	\$	29,955	\$ 37,743	\$ 59,155	\$ 126,853
NSAM executive compensation accrual(ii)		_	(22,766)	(2,039)	(24,805)
Interest payable related to NSAM's corporate borrowing(iii)		<u> </u>	 (4,262)	<u> </u>	(4,262)
Total	\$	29,955	\$ 10,715	\$ 57,116	\$ 97,786

(i) Represents non-recurring transaction and other costs incurred in connection with the Mergers, consisting primarily of advisory, legal, accounting, tax and other professional services and are factually supportable as such amounts are based on reliable, documented evidence such as invoices for costs incurred to date and estimates from third-parties for additional costs expected to be incurred until the closing of the Mergers. Such costs are reflected as a reduction to retained earnings and not included in the unaudited pro forma condensed consolidated statements of operations. Refer to footnote 12.

(ii) Represents an adjustment to eliminate compensation payable related to arrangements entered into with the NSAM executive officers in connection with the Mergers. Refer to footnote

(iii) Represents an adjustment to eliminate interest payable related to NSAM's corporate borrowings. Refer to footnote 3 for additional information.

(10) The following table presents a summary of the merger-related adjustment to common stock par value as of September 30, 2016 (in thousands, except for exchange ratios and par value per share):

Adjustments to common stock at par:	(Colony	NSAM	NRF	 ro Forma Colony orthStar
Outstanding shares of common stock as of September 30, 2016(i)	-	113,928	188,983	180,730	
Equity awards to vest upon the Mergers and converted into Colony NorthStar common stock, net of shares withheld for $\tan^{(ii)}$		_	504	665	
NRF LTIP units converted to common stock(iii)			 <u> </u>	 1,856	
Outstanding shares of common stock prior to the Mergers		113,928	189,487	183,251	
Exchange ratio		1.4663	 1.0000	 1.0996	
Shares of Colony NorthStar common stock—pro forma basis		167,053	189,487	201,503	
Shares of NRF common stock owned by NSAM(iv)		NA	NA	(2,969)	
Shares of Colony NorthStar common stock—pro forma basis (as adjusted)(v)		167,053	 189,487	 198,534	555,074
Par value per share	\$	0.01	\$ 0.01	\$ 0.01	\$ 0.01
Common stock at par of Colony NorthStar—pro forma basis	\$	1,671	\$ 1,895	\$ 1,985	5,551
Common stock at par as of September 30, 2016		(1,139)	(1,890)	(1,807)	(4,836)
Pro forma adjustment to common stock at par	\$	532	\$ 5	\$ 178	\$ 715

(i) Includes restricted common stock issued as equity-based awards.

(ii) Represents 9.4 million equity-based shares of NSAM that will convert to Colony NorthStar class A common stock upon completion of the Mergers, net of 4.1 million shares forfeited by NSAM executives and 4.8 million shares estimated to be retired upon vesting for NSAM executive and employee tax withholding. Represents 3.2 million equity-based shares of NRF common stock that will convert to Colony NorthStar class A common stock upon completion of the Mergers, net of 1.1 million shares forfeited by NSAM executives and 1.4 million shares estimated to be retired upon vesting for NSAM executive and employee tax withholding. Refer to the section entitled "The Mergers-Interests of NSAM's Directors and Executive Officers in the Mergers" for further information regarding shares forfeited by NSAM executives. Shares withheld for taxes include amounts related to restricted common stock included in outstanding common stock.

(iii) In connection with the Mergers, NRF LP will be merged into NRF resulting in the conversion to NRF common stock of existing LTIP units in NRF LP.

(iv) Represents the 2.7 million shares of NRF common stock owned by NSAM after giving effect to the exchange ratio.

(v) Includes shares of both class A and class B pro forma Colony NorthStar common stock.

(11) The following table presents a summary of the merger-related adjustments to additional paid-in capital as of September 30, 2016 (in thousands):

Adjustments to additional paid-in capital:	C	olony	 NSAM		NRF	Total
Adjustment to common stock par value (refer to footnote 10)	\$	(532)	\$ (5)	\$	(178)	\$ (715)
Elimination of retained earnings (accumulated deficit)		_	(38,554)	(2	,891,153)	(2,929,707)
Elimination of accumulated other comprehensive income (loss)		_	(210)		(63,709)	(63,919)
Adjustment to non-controlling interests in operating partnership		_	5,436 (iv)		31,798 ⁽ⁱ⁾	37,234
Elimination of carrying value of NRF common stock owned by NSAM		_	_		(35,741) (ii)	(35,741)
Acceleration of equity-based awards vested upon the Mergers(iii)		_	69,210		_	69,210
Elimination of retained earnings from NRF Sale Initiatives		_	_		384,044	 384,044
Total merger-related adjustments to additional paid-in capital						\$ (2,539,594)

⁽i) In connection with the Mergers, NRF LP will be merged into NRF.

⁽iv) The following table presents a summary of the adjustment to Colony NorthStar's non-controlling interests in the operating partnership as of September 30, 2016 (in thousands, except for exchange ratio):

Pro forma Colony NorthStar non-controlling interest in the operating partnership:(1)	Colony	NSAM	Total Colony NorthStar
OP units owned by non-controlling interests as of September 30, 2016	20,787	1,790	
Exchange ratio	1.4663	1.0000	
Non-controlling interests' ownership of Colony NorthStar OP units—pro forma basis	30,480	1,790	32,270
Shares of Colony NorthStar common stock—pro forma basis			555,074
Pro forma non-controlling OP unit ownership % in Colony NorthStar			5.5%
Adjustment to non-controlling interests in operating partnership			\$ 5,436

⁽i) In connection with the Mergers, NRF LP will be merged into NRF.

(12) The following table presents a summary of merger-related adjustments to retained earnings (accumulated deficit) as of September 30, 2016 (in thousands):

Adjustments to retained earnings (accumulated deficit):	Colony	NSAM	NRF	Tot	tal
Elimination of retained earnings (accumulated deficit) as of September 30, 2016	\$ —	\$ 38,554	\$ 2,891,153	\$ 2,92	29,707
NSAM special dividend (refer to footnote 1)	_	(228,000)	_	(22	28,000)
Merger-related transaction costs ⁽ⁱ⁾	(29,955)	(37,743)	(59,155)	(12	26,853)
NSAM executive compensation accrual ⁽ⁱⁱ⁾	_	22,766	2,039	2	24,805
Acceleration of equity-based awards vested upon the Mergers ⁽ⁱⁱⁱ⁾	_	(69,210)	_	(6	59,210)
NorthStar corporate borrowings deferred financing costs ^(iv)	_	(28,822)	(5,316)	(3	34,138)
Elimination of retained earnings from NRF Sale Initiatives	_	_	(384,044)	(38	34,044)
Total merger-related adjustments to retained earnings (accumulated deficit)				\$ 2,11	12,267

⁽i) Represents non-recurring transaction costs directly attributable to the Mergers, of which \$126.9 million is a pro forma adjustment to accounts payable and other liabilities (refer to footnote 9) in the historical financial statements as of September 30, 2016.

⁽ii) Represents the carrying value of 2.7 million shares of NRF common stock owned by NSAM (refer to footnote 4).

⁽iii) Represents the acceleration of amortization of equity-based compensation related to substantially all outstanding NSAM equity awards that will vest in accordance with their terms upon the closing of the Mergers. Colony and NRF equity awards that vest in connection with the Mergers are not included as adjustments as such events occur prior to the Mergers.

⁽ii) Represents an adjustment to eliminate compensation payable related to arrangements entered into with the NSAM executive officers in connection with the Mergers.

⁽iii) Represents the acceleration of amortization of equity-based compensation related to substantially all outstanding NSAM equity awards that will vest in accordance with their terms upon the closing of the Mergers. NRF equity awards that vest in connection with the Mergers are not included as an adjustment as such event occurs prior to the Mergers.

⁽iv) Represents an adjustment to eliminate deferred financing costs of \$0.7 million related to NRF's corporate revolving credit facility and \$33.4 million related to NSAM and NRF's respective term borrowings. The NorthStar corporate borrowings are expected to be paid off and terminated in connection with the Mergers.

C. Fair Value Adjustments

The fair value adjustments reflected in the unaudited pro forma condensed consolidated balance sheet represent the differences between fair value amounts based on a preliminary purchase price allocation of the assets acquired and liabilities assumed of NSAM and NRF and the corresponding historical balances of NSAM and NRF, as adjusted (in thousands):

	NSAM				NRF							
	Fair	Value ⁽¹⁾		Adjusted Historical	_	F	Fair Value ⁽¹⁾⁽²⁾		Adjusted Historical ⁽³⁾		Fair Value Adjustment	
Assets					_		<u> </u>					
Cash and cash equivalents	\$	125,037	\$	125,037	(15)	\$	2,043,280	\$	2,043,280	\$	_	
Restricted cash		26,599		26,599			175,082		175,082		_	
Operating real estate, net		_		_			9,082,788		7,371,996		1,710,792	(4)
Real estate debt investments, net		_		_			349,083		348,539		544	
Investments in private equity funds		_		_			484,876		484,876		_	(5)
Investments in unconsolidated ventures		57,664		57,664	(15)		177,994		161,744		16,250	(5)
Real estate securities, available for sale		_		_			526,966		526,966		_	(5)
Securities, at fair value		2,697		2,697	(16)		_		_		_	
Due from affiliates		109,753		109,753			1,888		1,888		_	
Goodwill		329,628		243,328			53,273		44,767		94,806	(6)
Intangible assets, net		2,511,020		203,728			645,249		298,950		2,653,591	(4)
Assets of properties held for sale		_		_			109,096		120,871		(11,775)	(3)
Other assets		120,895		41,784			290,438		357,516		12,033	(7)
Total assets	\$	3,283,293	\$	810,590		\$	13,940,013	\$	11,936,475	\$	4,476,241	Ī
Liabilities					=							=
Mortgage and other notes payable	\$	_	\$	_		\$	6,166,576	\$	6,229,796	\$	(63,220)	(8)
Credit facilities and term borrowings		497,500		468,679			420,409		420,409		28,821	(9)
Convertible senior notes		_		_			29,441		27,356		2,085	(8)
Securitization bonds payable		_		_			257,877		257,877		_	(5)
Junior subordinated notes		_		_			191,175		191,175		_	(5)
Accounts payable and other liabilities		268,235		91,155			212,463		212,463		177,080	(10
Due to related parties							46,939		46,939		_	
Intangible liabilities, net		_		_			1,944,091		113,967		1,830,124	(4)
Liabilities of properties held for sale		_		_			53,429		94,480		(41,051)	
Derivative liabilities, at fair value		_		_			302,316		302,316		_	
Total liabilities		765,735		559,834	_		9,624,716		7,896,778		1,933,839	Ī
Commitments and contingencies	-	·	_	·	_	-	· · · · · · · · · · · · · · · · · · ·					-
Redeemable non-controlling interests		75,149		75,149			_		_		_	
Equity		-, -		-, -								
Performance common stock		52		52			_		_		_	
Preferred stock		_		_			969,458		939,118		30,340	(12)
Common stock		1,890		1,890			1,807		1,807		_	
Additional paid-in capital		2,477,233		210,431	(15)(16)		2,243,317		2,224,947		2,285,172	(13)
Accumulated other comprehensive income (loss)		(210)		(210)			(63,709)		(63,709)			
Retained earnings (accumulated deficit)		(38,554)		(38,554)			384,044		384,044		_	
Total stockholders' equity		2,440,411		173,609	_		3,534,917		3,486,207		2,315,512	
Non-controlling interests—investments				_	_		748,582		521,692		226,890	(14
Non-controlling interests—operating partnership		1,998		1,998			31,798		31,798			(14)
Total equity		2,442,409	-	175,607	_		4,315,297		4,039,697		2,542,402	_
Total liabilities, redeemable non-controlling interests and equity	\$	3,283,293	\$	810,590	_	\$	13,940,013	\$	11,936,475	\$	4,476,241	ĺ

⁽¹⁾ Fair value reflected in the unaudited pro forma condensed consolidated balance sheet was estimated as follows:

⁽i) Real estate and related intangibles—based on a discounted cash flow analysis or direct capitalization analysis, and for real estate held for sale, contracted sale price or a sales comparison approach, adjusted for estimated selling costs. The fair value is allocated to tangible assets such as land, building, tenant and land improvements and identified intangibles, such as above- and below-market leases, above- and below market ground lease obligations, in-place lease value and goodwill.

- (ii) Real estate debt investments—determined by comparing the current yield to the estimated yield for newly originated loans with similar credit risk or the market yield at which a third party might expect to purchase such investment or based on discounted cash flow projections of principal and interest expected to be collected, which include consideration of borrower or sponsor credit, as well as operating results of the underlying collateral. For certain real estate debt investments considered to be impaired, their carrying value approximates fair value.
- (iii) Private equity funds and investments in unconsolidated ventures—based on the timing and amount of expected future cash flow for income and realization events for underlying assets.
- (iv) Real estate securities—based on quotations from brokers or financial institutions that act as underwriters of the securities, third-party pricing service or discounted cash flow depending on the type of securities.
- (v) Management agreements and related intangible assets—comprised of NSAM's management contracts, customer relationships and trade name. The fair value of management contracts and customer relationships represent the discounted excess earnings attributable to the future management fee income from in-place management contracts and to the potential fee income from repeat customers through future sponsored funds, respectively. The fair value of trade name is estimated based on a discounted royalty rate.
- (vi) Mortgage and other notes payable—estimated by discounting expected future cash outlays at interest rates currently available for instruments with similar terms and remaining maturities.
- (vii) Convertible senior notes—based on quoted market prices or recent transactions.
- (viii) Securitization bonds payable and junior subordinated notes—based on quotations from brokers or financial institutions that act as underwriters of the securitized bonds or subordinated notes.
- (2) Fair value excludes assets and liabilities associated with the NRF Sales Initiatives (refer to footnote 3).
- (3) The following table presents the assets and liabilities of NRF as of September 30, 2016, adjusted to reflect the impact of the NRF Sales Initiatives (in thousands):

	NRF								
		Historical		NRF Sales Initiatives ⁽ⁱ⁾		Adjusted Historical			
Assets									
Cash and cash equivalents	\$	725,360	\$	1,317,920	\$	2,043,280			
Restricted cash		180,068		(4,986)		175,082			
Operating real estate, net		7,371,996		_		7,371,996			
Real estate debt investments, net		348,539		_		348,539			
Investments in private equity funds, at fair value		484,876		_		484,876			
Investments in unconsolidated ventures		161,744		_		161,744			
Real estate securities, available for sale		526,966		_		526,966			
Due from affiliates		1,888		_		1,888			
Goodwill		44,767		_		44,767			
Intangible assets, net		298,950		_		298,950			
Assets of properties held for sale		2,653,959		(2,533,088)		120,871			
Other assets		565,776		(208,260)		357,516			
Total assets	\$	13,364,889	\$	(1,428,414)	\$	11,936,475			
Liabilities									
Mortgage and other notes payable	\$	6,922,027	\$	(692,231)	\$	6,229,796			
Credit facilities and term borrowings		420,409		_		420,409			
Convertible senior notes		27,356		_		27,356			
Securitization bonds payable		257,877		_		257,877			
Junior subordinated notes		191,175		_		191,175			
Accounts payable and other liabilities		205,142		7,321		212,463			
Due to related parties		46,939		_		46,939			
Intangible liabilities, net		113,967		_		113,967			
Liabilities of properties held for sale		1,502,659		(1,408,179)		94,480			
Derivative liabilities, at fair value		302,316				302,316			
Total liabilities	\$	9,989,867	\$	(2,093,089)	\$	7,896,778			

 $⁽i) \quad \ \ \text{Refer to A, Adjustments to the Unaudited Pro Forma Condensed Consolidated Balance Sheet-NRF Sales Initiatives}.$

(4) The acquired assets and liabilities assumed for real estate generally include, but are not limited to land, buildings and improvements, identified tangible and intangible assets and liabilities associated with in-place leases, above- and below-market leases and goodwill, if any. The following table presents a summary of the major classes of intangible assets acquired and liabilities assumed as part of the Mergers (in thousands):

	NSAM			NRF				Fair Value		
Intangible assets	I	Fair Value		Iistorical]	Fair Value		Historical	Α	Adjustment
Management agreements and related intangibles(i)	\$	2,511,020	\$	203,728	\$	_	\$	_	\$	2,307,292
In-place lease values		_		_		151,921		104,668		47,253
Above-market lease values		_		_		400,256		157,003		243,253
Below-market ground lease obligations		_		_		34,082		_		34,082
Other real estate intangible assets						58,990		37,279		21,711
Total	\$	2,511,020	\$	203,728	\$	645,249	\$	298,950	\$	2,653,591
Intangible liabilities										
NRF management agreement	\$	_	\$	_	\$	1,800,000	\$	_	\$	1,800,000
Below-market lease values		_		_		133,222		111,828		21,394
Other real estate intangible liabilities		_				10,869		2,139		8,730
Total	\$		\$		\$	1,944,091	\$	113,967	\$	1,830,124

NSAM's management agreements and related intangibles are summarized as follows (in thousands):

	NSAM						
	I	Historical					
NSAM:							
NSAM Retail Companies management agreements(a)	\$	333,100	\$	_			
NorthStar Europe management agreement(a)		109,600		_			
Trade name		59,000		_			
NRF management agreement(a)		1,800,000		_			
<u>Townsend:</u>							
Customer relationships		185,580		180,862			
Performance fees		5,710		5,289			
Trade name		17,820		17,424			
Proprietary technology		210		153			
Total	\$	2,511,020	\$	203,728			

- (a) The preliminary value was estimated using a discounted cash flow analysis, comparing the existing NSAM management agreements with a range of observable inputs for similar contracts including discount rates ranging between 7.0% and 9.0%. The NRF management agreement represents the off market fair value of such agreement. The NRF management agreement will cease to exist upon completion of the Mergers.
- (b) The estimated useful lives of NSAM's management agreement and related intangibles range from three years to 30 years.
- (5) NRF has historically elected the fair value option for its investments in private equity funds, certain investments in unconsolidated ventures, real estate securities, securitization bonds payable and junior subordinated notes, where carrying value represents fair value. The adjustment reflects the fair value of certain investments in unconsolidated ventures carried at historical cost.
- (6) Represents elimination of historical goodwill of NSAM and certain NRF properties and an adjustment for goodwill based on the preliminary purchase price allocation (refer to Note 3. *Pro Forma Merger Consideration*).
- (7) Straight-lining of rent pursuant to the underlying leases associated with the real estate acquired in connection with the Mergers will commence at the effective date of the Mergers; therefore the amount of unbilled rent receivable on the balance sheet as of September 30, 2016 has been eliminated.
- 8) Represents fair value adjustments, including the elimination of deferred financing costs.
- (9) The carrying value of NRF credit facilities and term borrowings approximate fair value. The adjustment represents elimination of NSAM historical deferred financing costs related to its credit facility.
- (10) Represents the estimated deferred tax effect of the pro forma adjustments related to NSAM management agreements and investment in The Townsend Group using an estimated 40% effective income tax rate.
- (11) Represents an adjustment to eliminate a mortgage note payable related to certain properties for which NRF is currently in negotiations with the lender to foreclose. In April 2016, NRF gave three properties back to the lender and is expected to give the remaining property back to the lender in the fourth quarter 2016 upon which the mortgage note will be extinguished.
- (12) Represents an adjustment to reflect the fair value of Colony NorthStar preferred stock to be issued as merger consideration, as discussed in Note 3, Pro Forma Merger Consideration.
- (13) Adjustment to additional paid-in capital represents the remaining net asset value of NSAM and NRF after adjustments to retained earnings (accumulated deficit) and non-controlling interests (refer to footnote 14).
- (14) Fair value of non-controlling interests are derived as their proportionate share of the fair value of net assets attributable to them, such fair value is determined on same basis as described above.

- (15) NSAM historical includes an adjustment to eliminate NSAM's interest in Island Hospitality Management Inc. Refer to footnote 2 in Note 4.B, Merger Adjustments, for additional information.
- (16) NSAM historical includes an elimination of the carrying value of NSAM's ownership of 2.7 million shares of NRF common stock. Refer to footnote 4 in Note 4.B, Merger Adjustments for additional information

Note 5. Adjustments to the Unaudited Pro Forma Condensed Consolidated Statements of Operations

D. NRF Sales Initiatives

The following tables present summarized pro forma adjustments for the nine months ended September 30, 2016 and year ended December 31, 2015 related to NRF Sales Initiatives. Such adjustments eliminate any activity related to assets sold or under contract to sell through November 4, 2016, as disclosed in NRF's Quarterly Report on Form 10-Q for the nine months ended September 30, 2016, however, it excludes the impact of potential reinvestment of proceeds from the NRF Sale Initiatives (in thousands):

	Nine Months Ended September 30, 2016											
	Manufactured Housing	Multifamily	Healthcare Portfolio ⁽¹⁾	Industrial Portfolio	Private Equity Portfolio	CRE Debt Investments	CRE Securities	Total NRF Sales Initiatives				
Revenues												
Rental and escalation income	\$ (147,680)	\$ (21,189)	\$ (69,239)	\$ (27,228)	\$ —	\$ —	\$ —	\$ (265,336)				
Interest income	(4,341)	_	(3)	(3)	_	(8,405)	(772)	(13,524)				
Other income	(4,018)	(1,273)	(930)					(6,221)				
Total revenues	(156,039)	(22,462)	(70,172)	(27,231)	_	(8,405)	(772)	(285,081)				
Expenses												
Interest expense	(42,781)	(5,481)	(22,729)	(4,897)	_	(393)	_	(76,281)				
Property operating expenses	(58,034)	(9,660)	(23,735)	(3,849)	_	_	_	(95,278)				
Other expense—investment and servicing expenses	(389)	(113)	(95)	(785)	_	(42)	_	(1,424)				
Transaction costs	(186)	_	_	_	_	_	_	(186)				
Provision for loan losses	(245)	_	_	_	_	(2,806)	_	(3,051)				
Depreciation and amortization			(27,889)	(5,217)	_			(33,106)				
Total expenses	(101,635)	(15,254)	(74,448)	(14,748)	_	(3,241)		(209,326)				
Other income (loss)												
Unrealized gain (loss) on investments and other	_	_	_	_	_	_	_	_				
Realized gain (loss) on investments and other	3,626	(21,800)	(16,696)	(13,235)	9,889	1,302	_	(36,914)				
Equity in earnings of unconsolidated joint ventures	_	_	_	_	(10,799)	_	_	(10,799)				
Income tax benefit (expense)	(542)	_	(21)	_	3,707	18	_	3,162				
Income (loss) from continuing operations	(51,320)	(29,008)	(12,441)	(25,718)	2,797	(3,844)	(772)	— (120,306)				
Non-controlling interests—investments	_	(2,179)	(1,423)		_		_	(3,602)				
Net income (loss) from continuing operations attributable to common stockholders	\$ (51,320)	\$ (26,829)	\$ (11,018)	\$ (25,718)	\$ 2,797	\$ (3,844)	\$ (772)	\$ (116,704)				

Year Ended December 31, 2015

	Manufactured Housing	Multifamily	Healthcare Portfolio ⁽¹⁾	Industrial Portfolio	Private Equity Portfolios	CRE Debt Investments	CRE Securities	Total NRF Sales Initiatives
Revenues								
Rental and escalation income	\$ (174,559)	\$ (32,201)	\$ (135,371)	\$ (39,850)	\$ —	\$ —	\$ —	\$ (381,981)
Interest income	(6,251)	_	(1)	(1)	_	(34,023)	(5,175)	(45,451)
Other income	(6,209)	(1,951)	(2,179)		_			(10,339)
Total revenues	(187,019)	(34,152)	(137,551)	(39,851)		(34,023)	(5,175)	(437,771)
Expenses								
Interest expense	(51,819)	(8,560)	(39,774)	(7,319)	_	(2,027)	_	(109,499)
Property operating expenses	(72,808)	(15,509)	(57,033)	(5,604)	_	_	_	(150,954)
Other expense—investment and servicing expenses	(2,078)	(87)	(8,553)	(1,116)	_	(158)	_	(11,992)
Transaction costs	(1,779)	_	_	(54)	_	_	_	(1,833)
Provision for loan losses	(766)	_	_		_	(1,195)	_	(1,961)
Depreciation and amortization	(68,331)	(6,928)	(62,641)	(10,401)	_			(148,301)
Total expenses	(197,581)	(31,084)	(168,001)	(24,494)	_	(3,380)	_	(424,540)
Other income (loss)								
Realized gain (loss) on investments and other	1,709	_	_	_	_	_	_	1,709
Income (loss) before equity in earnings (losses) of unconsolidated ventures and income tax benefit (expense)	12,271	(3,068)	30,450	(15,357)	_	(30,643)	(5,175)	(11,522)
Equity in earnings of unconsolidated joint ventures	_	_	_		(77,851)	_	_	(77,851)
Income tax benefit (expense)	435	_	55		7,911	164	_	8,565
Income (loss) from continuing operations	12,706	(3,068)	30,505	(15,357)	(69,940)	(30,479)	(5,175)	(80,808)
Non-controlling interests—investments		(165)	8,061					7,896
Net income (loss) from continuing operations attributable to common stockholders	\$ 12,706	\$ (2,903)	\$ 22,444	\$ (15,357)	\$ (69,940)	\$ (30,479)	\$ (5,175)	\$ (88,704)

⁽¹⁾ Includes a portfolio of medical office buildings and a senior housing portfolio.

E. Merger Adjustments

As a result of the Mergers, Colony NorthStar expects estimated annualized synergies of \$115 million, consisting of \$80 million of cash savings and \$35 million of equity-based compensation savings. The merger adjustments to the unaudited pro forma condensed consolidated statements of operations exclude integration activities or full savings that may result from realization of such future cost savings from operating efficiencies, revenue or other incremental synergies expected to result from the Mergers other than the executive compensation adjustments noted in footnotes 1(v) and 1(vii) below. The nine months ended September 30, 2016 and year ended December 31, 2015 pro forma condensed consolidated statements of operations exclude \$48 million and \$34 million, respectively, of expected cash savings as such amounts are not currently supportable through contracts or other agreements in place.

(1) The following table presents a summary of pro forma adjustments related to the Mergers (in thousands):

	Nine Months Ended September 30, 2016								Year Ended December 31, 2015								
	Colony			NSAM			NRF		Total	(Colony		NSAM		NRF		Total
Pro Forma Adjustments to Revenues																	
Fee income	\$ 		\$	(139,955)	(i)	\$	_	\$	(139,955)	\$		\$	(198,695) (i)	\$		\$	(198,695)
Other income																	_
Dividend income	\$ _		\$	(3,256)	(ii)	\$	_	\$	(3,256)	\$	_	\$	_	\$	_	\$	_
Loan origination fee	_			_			(843) (iv)		(843)		_		_		(2,995) (iv)		(2,995)
Reimbursement between NSAM and managed companies				24,229	(iii)		_		24,229		_		39,479 (iii)		_		39,479
Total other income	\$ 		\$	20,973		\$	(843)	\$	20,130	\$		\$	39,479	\$	(2,995)	\$	36,484
Pro Forma Adjustments to Expenses																	_
Management fee	\$ 		\$			\$	(139,955) (i)	\$	(139,955)	\$		\$		\$	(198,695) (i)	\$	(198,695)
Transaction costs	\$ (11,345)	(ix)	\$	(24,431)	(ix)	\$	(10,338) (ix)	\$	(46,114)	\$	_	\$	_	\$	_	\$	
Compensation costs																	_
Reimbursement between NSAM and managed companies	\$ _		\$	19,383	(iii)	\$	_	\$	19,383	\$	_	\$	31,583 (iii)	\$	_	\$	31,583
Cash compensation	_			(11,595)	(v)		(186) (v)		(11,781)		_		(45,353) (v)		(499) (v)		(45,852)
Equity-based compensation expense	(815)	(vi)		(11,746)	(v)		(4,256) (v)		(16,817)		(1,726) (vi)		55,127 (v)(vii)		(20,392) (v)		33,009
Total compensation costs(viii)	\$ (815)		\$	(3,958)		\$	(4,442)	\$	(9,215)	\$	(1,726)	\$	41,357	\$	(20,891)	\$	18,740
Other general and administrative expenses							_				_		_				_
Loan origination fee	\$ _		\$	(843)	(iv)	\$	_	\$	(843)	\$	_	\$	(2,995) (iv)	\$	_	\$	(2,995)
Reimbursement between NSAM and managed companies	_			4,846	(iii)		_		4,846		_		7,896 (iii)		_		7,896
Total other general and administrative expenses	\$ _		\$	4,003		\$	_	\$	4,003	\$	_	\$	4,901	\$	_	\$	4,901

(i) Represents elimination of the management fee income (expense) between NSAM and NRF, respectively.

iii) Represents reclassification of reimbursable expenses incurred on behalf of NSAM's managed companies (excluding amounts allocated to NRF which do not result in an adjustment).

(iv) Represents elimination of loan origination fees from NSAM to NRF.

(v) Includes an adjustment to eliminate cash and equity-based compensation related to arrangements entered into by the NSAM executive officers in connection with the Mergers. The nine months ended September 30, 2016 does not adjust to eliminate the historical cash and equity-based compensation related to Messrs. Hamamoto and Gilbert who will remain at Colony NorthStar, and therefore, their historical cash and equity-based compensation will have a continuing impact on Colony NorthStar. Messrs. Hamamoto and Gilbert are expected to enter into new employment arrangements prior to the Mergers. Their historical cash and equity-based compensation may not be indicative of any future cash and equity-based compensation.

vi) Represents an adjustment to recognize equity-based compensation expense on outstanding Colony equity awards at their remeasured fair value.

- (vii) Includes the amortization of \$97.3 million of equity-based compensation related to NSAM executive restricted stock units, which we refer to as RSUs, which will vest one year from issuance. This amount was determined using the November 9, 2016 closing price of NSAM common stock multiplied by the maximum number of RSUs to be issued of 7,977,000; refer to Note G, *Pro Forma Shares Outstanding and Earnings Per Share* (3).
- (viii) The consolidated pro forma compensation expense of \$202.3 million and \$270.5 million for the nine months ended September 30, 2016 and year ended December 31, 2015 includes \$56.8 million and \$126.0 million of equity-based compensation expense, respectively.
- (ix) Represents the elimination of merger-related transaction costs incurred for the nine months ended September 30, 2016.
- (2) The following table summarizes adjustments to interest expense (in thousands):

		Nine Months Ended September 30, 2016								Year Ended December 31, 2015						
Adjustments to interest expense:	Co	lony ⁽ⁱ⁾		NSAM		NRF		Total	C	olony ⁽ⁱ⁾		NSAM		NRF		Total
Interest expense on NorthStar corporate borrowings(ii)	\$	_	\$	(16,060)	\$	(16,552)	\$	(32,612)	\$	_	\$	(437)	\$	(29,519)	\$	(29,956)
Amortization of deferred financing		_		(4,115)		(2,908)		(7,023)		_		(341)		(5,738)		(6,079)
Total	\$	_	\$	(20,175)	\$	(19,460)	\$	(39,635)	\$	_	\$	(778)	\$	(35,257)	\$	(36,035)

⁽i) The pro forma adjustments assume no additional borrowing on Colony's credit facility or bridge financing. Refer to Note 4.B, Adjustments to the Unaudited Pro Forma Condensed Consolidated Balance Sheet—Merger Adjustments footnote 7 for further information.

- (ii) NorthStar corporate borrowings are expected to be paid off and terminated in connection with the Mergers.
- Represents elimination of historical unrealized losses related to NSAM's ownership of NRF common stock.
- (4) Represents an adjustment to eliminate equity in earnings related to NSAM's interest in Island Hospitality Management Inc. Refer to footnote 2 in Note 4.B, Merger Adjustments—NRF Sales Initiatives, for additional information.
- (5) Represents the income tax effect of pro forma adjustments related to the Mergers, calculated using an estimated 40% effective income tax rate on assets held in taxable REIT subsidiaries.

⁽ii) Represents elimination of dividend income NSAM received from its ownership of NRF common stock for the nine months ended September 30, 2016. NSAM did not earn any dividend income for the year ended December 31, 2015.

- (6) Represents an adjustment related to the non-controlling interest partner in the healthcare joint venture.
- (7) The following table summarizes adjustments to non-controlling interests in the operating partnership (in thousands):

Adjustments to non-controlling interests-operating partnership:	 Months Ended mber 30, 2016	_	ear Ended mber 31, 2015
Allocation to non-controlling interests—Colony NorthStar operating partnership(i)	\$ (33,813)	\$	(26,806)
Elimination of NRF operating partnership(ii)	3,537		3,206
Total	\$ (30,276)	\$	(23,600)

- (i) Represents an adjustment to allocate the pro forma ownership interest of Colony NorthStar of 5.5%. Refer to Note 4.B, Adjustments to the Unaudited Pro Forma Condensed Consolidated Balance Sheet—Merger Adjustments, footnote 11 for additional information.
- (ii) Represents elimination of the non-controlling interests in NRF LP. In connection with the Mergers, NRF LP will merge with NRF, converting non-controlling LTIP unit interests into common stock.

F. Fair Value Adjustment

(8) The following table presents a summary of adjustments related to NRF to amortization of above and below-market leases based on remaining lease terms ranging from one to 29 years (in thousands):

Adjustments to amortization of above/below market leases:	 Months Ended mber 30, 2016	Year Ended December 31, 2015		
Remove historical	\$ 5,016	\$	11,289	
Amortization using preliminary fair value	(21,387)		35,341	
Total	\$ (16,371)	\$	46,630	

(9) The following table presents a summary of adjustments to interest expense related to the fair value of convertible senior notes, securitization bonds payable and mortgage and other notes payable related to NRF amortized over the respective remaining term of each borrowing (in thousands):

Adjustments to interest expense:	Months Ended ember 30, 2016	Year Ended December 31, 2015			
Convertible senior notes	\$ (5)	\$	(7)		
Mortgage and other notes payable	 (1,192)		(2,666)		
Total	\$ (1,197)	\$	(2,673)		

- (10) Represents adjustments to amortization of NRF's below-market ground lease and straight-line ground rent of \$0.5 million and \$0.5 million for the nine months ended September 30, 2016 and year ended December 31, 2015, respectively.
- (11) The following table presents a summary of adjustments to depreciation and amortization based on the historical useful lives for operating real estate and lease and other terms for intangible assets ranging from four to 40 years (in thousands):

	 Nine Months Ended September 30, 2016						Year Ended December 31, 2015						
Adjustments to depreciation and amortization:	NSAM		NRF		Total		NSAM		NRF		Total		
Remove historical	\$ (7,355)	\$	(227,181)	\$	(234,536)	\$	_	\$	(308,615)	\$	(308,615)		
Depreciation and amortization using preliminary fair value	57,298		271,859		329,157		76,398		389,402		465,800		
Total	\$ 49,943	\$	44,678	\$	94,621	\$	76,398	\$	80,787	\$	157,185		

(12) Represents the share of pro forma adjustments to interest and depreciation expenses attributable to non-controlling interests—investments based upon their respective ownership in each venture, as a result of the preliminary fair value adjustments to assets and liabilities.

G. Pro Forma Shares Outstanding and Earnings Per Share

Shares and Units Outstanding

The following table presents a summary of pro forma shares, OP units and RSUs outstanding at the effective time of the Mergers (in thousands):

Colony	NSAM	NRF	Total ⁽⁵⁾
167,053	189,487	198,534	555,074
30,480	1,790	_	32,270
_	7,977	_	7,977
	660	275	935
197,533	199,914	198,809	596,256
	167,053 30,480 —	167,053 189,487 30,480 1,790 — 7,977 — 660	167,053 189,487 198,534 30,480 1,790 — — 7,977 — — 660 275

Refer to Note 4.B, Adjustments to the Unaudited Pro Forma Condensed Consolidated Balance Sheet—Merger Adjustments, footnote 9. Includes shares of both class A and class B pro forma Colony NorthStar common stock.

Refer to Note 4.B, Adjustments to the Unaudited Pro Forma Condensed Consolidated Balance Sheet—Merger Adjustments, footnote 10.

Earnings (Loss) Per Share

The following table presents pro forma basic and diluted earnings (loss) per share after giving effect to the pro forma adjustments to the unaudited consolidated statements of operations (in thousands, except for per share data):

Pro forma earnings per share:	 Months Ended ember 30, 2016	Year	r Ended December 31, 2015
Numerator:			
Net income (loss) from continuing operations attributable to common stockholders	\$ (305,346)	\$	(101,839)
Denominator:			
Weighted average number of shares outstanding—basic	550,078 ⁽¹⁾		548,328 ⁽²⁾
Weighted average number of shares outstanding—diluted	551,916 ⁽¹⁾		549,781 ⁽²⁾
Earnings (loss) per share:			
Net income (loss) from continuing operations attributable to common stockholders per share—basic	\$ (0.56)	\$	(0.19)
Net income (loss) from continuing operations attributable to common stockholders per share—diluted $$	\$ (0.55)	\$	(0.19)

The following table presents pro forma basic and diluted weighted average shares outstanding for the nine months ended September 30, 2016 (in thousands, except for exchange ratios):

Weighted Average Shares—Basic	Colony	NSAM	NRF	Pro Forma Colony NorthStar
Historical weighted average shares—basic	112,133	183,251	180,803	
NSAM executive officers equity-based awards vested upon the Mergers and converted into common stock, $net^{(i)}$	_	3,593	846	
NRF LTIP units converted to common stock(iii)	_	_	1,856	
Shares of NRF common stock owned by NSAM			(2,700)	
Adjusted basic weighted average shares of common stock prior to the Mergers	112,133	186,844	180,805	
Exchange ratio	1.4663	1.0000	1.0996	
Weighted average shares of Colony NorthStar common stock—basic(iii)	164,421	186,844	198,813	550,078

The number of RSUs subject to replacement equity awards may vary based on a per share price equal to the greater of \$15.00 or the volume weighted average price of a share of Colony NorthStar common stock over the first five trading days immediately following the closing of the Mergers. The maximum number of RSUs to be issued are 7,977,000 with a maximum value of \$119.6 million. The above represents the maximum number of NSAM executive RSUs that would be issued; refer to Note E. Merger Adjustments (1)(vii). Represents RSU grants to a non-employee of NSAM and NRF.

Excludes potential shares that may be issued in connection with retention plans or other equity awards issued prior to the Mergers.

Weighted Average Shares—Dilutive	Colony	NSAM	NRF	Colony NorthStar
Historical weighted average shares—dilutive	112,133	185,083	182,664	
NSAM executive officers equity-based awards vested upon the Mergers and converted into common stock, $net^{(i)}$	_	3,593	846	
Shares of NRF common stock owned by NSAM			(2,700)	
Adjusted dilutive weighted average shares of common stock prior to the Mergers	112,133	188,676	180,810	
Exchange ratio	1.4663	1.0000	1.0996	
Weighted average shares of Colony NorthStar common stock—dilutive(iii)	164,421	188,676	198,819	551,916

(i) Represents an adjustment related to NSAM and NRF executive equity-based awards that vest upon the Mergers and converted into class A common stock, net of forfeitures, estimated shares withheld for tax and adjustments due to timing. The adjustment assumes such awards convert to common stock on January 1, 2015, the beginning of the earliest period presented. The adjustment related to NSAM includes 3.0 million executive equity-based shares (10.6 million shares issued net of 4.0 million shares forfeited and 3.7 million shares estimated to be retired upon vesting for tax withholding) and 0.6 million shares due to timing. The adjustment related to NRF includes 0.8 million executive equity-based shares (2.9 million issued net of 1.1 million shares forfeited and 1.0 million shares estimated to be retired upon vesting for tax withholding) and an immaterial amount due to timing.

Dro Forma

- (ii) In connection with the Mergers, NRF LP will be merged into NRF resulting in existing LTIP units to be converted into common stock.
- (iii) Excludes the effect of (a) convertible senior notes that were not dilutive as of September 30, 2016; (b) 8.0 million of NSAM executive RSUs; (c) 2.7 million and 0.5 million shares of NSAM and NRF non-executive RSUs and restricted stock, respectively; (d) 0.7 million and 0.3 million shares of NSAM and NRF RSUs to non-employees, respectively; (e) 0.7 million shares of NSAM restricted stock issued to Townsend Holdings LLC; (f) 2.1 million shares of Colony restricted stock; and (g) potential shares that may be issued in connection with retention plans or other equity awards issued prior to the Mergers.
- (2) The following tables present pro forma basic and diluted weighted average shares outstanding for the year ended December 31, 2015 (in thousands, except for exchange ratios):

Weighted Average Shares—Basic	Colony	NSAM	NRF	Pro Forma Colony NorthStar
Historical weighted average shares—basic	110,931	188,706	174,873	
NSAM executive officers equity-based awards vested upon the Mergers and converted into common stock, $net^{(i)}$	_	4,332	1,155	
NRF LTIP units converted to common stock ⁽ⁱⁱ⁾	_	_	1,856	
Shares of NRF common stock owned by NSAM	_	_	(2,700)	
Adjusted basic weighted average shares of common stock prior to the Mergers	110,931	193,038	175,184	
Exchange ratio	1.4663	1.0000	1.0996	
Weighted average shares of Colony NorthStar class A common stock—basic(iii)	162,658	193,038	192,632	548,328
Weighted Average Shares—Dilutive	Colony	NSAM	NRF	Pro Forma Colony NorthStar
Historical weighted average shares—dilutive	110,931	193,119	176,345	
NSAM executive officers equity-based awards vested upon the Mergers and converted into common stock, $net^{(i)}$	_	1,794	1,155	
Shares of NRF common stock owned by NSAM			(2,700)	
Adjusted dilutive weighted average shares of common stock prior to the Mergers	110,931	194,913	174,800	

1.4663

162,658

1.0000

194,913

1.0996

192,210

549,781

Exchange ratio

Weighted average shares of Colony NorthStar common stock—dilutive(iii)

- (i) Represents NSAM and NRF executive equity-based awards that vest upon the Mergers and converted into common stock, net of forfeitures, estimated shares withheld for tax and adjustments due to timing. The adjustment assumes such awards convert to common stock on January 1, 2015, the beginning of the earliest period presented. The adjustment related to NSAM includes 3.0 million net shares issued, as described in footnote 1(i), and 1.3 million shares due to timing. Diluted shares for the year ended December 31, 2015 also includes an adjustment to exclude 2.6 million shares that were included in the historical dilutive for such period presented. The adjustment related to NRF includes 0.8 million net shares issued, as described in footnote 1(i), and 0.3 million shares due to timing.
- (ii) In connection with the Mergers, NRF LP will be merged into NRF resulting in existing LTIP units converted to common stock.
- (iii) Excludes the effect of: (a) convertible senior notes that were not dilutive as of December 31, 2015; (b) 8.0 million of NSAM executive RSUs; (c) an immaterial amount of NRF and 2.3 million shares of NSAM non-executive RSUs and restricted stock; (d) 0.7 million and 0.3 million of NSAM and NRF RSUs to non-employees, respectively; (e) 1.2 million shares of Colony restricted stock; and (f) potential shares that may be issued in connection with retention plans or other equity awards issued prior to the Mergers.